ODISHA ELECTRICITY REGULATORY COMMISSION PLOT NO. 4, CHUNOKOLI, SHAILASHREE VIHAR, BHUBANESWAR – 751021 Tel. No. (0674) 2721048/ 2721049 (PBX), Fax : (0674) 2721053/2721057 E-mail : <u>orierc@gmail.com</u> Website : <u>www.orierc.org</u> **********

<u>PUBLIC NOTICE</u> Case No. 48 of 2024

Hearing of Application for Approval of Tariff for purchase of power from Solar Power Plants set up by Solar Power Generators (SPGs) under RESCO mode for implementation of Feeder Level Solarization (FLS) under PM Kusum C (Component C-2) and (b) Approval of the Draft PPA.

Sir,

M/s.TP Central Odisha Distribution Ltd.(TPCODL), M/s. TP Western Odisha Distribution Ltd. (TPWODL), M/s. TP Northern Odisha Distribution Ltd.(TPNODL) and M/s. TP Southern Odisha Ltd.(TPSODL) have filed a common application before this Commission for Approval of Tariff for purchase of power from Solar Power Plants set up by Solar Power Generators (SPGs) under RESCO mode for implementation of Feeder Level Solarization (FLS) under PM Kusum C (Component C-2) and (b) Approval of the Draft PPA.The Commission has registered it as Case No.48 of 2024 and has decided to dispose of this case through a public hearing. The common petition along with all Annexures submitted by M/s. TPCODL in this regard is available in OERC website (www.orierc.org) and also in TPCODL, TPWODL, TPNODL & TPSODL's website www.tpcentralodisha.com, www.tpwesternodisha.com, www.tpnodl.com and www.tpsouthernodisha.com. The persons/ organizations those who are interested to participate in the above proceeding may file their objections/suggestions, if any, on the common petition of M/s. TPCODL & other DISCOMs of Odisha on or before 03.09.2024 through affidavits with a copy to the petitioners. M/s. TPCODL, M/s. TPWODL, M/s. TPNODL & M/s. TPSODL are directed to file their rejoinders to the objections/suggestions of the Respondents by 03.09.2024 with the Commission along with serving a copy of the same on the Respondents. Those persons who have filed their objection/suggestion by the due date to the above common application of M/s. TPCODL, M/s. TPWODL, M/s. TPNODL & M/s. TPSODL shall be summoned to participate in the hearing of the application by the Commission on 03.09.2024 at 11.00 AM through Hybrid mode to be held OERC Office at Bhubaneswar.

By Order of the Commission

Dated:14.08.2024

Sd/-SECRETARY

BEFORE THE ODISHA ELECTRICITY REGULATORY COMMISSION, BIDYUT NIYAMAK BHAWAN. PLOT No-4, CHUNOKOLI, SHAILASHREE VIHAR, BHUBANESWAR-751021

Case No ____ of 2024

IN THE MATTER OF: Petition filed by TPCODL on behalf of all the four Discoms of Odisha for a) Approval of Tariff for purchase of power from Solar Power Plants set up by Solar Power Generators (SPGs) under RESCO mode for implementation of Feeder Level Solarization (FLS) under Kusum C (Component C-2) and (b) Approval of the Draft PPA.

And

IN THE MATTER OF: TP Central Odisha Distribution Ltd (TPCODL), TP Western Odisha Distribution Ltd (TPWODL), TP Northern Odisha Distribution Ltd (TPNODL), TP Southern Odisha Distribution Ltd (TPSODL)--represented by its Chief –Regulatory & Enforcement of TPCODL, Corporate Office, Power House, Unit 8, Bhubaneswar-751 012.

> Department of Energy, Government of Odisha Kharavel Bhavan Bhubaneshwar 751001

Odisha Renewable Energy Development Agency (OREDA)

Address S-3/59, Mancheswar Industrial Estate Bhubaneshwar- 751010

...Petitioner

....Respondents

IN THE MATTER OF:



Affidavit

The Chief Executive Officer

Janpath, Bhubaneshwar 751022

All concerned Stakeholders

M/s GRIDCO Ltd

09 JUL 2024

I, Bharat Kumar Bhadawat, aged about 53 son of late Shri Shankar Lal Bhadawat residing at Bhubaneswar do hereby solemnly affirm and say as follows:

- 1. I am the Chief-Regulatory & Enforcement of TP Central Odisha Distribution Ltd. the Petitioner in the above matter and I am duly authorized to swear this affidavit on behalf of above petitioners.
- 2. The statements made in the submission herein shown to me are based on information provided to me and I believe it to be true.

Bhubaneswar. Dated: 09.07.2024



The Deponent above named being dury identified by Sri. Advocate at appears before me on O dt O 1912 by about. P.M. and states on oath that the contents of this affidavit are true to the best of his/her knowledge

Beliadau

(Bharat Kumar Bhadawat) Chief-Regulatory & Enforcement

ANIL KUMAR MOHANTY NOTARY, BBSR REGD. No.+ON-116/2009

BEFORE THE ODISHA ELECTRICITY REGULATORY COMMISSION, BIDYUT NIYAMAK BHAWAN. PLOT No-4, CHUNOKOLI, SHAILASHREE VIHAR, BHUBANESWAR-751021

Case No _____ of 2024

- IN THE MATTER OF: Petition filed by TPCODL on behalf of all the four Discoms of Odisha for a) Approval of Tariff for purchase of power from Solar Power Plants set up by Solar Power Generators (SPGs) under RESCO mode for implementation of Feeder Level Solarization (FLS) under Kusum C (Component C-2) and (b) Approval of the Draft PPA.
- And TP Central Odisha Distribution Ltd (TPCODL), TP Western Odisha IN THE MATTER OF: Distribution Ltd (TPWODL), TP Northern Odisha Distribution Ltd (TPNODL) , TP Southern Odisha Distribution Ltd (TPSODL)--represented by Chief --Regulatory & Enforcement of TPCODL, Corporate Office, Power House, Unit 8, Bhubaneswar- 751 012. ...Petitioner IN THE MATTER OF: The Chief Executive Officer Odisha Renewable Energy Development Agency (OREDA) Address S-3/59, Mancheswar Industrial Estate Bhubaneshwar-751010 M/s GRIDCO Ltd Janpath, Bhubaneshwar 751022 Department of Energy, Government of Odisha Kharavel Bhavan Bhubaneshwar 751001 All concerned Stakeholders

.....Respondents

A. Background for the petition

- PM- KUSUM is a scheme launched by government of India in 2019 for the economic betterment of farmers through deployment of renewable energy for decentralized power generation, water pumping etc. Then scheme has three component out of which component C has two sub-components namely Individual Pump Solarisation and Feeder Level Solarisation (FLS). Government of Odisha has declared the DISCOMs of the state as the State Implementing Agency (SIA) for this component.
- 2. For Implementation of Individual Pump Solarisation (IPS) in the State the Discoms have filed a petition (registered as Case 41/2022 and 62/2022) for determination of tariff of solar projects set up under IPS and approval of PPA to be signed by DISCOMS / GRIDCO with farmers. The Hon'ble Commission vide order dated 19.01.2023 in the above case matters had approved the tariff and also a model Power Purchase Agreement (PPA).

Besuadaway.

- 3. However, the sub-component for which the current petition is being filed i.e. Feeder Level Solarisation is different from the Individual Pump Solarisation component as this envisages setting up of large solar power plants to cater to the aggregate power demand of all pumps connected to an agricultural feeder or mixed feeder in RESCO mode and purchase of the generated power from the RESCO operator.
- 4. In this petition, we are seeking approval of the Tariff for purchase of power from Solar Power Plants set up by Solar Power Generators (SPGs) under RESCO mode for implementation of Feeder Level Solarization (FLS), Approval of Draft PPA and approval for signing PPAs with the SPGs.

B. Kusum Scheme in Brief

- 5. Government of India has taken various policy measures to increase the installed power generation capacity from non-fossil fuel sources by 2030. To provide energy and water security to farmers and enhance their income, de-dieselize the farm sector, and reduce environmental pollution, the Government of India launched PM-KUSUM on 08.03.2019. A Copy of the Order is enclosed as **Annexure-1**.
- 6. During 2020-21, the scheme was scaled-up and expanded from its pilot stage and notified with due approval of Ministry of Finance. In August 2022, the scheme was extended till March 2026. A Copy of the Order is enclosed as **Annexure-2**.
- 7. The various schemes under Kusum proposed by the Government of India are as given below:

i. *Component-A*: Setting up of 10,000 MW of Decentralized Ground/ Stilt Mounted Grid Connected Solar or other Renewable Energy based Power Plants by the farmers on their land;

ii. Component-B: Installation of 14 Lakh Stand-alone Solar Agriculture Pumps; and

iii. *Component-C*: Solarisation of 35 Lakh Grid Connected Agriculture Pumps including through feeder level solarisation.

- 8. The PM-KUSUM Scheme allows inter-se transfer of quantities between Component B and Component –C. All the three components of the scheme aim to add Solar capacity of about 34,800 MW with total Central Financial Support of Rs. 34,422 Cr.
- 9. The Department of Energy, Govt. of Odisha, vide Letter no- 8686 dated 12th Sep 2022 has allocated solarization of total 10,000 no. of feeder level Agricultural Pump under FLS of PM Kusum C Scheme. The copy of the mentioned letter is enclosed as Annexure-3 to this submission.
 - C. Implementation under RESCO mode and eligible CFA

Ecoladorof.

10. As per MNRE Order dated 17.01.2024 on Comprehensive Guidelines for implementation of PM-KUSUM scheme following is stipulated with regards to implementation of Feeder level Solarization (KUSUM – C-2) under RESCO mode and eligible CFA. A copy of the mentioned order is enclosed as Annexure-4.

i. For selection of RESCO developer and PPA, the Guidelines and Model PPA issued by MNRE for implementation of Component-A of PM-KUSUM Scheme may be used, with suitable modifications.

ii. The maximum timeline allowed for commissioning of solar power plant by RESCO Developer will be 12 months from date of issuance of LoA. However, the implementing agency may increase this time on case to case basis, up to the sanctioned timeline given by MNRE.

iii. The process of selection of RESCO Developer and signing of PPA should normally be completed by implementing agency within six months from the date of issuance of sanction by MNRE.

iv. A suitable clause on PPA shall be included to ensure that RESCO developer does not sell solar power to any other buyer except the concerned DISCOM during tenure of PPA, including penal provision calling for refund of proportionate CFA granted to the project by MNRE on violation of this condition.

v. For installation of feeder level solar power plants through RESCO model, the developers will be selected on the basis of lowest tariff offered for supply of required solar power for a period of 25 years after taking into account the subsidy being given by Govt. of India.

vi. In case of any liquidation of assets of the solar power plant prior to completion of PPA period, the first charge shall be towards recovery of proportionate CFA granted to the project by MNRE.

vii. The developer will get CFA @ 30% of the estimated cost of installation of solar power plant.

viii. The solar power supplied by RESCO developer would much cheaper than present cost of power delivered at distribution sub-station and therefore, DISCOM will save the amount equal to difference between the two.

ix. In the RESCO model the burden of electricity subsidy for agriculture will be reduced to the extent of difference mentioned above and not become zero as in case of CAPEX model, where once the loan is repaid, subsidy support from state Government is no longer required.

x. States may choose to provide upfront subsidy in lieu of electricity subsidy being given to agriculture consumers. This upfront subsidy from state could be

Besundaion

in the form of VGF to RESCO developer, in addition to CFA, to supply power to farmers of an agriculture feeder at present subsidised rates or any other rate fixed by state Government.

D. Appropriate Tariff for power from projects under Kusum _C Set up for Feeder Level Solarisation

11. In our humble submission, for the scheme to be successful, the tariff for power injected from the solar installation should be adequate for any Solar Power Generators to set up the plants under the scheme. At the same time, the tariff should not bring additional burden on the consumers.

E. Tariff Discovered through Competitive Bidding

- 12. For arriving at a competitive tariff for purchase of power from SPGs under the schemes, TP Discoms have taken the route of tariff discovery through competitive bidding process. TPCODL on behalf of all four discoms had invited competitive bids from SPGs for discovery of levelised tariff for all four Discoms. The Scope of the bid was selection of SPGs for setting up of Solar plants under RESCO mode on the basis of lowest levelised tariff for supply of solar power for 25 years which shall be guided by the PPA to be signed between M/s GRIDCO Ltd. and the SPGs.
- 13. The Lowest Tariff Discovered through this competitive bidding is provided in table below. The minutes of the meeting (MoM) of the mentioned final discovered tariff is enclosed as **Annexure-5** to this submission.

	Particulars	
With DCR	If land is purchased /taken on lease by the Developer mutually agreed lease rent	3.75
	If land is provided by Government on nominal lease rent	3.65
Without DCR	If land is purchased /taken on lease by the Developer mutually agreed lease rent	3.35
	If land is provided by Government on nominal lease rent	3.25

Table-1: Lowest Tariff Discovered through Competitive Bidding

14. It is submitted that as per the Office Memorandum No- 32/645/2017-SPV Divison of MNRE dated 14th Feb 2024, the DCR exemption would be valid for all those Letter of Awards which have been issued on or before 31.03.2024. However, as the competitive bidding process by TPCODL for all Discoms was initiated in Dec 2023 the

Bebladaion

tariff bidding was done for both DCR and without DCR basis. A copy of the mentioned MNRE Office Memorandum is enclosed as **Annexure-6** to this submission.

15. It is submitted that the approved tariff for purchase of Solar Power in GRIDCO Tariff Order FY 2024-25 is Rs. 3.6827 per unit (Table -56 of GRIDCO Tariff Order FY-25). Further, the average power purchase cost approved for all Four Discoms in GRIDCO Tariff Order FY 2024-25 is Rs. 3.28 per unit (Table -70 of the Gridco Tariff Order FY-25 is provided below for ease of reference). Considering these, the discovered tariff as provided in Table-1 above is quite competitive.

Name of the DISCOMs	Existing Bulk Supply Price approved for FY 2023-24 (P/U)	Quantum of Energy approved for sale during FY 2024-25 (MU)	Total Expected Revenue for FY 2024-25 (Rs. Crore)	Bulk Supply Price approved for FY 2024-25 (P/U)	
TPCODL	305.00	12513.00	3941.60	315.00	
TPNODL	335.00	8163.00	2857.05	350.00	
TPWODL	390.00	11940.00	4537.20	380.00	
TPSODL	210.00	4924.00	984.80	200.00	
Total	329.00	37540.00	12320.65	328.20	

Table - 70	
BSP and Quantum of Energy Approved by the Commission for FY 2024-25	

F. Suggested Tariff for Injection of Power under Kusum C-2 (FLS)

16. In view of the above discussions, it is submitted that Tariff for Kusum C-2 Feeder Level Solarization Scheme (FLS) be fixed at the lowest levelised tariff discovered through competitive bidding as mentioned in **Table-1**.

G. Directive of the Department of Energy, Govt. of Odisha

17. In the meeting held on PM-Suryaghar Muft Bijli Yojana scheme and PM KUSUM Scheme under the Chairmanship of Principal Secretary to Govt. of Odisha, Department of Energy it was directed that the Discoms would file a petition before the Hon'ble Commission for approval of discovered tariff under KUSUM-A and Feeder level solarization scheme of KUSUM-C. The MoM of this meeting is enclosed as **Annnexure-7.**

H. Approval of PPA

- 18. Under this scheme a PPA needs to be signed between M/s GRIDCO Ltd. and the SPGs for purchase of power. A draft model PPA in this regard has been prepared and enclosed as **Annexure-8** for approval of the Hon'ble Commission.
- 19. Since all the Power Purchases related affairs for Odisha Discoms are being look after by M/s GRIDCO Ltd., therefore we are of the view that the PPA for purchase of power from SPGs should be executed between M/s GRDICO Ltd. and the SPGs.

Berkharlander.

I. Other Methodology

20. Other methodology for implementation of the scheme will be as per MNRE guideline as released from time to time.

J. Prayers

- 21. TPCODL on behalf of all the Discoms prays for the following before the Hon'ble Commission
 - a) Approval of the tariff (as per Table-1) for purchase of power form Solar Power Generators (SPGs) for projects to be set up under Component 2 of Kusum C Scheme under Feeder Level Solarisation (FLS).
 - b) Approval for signing of the PPA between M/s GRIDCO Ltd. and the Solar Power Generators (SPGs) for purchase of power injected to the network.
 - c) Approval of the draft model PPA attached along with the petition.
 - d) Pass any other direction it deems fit.

Bebinordardef.

Annexure -1

No. 32/645/2017-SPV Division Government of India Ministry of New and Renewable Energy

Block No. 14, CGO Complex, Lodhi Road, New Delhi-110003

Dated: 08 March 2019

<u>ORDER</u>

Subject: Scheme for farmers for installation of solar pumps and grid connected solar power plants.

The sanction of the President of India is hereby accorded for a scheme for farmers for installation of solar pumps and grid connected solar and other renewable power plants in the country. The scheme aims to add solar and other renewable capacity of 25,750 MW by 2022 with total central financial support of Rs. 34,422 Crore including service charges to the implementing agencies.

- 2. The Scheme consists of three components:
 - i. Component A: 10,000 MW of Decentralized Ground Mounted Grid Connected Renewable Power Plants of individual plant size up to 2 MW.
 - ii. Component B: Installation of 17.50 lakh standalone Solar Powered Agriculture Pumps of individual pump capacity up to 7.5 HP.
- iii. Component C: Solarisation of 10 Lakh Grid-connected Agriculture Pumps of individual pump capacity up to 7.5 HP.

3. Components A and C of the Scheme will be implemented in Pilot mode till 31st December 2019. The Component B, which is a ongoing sub-programme, will be implemented in entirety without going through pilot mode. The capacities to be implemented under pilot mode for the Components A and C are as follows:

- i. Component A: Commissioning of 1000 MW capacity of ground/ stilt mounted solar or other renewable energy source based power projects
- ii. Component C: Solarization of 1,00,000 grid connected agriculture pumps

On successful implementation of pilot run of Components A and C of the Scheme, these components would be scaled-up, after getting necessary approval.

4. Financial outlay for implementation of Component A and Component C in pilot mode and full-fledged implementation of Component B are given below:

Total Project Capacity	: 10,000 MW
Total Central Financial Assistance (CFA)	Rs. 18,667.5 Crore
Service Charges	: Rs. 369 Crore
Total Central Financial Support	: Rs. 19,036.5 Crore

On successful completion of pilot run of Component A and Component C, additional central financial support required for scaling-up of these components is given below:

Additional Project Capacity for scaling-up	:	15,750 MW
of Component A and Component C		
Total CFA	:	Rs. 15,120 Crore
Service Charges	:	Rs. 265.5 Crore
Total Central Financial Support	:	Rs. 15,385.5 Crore

5. The above-mentioned financial requirement for implementation of Component A and Component C in pilot mode and full-fledged implementation of Component B, Rs. 10,000 Crore are proposed to be provided through Gross Budgetary Support (GBS) and balance Rs. 9,036.5 Crore through Extra Budgetary Resources (EBR). The financial requirement of Rs. 15,385.5 Crore for scaling-up of Component A and Component C after successful completion of pilot run will be met entirely through EBR.

6. Year wise CFA requirement and break up between GBS (Gross budgetary Support) and EBR (Extra Budgetary Sources) considering the pilot run of Component A and Component C and full-fledge implementation of Component B are mentioned below:

Year	Funds required for. Component			CONTRACTOR OF CONT	Funds through	
	A	в	C	Requirement	GBR	EBR
2019-20	35.5	900	600	1535.5	1535.5	0
2020-21	66	2000	777	2843	2000	843
2021-22	66	5000		5066	2000	3066
2022-23	66	5000	1	5066	2000	3066
2023-24	66	4427		4493	2464.5	2028.5
2024-25	33	0	0	33		33
Total	332.5	17,327	1,377	19,036.5	10,000	9,036.5

Year-wise requirement of additional funds through EBR for scaling-up of Component A and Component C after success of pilot run are mentioned below:

Year	Funds requir	ed for Component	Total Requirement to be funded
теяг	A	С	through EBR
2020-21	106.5	500	606.5
2021-22	304.5	4000	4304.5
2022-23	502.5	4000	4502.5
2023-24	594	3893	4487
2024-25	594	0	594
2025-26	495	0	495
2026-27	297	0	297
2027-28	99	0	99
Total	2,992.50	12,393.00	15,385.50

Page 2 of 3

7. The detailed Operational Guidelines for implementation of the Scheme will be issued separately.

8. This is issued in exercise of delegated powers conferred to this Ministry and with the concurrence of IFD vide Dy. No. 579 dated 07.03.2019.

(Shobhit S astava) Scientist-C

То

- 1. All Central Ministries and Departments;
- ACS/Principal Secretaries/Secretaries(Renewable Energy/Energy/Power) of all States/UTs
- 3. Discoms/State Nodal Agencies of all States/UTs
- 4. Principle Director of Audit, Scientific Audit-II, DGACR Building, I.P. Estate, Delhi
- 5. Pay and Accounts Officer, Ministry of New and Renewable Energy, New Delhi

Copy to:

- 1. PS to Hon'ble Minister for Power and New & Renewable Energy
- 2. PPS to Secretary/AS/AS&FA, MNRE
- 3. All Advisers & Group Heads
- 4. Director (NIC) to upload this on the Ministry's website
- 5. Sanction folder

(Shobhit S Scientist-C



Annexure-2

No. 32/645/2017-SPV Division Government of India Ministry of New and Renewable Energy

Atal Akshay Urja Bhawan, Lodhi Road, New Delhi-110003 Dated: 1 August 2022

<u>ORDER</u>

Subject: Extension of Pradhan Mantri Urja Suraksha evam Utthaan Mahabhiyaan (PM-KUSUM).

This is in continuation to Order of even number dated 08 March 2019 regarding issuance of administrative approval of Pradhan Mantri Urja Suraksha evam Utthaan Mahabhiyaan (PM-KUSUM) and Order of even number dated 04 November 2020 regarding scale-up and expansion of the PM-KUSUM. Implementation Guidelines of the PM-KUSUM were issued vide OM dated 22.07.2019. Further, implementation guidelines for Component-C were issued on 08.11.2019 and for Feeder Level Solarization on 04.12.2020. Based on the feedback received from the stakeholders, amendments have been issued to these Guidelines from time to time.

2. The PM-KUSUM Scheme has been extended till 31.03.2026, with targets under the three components remaining unchanged and allowing inter-se transfer of quantities across Component-B and Component-C.

3. Along with the extension, the following amendments have been made to the implementation Guidelines of the Scheme:

- i. Under Component-B and Component-C of the Scheme, Central Financial Assistance (CFA) will be available for pump capacity up to 15 HP to the individual farmers in the North-castern States; UTs of Jammu & Kashmir and Ladakh; and the States of Uttarakhand and Himachal Pradesh. However, the CFA for pumps up to 15 HP will be restricted to 10% of the total installations.
- ii. The condition of the domestic content requirement for solar cells has been waived off for the feeder level solarization under Component-C for the projects awarded by the implementing agency on or before 20.06.2023.
- Budgetary allocation of Rs. 10,000 Crore approved by CCEA will be utilized before accessing the extra-budgetary resources.
- 4. This Order is issued with the approval of the Competent Authority.

(Jeevan Kumar Sr. Director/Scientist-F

To

- 1. All Central Ministries and Departments;
- 2. Principle Director of Audit, Scientific Audit-II, DGACR Building, I.P. Estate, Delhi
- 3. ACS/Principal Secretaries/Secretaries (Energy Departments) of all States/UTs



- 4. State Implementation Agencies for all three components in all States
- 5. Pay & Accounts Officer, MNRE
- 6. Indian Banks Association (IBA)

Copy to:

- 1. PS to Hon'ble Minister for Power and New & Renewable Energy
- 2. Sr. PPS to Secretary, MNRE/ Sr. PPS to AS, MNRE/ PPS to JS & FA. MNRE/ PPS to JS(Solar), MNRE
- 3. All Advisers & Group Heads
- 4. Director (NIC) to upload this on the Ministry's website
- 5. Hindi section for Hindi version
- 6. Sanction földer

(Jeevan Kumar Jethani) Sr. Director/Scientist-F





Annexure-3

GOVERNMENT OF ODISHA ENERGY DEPARTMENT

No. 8686 /En., Bhubaneswar, dt. / 2 September, 2022

From

Sri Sambit Parija, OFS(SAG)

FA-cum-Additional Secretary to Government

To

The Chief Executive Officer, TPCODL, Bhubaneswar / Chief Executive Officer, TPWODL, Burla, Sambalpur / Chief Executive Officer, TPNODL, Balasore / Chief Executive Officer, TPSODL, Berhampur.

Sub: Allocation of quantities of agricultural pumps under Component-C of PM-KUSUM scheme.

Sir,

With reference to this Department letter No.7695/E dt.10.08.2022 on the above noted subject I am directed to inform that the allocation of quantities of individual grid-connected agricultural pumps and feeder level agricultural pumps for solarisation under Component-C of PMKUSUM scheme is made among 04 DISCOMS of the State in the following manner.

DISCOM	pumps fo	Grid connect r solarisation	ed Agri. (Nos.)	No. of feeder level agricultural pumps (Nos.)		
	Earlier allocation	Present allocation	Total	Earlier allocation	Present allocation	Total
1	2	3	4	S	anocation	
TPCODL	1246	4754	6000	107	0	7
TPSODL		and the second se		197	1303	1500
	1413	5387	6800	223	1477	1700
TPWODL	4155	15845	20000	655	4345	
TPNODL	1496	5704				5000
Total			7200	236	1564	1800
TOTAL	8310	31690	40000	1311	8689	10000

DISCOM-WISE ALLOCATION OF COMPONENT-C OF PM-KUSUM SCHEME

You are, therefore, requested to take follow up action on implementation of the scheme following due instructions of Ministry of New & Renewable Energy (MNRE), Government of India.

Yours faithfully,

Governmen

Memo No.

12/9/2022

Copy forwarded to Sri Shobhit Srivastava, Scientist-D, Ministry of New & Renewable Energy, Govt. of India, Block No.14, CGO Complex, New Delhi -110003 for favour of his kind information and necessary action with reference to this Department Memo No.7696/E dt.10.08.2022 and his email dtd.30.08.2022.

/En., dtd.

Memo No.

cum-Additional secretary to Government En., dtd.

FA-cum-Additional Sec

Copy forwarded to the Chief Executive, OREDA, Bhubaneswar for favour of her kind information and necessary action with reference to this Department Memo No.7697/E dt.10.08.2022.

Annexure-4

32/645/2017-SPV Division Government of India Ministry of New & Renewable Energy

Atal Akshay Urja Bhawan Lodhi Road, New Delhi Date: 17.01.2024

<u>Order</u>

Subject: Comprehensive guidelines for implementation of Pradhan Mantri Kisan Urja Suraksha Evam Utthaan Mahabhiyaan (PM-KUSUM) Scheme

In supersession to this Ministry's Office Memorandum 32/645/2017-SPV Division dated 22.07.2019 regarding guidelines for implementation of Pradhan Mantri Kisan Urja Suraksha Evam Utthaan Mahabhiyaan (PM-KUSUM) Scheme and all the amendments so far under this scheme, it has been decided to issue a comprehensive guidelines for implementation of PM KUSUM Scheme, with following components:

- i. Component-A: Setting up of 10,000 MW of Decentralized Ground/ Stilt Mounted Grid Connected Solar or other Renewable Energy based Power Plants;
- ii. Component-B: Installation of 14 Lakh Stand-alone Solar Agriculture Pumps;
- iii. Component-C: Solarisation of 35 Lakh Grid Connected Agriculture Pumps including Feeder Level Solarization.

2. The comprehensive guidelines for implementation of PM KUSUM Scheme are enclosed.

This is issued with the approval of Competent Authority.

Enclosed: As above.

3.

(Dr. Nikhil Gakkhar) Scientist 'D' 011-20849115

To All concern



Guidelines for Implementation of New Scheme for Farmers for Installation of Solar Pumps and Grid Connected Solar Power Plants

1. Background

- 1.1. Government of India has taken various policy measures to increase the installed power generation capacity from non-fossil fuel sources by 2030. To provide energy and water security to farmers and enhance their income, de-dieselize the farm sector, and reduce environmental pollution, the Government of India launched PM-KUSUM on 08.03.2019.
- **1.2.** During 2020-21, the scheme was scaled-up and expanded from its pilot stage and notified with due approval of Ministry of Finance. In August 2022, the scheme was extended till March 2026.

2. Components of PM KUSUM

- 2.1. The PM KUSUM Scheme has the following components:
 - i. Component-A: Setting up of 10,000 MW of Decentralized Ground/ Stilt Mounted Grid Connected Solar or other Renewable Energy based Power Plants by the farmers on their land
 - ii. Component-B: Installation of 14 Lakh Stand-alone Solar Agriculture Pumps
- iii. Component-C: Solarisation of 35 Lakh Grid Connected Agriculture Pumps including Feeder Level Solarization
- 2.2. The PM-KUSUM Scheme allows inter-se transfer of quantities between Component-B and Component-C. All three components of the scheme aim to add Solar capacity of about 34,800 MW by March 2026 with the total Central Financial support of ₹ 34,422 crore.

3. Implementation mechanism

These consolidated guidelines subsume all the previously issued amendments from time to time.



4. Component A

4.1. Salient Features

- i. This Component aims at setting up of 10,000 MW of Decentralized Ground/ Stilt Mounted Grid Connected Solar or other Renewable Energy based Power Plants by farmers on their land.
- ii. Solar or other Renewable Energy based Power Plants (REPP) of capacity of 500 kW to 2 MW will be setup by Renewable Power Generator (RPG).
- iii. The REPP will be preferably installed within five km radius of the sub-stations in order to avoid high cost of sub-transmission lines and to reduce transmission losses.
- iv. The total energy purchased from these RE plants will be accounted for Renewable Purchase Obligation (RPO) under Decentralized Renewable Energy (DRE) category by the DISCOM.

4.2. Eligibility

- 1. The REPP under the scheme would be installed by the farmers on his own land either directly by himself or in partnership with group of farmers/ cooperatives/ panchayats/ Farmer Producer Organisations (FPO)/Water User associations (WUA), or through a developer.
- ii. Group of farmers/ cooperatives/ panchayats/ Farmer Producer Organisations (FPO)/ Water User associations (WUA) etc. can opt for developing the REPP through developer(s). The farmer may provide his land to the DISCOM, which will then be considered as RPG in this case. In such a case, the land owner will get lease rent as mutually agreed between the parties. The lease rent may be in terms of Rs. per year per acre of land or in terms of Rs. per unit energy generated per acre of land area. The farmer(s) may opt for payment of lease rent directly in their bank account by the DISCOM, from the payment due to the developer. A model Land Lease Agreement to facilitate the beneficiaries is as per Annexure I. However, the terms of Land Lease Agreement may be finalised with mutual consent of concerned parties.

4.3. Notification of sub-station wise generation capacity and intent to setup REPP

- i. DISCOM shall assess and notify RE generation capacity that can be injected into 33/11 kV or 66/11 kV or 110/11 kV sub-station in rural areas and place such notification on its website for information of all stakeholders.
- ii. DISCOMs will notify sub-station wise surplus capacity which can be fed from such RE power plants to the Grid
- iii. DISCOMs should prefix the levelized tariff.
- iv. DISCOM shall invite applications from interested beneficiaries for setting up the renewable energy plants.
- v. The renewable power generated will be purchased by DISCOMs at the pre-fixed levelized tariff.
- vi. In case the aggregate capacity offered by Applicants is more than notified capacity for a particular sub-station, bidding route will be followed by DISCOMs to select Renewable Power generator and in such cases the pre fixed levelized tarifful be the ceiling tariff for bidding.



- vii. For setting up of plant, land will be required. To facilitate farmers willing to lease out their land for development of RE plants near the notified substation(s), DISCOMs may make a list of such farmers and place that list on their website.
- viii. However, the leasing of land of any farmers will be a bi-partite agreement between the farmer and the developer, and the DISCOM will not be held responsible for failure in getting the land leased out to a developer.

4.4. Application procedure & Selection of REPP

- i. DISCOM may invite applications for Expression of Interest (EOI) from prospective RPGs, through their online portal.
- ii. The RPGs shall submit their application for setting up REPP to DISCOM/SIA, as per prescribed format.
- iii. The application should be submitted along with a non-refundable processing fee from the interested RPGs, which in no case shall be higher than Rs. 5000 per MW or part thereof of the capacity applied for.
- iv. In case the total aggregate capacity of eligible applications received for a particular substation is less than or equal to the capacity notified for connectivity at the sub-station, Letter of Award (LoA) will be awarded to all eligible applicants for procurement of renewable power at a pre-fixed levelized tariff on first come first serve basis within one month from the receipt of application
- v. In case the total aggregate capacity of eligible application received for a particular substation is more than the capacity notified for connectivity at the sub-station, then DISCOM or any agency authorized by the DISCOM shall invite Bids from all these applicants.
- vi. Selection of bidders will be based on the lowest tariff offered in the ascending order as quoted by the bidders in the closed bid or e-reverse auction. In such cases, the pre-fixed levelised tariff will be the ceiling tariff for bidders.
- vii. In case of bidding, all eligible applicants will have to submit tariff bids within a prescribed time. Selection of bidders will be based on the **lowest tariff** offered in the ascending order as quoted by the bidders in the **closed bid**. LoA will be awarded to all successful bidders.

4.5. Clearances required from the State Government and other local bodies

The RPG is required to obtain necessary clearances as required for setting up the REPP in coordination with DISCOMs. District Collectors should facilitate the clearances required for issues related to land permissions, right of way, etc.

4.6. Power Purchase Agreement (PPA)

pcor

- i. A model PPA (Power Purchase Agreement) to be executed between RPG and DISCOMs is as per Annexure II.
- ii. The duration of PPA will be 25 years from Commercial Operation Date (COD) of the project.
- iii. The PPA will have to be executed by RPG within two months from the date of issue of LoA by DISCOM or any agency authorized by the DISCOM.
- iv. The DISCOM will be obliged to buy the entire power from RPG within the contract capacity.

- v. The RPG will be free to operate the plant after expiry of the 25 years of PPA period if other conditions (like land lease, etc.) permits.
- vi. However, any extension of the PPA period beyond 25 years shall be through mutual agreement between the RPG and DISCOM.
- vii. As a payment security measure, DISCOM will have to maintain Escrow Arrangement as defined in the PPA.

4.7. Bank Guarantees (BG)

- i. In case of bidding, the RPG shall provide Earnest Money Deposit (EMD) of Rs. 1 Lakh/MW in the form of Bank Guarantee along with EoI.
- ii. The EMD of Rs. 1 Lakhs per MW will be converted into Performance Bank Guarantee (PBG) for the successful bidders, who have signed the PPA within stipulated time period.
- iii. In case the selected RPG fails to execute the PPA within the stipulated time period, the Bank Guarantee equivalent to EMD shall be encashed by DISCOM as penalty.
- iv. In case any bidder is not selected, DISCOM shall release the EMD within 15 days of the date of issue of LoA to selected RPG(s).
- v. The PBGs shall be valid for a period of 15 months from the date of issue of LoA for the REPP.
- vi. The PBG will be returned to the RPG immediately after successful commissioning of REPP, after considering any penalties due to delay in commissioning as per provisions stipulated in the Guidelines.

4.8. Connectivity

- **1.** The selected RPG will be responsible for laying of dedicated 11 or 33 kV, 66/11 kV or 110/11 kV, line from REPP to sub-station, construction of bay and related switchgear at sub-station where the plant is connected to the grid and metering is done.
- ii. RPG can get the transmission line constructed through DISCOM by paying the applicable cost and other charges. RPG will generally be responsible for maintaining this dedicated transmission line.
- In case more than one bidder are awarded projects to be connected to same Sub-station, they shall be permitted to co-ordinate with each other for setting up common transmission line for feeding to Sub-Station if they so desire and with the approval of DISCOM. The RPG shall comply with grid connectivity and other regulations as applicable.

4.9. Release of Performance Based Incentive (PBI) to DISCOM

- L DISCOM will be eligible to get Performance Based Incentive (PBI) @ Rs. 0.40 per unit purchased or Rs. 6.6 lakh per MW of capacity installed, whichever is less, for a period of five years from the COD.
- ii. To avail the PBI, DISCOM shall submit following documents after completion of one year from the COD and every year thereafter till five years:

(a) Timely payment of monthly lease rent, if applicable, to the land owner of the project.

(b) Monthly units purchased from the plant and corresponding payment note to the project developer.

- iii. Applicable PBI would be released to the DISCOM after submission of these documents by DISCOM to MNRE.
- iv. The DISCOMs can, if they desire so, pass on the PBI given to DISCOMs by the Central Government under this component, to the REPP owner to get more competitive tariff of RE Power.

4.10. Timeline and Penalty

- i. The selected RPG shall commission the solar power plant within 15 months from date of issuance of LoA.
- ii. The DISCOM is obliged to purchase power from the commissioned REPP even in case of early commissioning.
- iii. A duly constituted Committee of DISCOM officials will physically inspect the Plant in not more than 03 days from the date of receiving a call from the RPG and certify successful commissioning of the plant.
- iv. In case any RPG fails to achieve this milestone, DISCOM shall encash the Performance Bank Guarantee (PBG) in the following manner:
 - (a) Delay up to six months The PBG on per day basis and proportionate to the balance capacity not commissioned.
 - (b) In case the commissioning of the solar power plant is delayed over six months, the PPA capacity shall stand reduced / amended to the Project capacity commissioned at the end of six month from scheduled Commissioning Date.
- v. In case of delays of plant commissioning due to the reasons beyond the control of the RPG, DISCOM after having satisfied with documentary evidences produced by the RPG for the purpose, can extend the time for commissioning date without any financial implications to the RPG.

4.11. Quality Control, Monitoring and Maintenance

- i. In order to ensure only quality systems are installed, prevailing MNRE/BIS specifications and quality control orders applicable for solar modules, inverters, Balance of System (BOS) and other equipment shall be followed. These shall be monitored by State Implementing Agency (SIA) periodically.
- ii. The SIA would be responsible for ensuring Quality Control, proper maintenance and monitoring of REPPs.
- iii. SIA may assign technical consultants to assist farmers with the preparation of DPR and financial models who could also provide capacity building and guidance on the best practices of system installation and O&M.

4.12. Roles and responsibilities of stakeholders

4.12.1. Ministry of New and Renewable Energy

II.

MNRE will provide Procurement Based Incentive (PBI) to the DISCOMs @ 40 paise/kWh or Rs.6.60 lakhs/MW/year, whichever is lower, for buying solar/ other renewable power under this scheme.

The PBI will be given to the DISCOMs for a period of five years from the Commercial Operation Date of the plant. Therefore, the maximum PBI that shall be payable to DISCOMs will be Rs. 33 Lakh per MW for the entire period.

Page 5 of 26

4.12.2. DISCOMs

- i. The DISCOMs/SIAs shall aggregate the demand and send their proposal for sanction under the scheme along with details on their readiness to implement the Component A of the scheme, to MNRE.
- ii. Implementing Agencies will ensure transparency & objectivity in allocating capacity
- iii. The DISCOMs will ensure "must-run" status to the solar/other renewable power plants installed under this scheme and will keep the feeders 'ON' during sunshine hours of a day.
- iv. DISCOM shall act as facilitator to the beneficiaries in implementation of this scheme.
- v. In case RPG has taken land from a farmer/group of farmers on lease for the project, the amount of monthly lease rent would be paid by the DISCOMs to the lessor directly in his/her bank account before 5th day of the month following the month for which the lease rent is due. In such a case, the lease rent paid by the DISCOM will be deducted from monthly payment due to the RPG.

4.12.3. State Implementing Agency (SIA)

- i. SIA will coordinate with State/UT, DISCOM and farmers for implementation of the scheme.
- ii. SIA will assist the farmers in project development activities including formulation of DPR, PPA/EPC contracts, getting funds from financial institutions, etc.
- iii. SIAs will provide the handholding support required by farmers/developers, like availing financing from banks, etc. It will also coordinate with District Authorities for ease of implementation.
- iv. A State level Committee under the chairmanship of Principal Secretary (Renewable Energy/Energy) will be setup by the participating SIA of that State. The committee will be responsible for settlement of any issues arising during selection of solar/ other renewable energy-based power plants and their implementation.
- v. SIA will be responsible for coordinating/organizing the quarterly meetings of the State Level Committee.
- vi. SIAs shall ensure publicity of the scheme and create awareness through advertisements etc, and monitor the implementation of the scheme.
- vii. The SIA will be eligible to get service charge of Rs. 0.25 Lakh per MW after commissioning of the projects.

4.12.4. Renewable Power Generator (RPG)

- **L** RPG will be responsible for signing PPA and installing the plant as per provisions of scheme guidelines
- ii. RPG must adhere to applicable rules and regulations and make necessary compliances towards proper commissioning and operation of the plant during the PPA period.



5. Component B

5.1. Salient Features

- i. This Component is for installation of standalone Solar Pumps/replacement of diesel pumps by solar pumps.
- ii. Under this Component, individual farmers will be supported to install standalone Solar Agriculture pumps or replacement of existing diesel Agriculture pumps / irrigation systems in off-grid areas, where grid supply is not available.
- iii. Priority would be given to small and marginal farmers and farmers using micro irrigation.
- iv. Pumps of capacity higher than 7.5 HP may be allowed, but the CFA will be limited to the CFA applicable for pump of 7.5 HP.
- v. For the individual farmers in the North-Eastern region (NER); Hilly region (Jammu & Kashmir, Ladakh, Uttarakhand and Himachal Pradesh) and Islands (Andaman & Nicobar, Lakshadweep), the CFA will be available for pump capacity up to 15 HP, however the CFA for pumps up to 15 HP will be restricted to 10% of total installations
- vi. Maximum Solar PV capacity in kW is allowed as per the pump capacity in HP, in accordance with the MNRE specifications. For example, 3 HP pump capacity cannot have solar capacity of more than 3 kW.
- vii. DISCOMs/ Agricultural Department/ Minor Irrigation Department/ any other Department designated by State Government will be the implementing agencies for this component.
- viii. Agricultural Department may be prioritized as State Implementing Agency (SIA) by the State Governments given their extensive network of Agri and allied services along with embedded human resource availability.
- ix. Implementing agency will get service charges as applicable under Scheme Guidelines

5.2. Pre-Commissioning Activities

5.2.1. Demand Aggregation and Allocation of Solar Pumps

- i. The SIA will submit proposals to MNRE for approval.
- ii. The SIA after submitting proposal to MNRE may choose to start the preparatory activities including tendering process. However, the Letter of Award/Purchase Order shall be issued after the issuance of sanction letter by MNRE.
- iii. Based on the demand received from SIAs, MNRE will allocate quantity of pumps to the implementation agencies in the States after approval by a Screening Committee under the chairmanship of Secretary, MNRE.
- iv. MNRE will have the discretion to modify/cancel the allocated capacity based on the pace of progress in any particular state, or as per requirements of the scheme

5.2.2. Implementation

Component B can be implemented in two modes – with State share and without State share. The Central Government will provide financial assistance of 30% (or 50% for North Eastern Region/Hilly region/Islands) of the Benchmark Cost or the Tender Cost, whichever is lower, of the stand-alone solar Agriculture pump. The State Government will provide a subsidy of atleast 30%; and the remaining up to 40% is to be provided by bank loan/farmer.



In case the State Government is not in a position to give its share of subsidy of 30% but farmers are willing to set up solar water pumping system with the Central Financial Assistance only, the farmers are permitted to do so. Central share will continue to remain 30% (or 50% for NER/Hilly region/Islands), with the remaining 70% in General category states and 50% in Hilly region/NER/Islands being borne by the farmer. The Ministry will develop a National Portal to enable beneficiaries to apply for solar pump with central subsidy only.

5.2.3. Bidding Procedure

The States will call for tender for installation of solar water pumping system through the SIAs or any other agency nominated by the State Government. These agencies will carry out tendering process as per the guidelines, standards, & specifications laid down by MNRE. Any deviation shall normally be not permissible except in specific cases with the approval of Screening Committee in MNRE. The MNRE will list out the pumps/motors/panels and other equipments of high quality and specify that only those pumps/motor/panels which are in the approved list, can be used. The following categories, subject to meeting the technical and financial requirement of the tender, are allowed to participate in the tendering:

(a) System Integrators/ Aggregators/ Any other entities fulfilling technical and financial criteria.

(b) Any manufacturer of solar PV modules or manufacturer of solar pumps or manufacturer of solar pump controllers.

(c) Joint venture of any of manufacturers mentioned at (b) above with system integrators/aggregators.

If the vendor fails to install the allocated pumps within the stipulated period, the said vendor shall be debarred for five years' period from the date of such decision of debarment. Extension may be granted for valid reasons. In all cases, the successful bidder shall provide AMC for five years from the date of installation, real time monitoring, helpline, district level service centers and comply with standards of performance in dealing with complaints/redressal mechanism. It will be mandatory to use indigenously manufactured solar modules with indigenous solar cells. Further, the motor-pump-set, controller and balance of system should also be manufactured indigenously. The vendor has to declare the list of imported components used in the manufacturing of solar water pumping system.

5.2.4. Provision of USPC

Since pumps are generally used for 150 days in a year, the installed solar capacity can be optimally utilized by using Universal Solar Pump Controller (USPC), through which farmers can use solar power for other activities like operating chaff cutter, floor mill, cold storage, drier, etc., and increase his income. Option would be given to the farmer to opt for USPC and separate bid price for solar water pumping system with USPC will be invited and subsidy will be made available for these pumps according to benchmark price of solar pumps without USPC, even if the price discovered for solar pumps without USPC and use available for these pumps without USPC and use available for these pumps according to benchmark price of solar pumps without USPC.

than benchmark price. States/UTs may bear this additional cost to facilitate use of solar energy for other activities and increasing the income of farmers.

5.2.5. Water Saving Techniques

New Solar Agriculture pumps would not be covered under this component in Dark zones/black zones. However, existing standalone diesel pumps, can be converted into standalone solar pumps in these areas provided they use micro irrigation techniques to save water. In order to minimize the water usage for irrigation purpose, preference will be given to the farmers using Micro irrigation systems or covered under Micro irrigation schemes or who opt for micro irrigation system. The size of pump would be selected on the basis of water table in the area, land covered and quantity of water required for irrigation. For solar pumps to be set up and used by Water User Associations (WUA)/Farmer Producer Organizations (FPO)/Primary Agriculture Credit Societies (PACS) or for cluster-based irrigation system, the CFA will be allowed for solar pump capacity of higher than 7.5 HP considering up to 7.5 HP capacity for each individual in the group (for example when a group of 5 farmers in above mentioned category installs a pump of 40 HP capacity than the CFA up to 37.5 HP will be given to the group).

5.3. Post Commissioning Activities

5.3.1. Central Financial Assistance

MNRE will release the CFA towards the installation of solar pumps under the scheme as per the Benchmark cost or tender cost, whichever is lower. MNRE will be issuing the benchmark cost for various capacities and models being implemented under the scheme, from time to time.

5.3.1.1. Eligible CFA with State share

- i. CFA of 30% of the benchmark cost or tender cost, whichever is lower, of a particular category/type of the stand-alone solar Agriculture pump will be provided. The State Government will give a subsidy of at-least 30%; and the remaining 40% will be provided by the farmer.
- ii. However, in North Eastern States, Sikkim, Jammu & Kashmir, Himachal Pradesh and Uttarakhand, Lakshadweep and A&N Islands, CFA of 50% of the benchmark or tender cost whichever is lower, of a particular category/type of the stand-alone solar Agriculture pump will be provided. The State Government will give a subsidy of at-least 30%; and the remaining at most 20% will be provided by the farmer.

5.3.1.2. Eligible CFA without State share

- i. CFA of 30% of the benchmark cost or tender cost whichever is lower, of a particular category/type of the stand-alone solar Agriculture pump will be provided. The remaining 70% will be provided by the farmer either directly or through loan.
 - In North Eastern States, Sikkim, Jammu & Kashmir, Himachal Pradesh and Uttarakhand, Lakshadweep and A&N Islands, CFA of 50% of the benchmark cost or tender cost



ij.

whichever is lower, of a particular category/type of the stand-alone solar Agriculture pump will be provided. The remaining 50% will be provided by the farmer.

III. MNRE will also develop an online portal where farmer can directly register themselves and avail transfer of subsidy in their account after successful completion of installation. Subsidy in this case will be restricted to 30% of Benchmark Cost or Tender Cost, relevant to the State, whichever is lower.

5.3.2. Access to Financing

- i. Bank finance may be made available for farmer's contribution, so that farmer has to initially pay only 10% of the cost and the remaining amount (other than subsidy) can come from loan.
- ii. SIAs will make special efforts to organize events like *Bank Mela* at block and district levels and sensitize the banks to the need of farmers for accessibility of finance under the scheme.

5.3.3. Installation timeline and Penalties

Projects for installation of Solar Agriculture pumps systems shall be completed within 24 months from the date of sanction by MNRE. Extension in project completion timelines, up to a maximum period of three months, will be considered at the level of Group Head in MNRE and up to 6 months or beyond at the level of Screening Committee in MNRE on submission of valid reasons by the implementing agency.

5.3.4. Release of funds by MNRE

- i. Funds up to 30% of the applicable CFA for the sanctioned quantity would be released as advance to the implementing agency after placement of letter of award(s) to the selected vendors.
- ii. Second instalment up to 60% of the applicable CFA would be released on submission of the provisional UCs of the first release and the details of partial installation of pumps.
- iii. The last installment of balance eligible CFA up to 10% along with applicable service charges would be released on acceptance of the Project Completion Report in the prescribed format, Utilization Certificates as per GFR and other related documents by the Ministry.
- iv. The implementing agencies may pass on these funds to the selected vendors in different stages on achievement of various milestones as per terms and conditions of letter of award(s).
- v. 2% of the eligible CFA will be provided as service charges in totality to all the agencies involved in implementation including the designated SIA. Part of service charges (to be decided by MNRE) shall be given to the central agency for centralised tendering.
- vi. MNRE may release 50% of eligible service charges for the sanctioned quantity to State Implementation Agencies after placement of LoA for preparatory activities. Additionally, MNRE will retain 33% of eligible service charges for nation-wide Information, Education and Communication (IEC) activities.
- vii. The CFA in case of online portal, as per Para 5.2.2 (iv), will be released after installation of pump.



5.4. Monitoring and maintenance

- **i.** Selected vendors shall be responsible for design, supply, installation and commissioning of solar agriculture pumps. Vendors will mandatorily provide AMC for a period of 5 years from the date of commissioning of the systems including insurance coverage for the installed systems against natural calamities and theft.
- ii. AMC will include inspection by Vendor at least once in a quarter and submission of quarterly inspection report of the installed pumps as per prescribed format. To ensure timely maintenance of the systems the vendor shall have one authorized service centre in each operational district or cluster of districts as decided by the SNA depending upon number of installed pumps and a helpline in local language should be made available. The helpline number shall be indicated on the pump/ controller at suitable location easily visible to the user.
- iii. All solar agriculture pumps sanctioned under the Programme shall be provided with remote monitoring system by the vendor. It will be mandatory to submit performance data of solar power plant online to MNRE in the manner and format prescribed by MNRE.
- iv. Monitoring of the Scheme and its implementation will be carried out during the period of implementation of the Scheme as given below:
 - a) The implementing agency would be responsible for monitoring include timely installation and functioning of the pumps. The progress report of its installation will be submitted to the MNRE by the SIAs every month.
 - b) Funds may be released by implementing agency to the vendor on submission of bank guarantee as specified in the tender by that vendor for a period of five years. Alternatively, BG may be provided initially for a period of one year which may be extended on year to year basis thereafter.
 - c) The Ministry officials/ designated agency may inspect the ongoing installation or installed plants.
- v. In case the installed systems are not as per standard, non-functional on account of poor quality of installation, or non-compliance of AMC, the Ministry/State Government reserves the right to debar the bidder/vendor and any of its successors up to a period of five years from the date of such decision of debarment.

5.5. Responsibilities of Implementation Agency

The Implementing Agencies will be responsible for the following activities:

- (i) Carry out wide publicity of the scheme to increase awareness, identify potential beneficiaries, and coordinate with banks for easy access of finances.
- (ii) Ascertain the demand for solar pumps in the state, prepare beneficiary list and submit to MNRE for allocation of demand.
- (iii) Carry-out the state bids/tenders for installation and maintenance of solar pumps sanctioned by MNRE under Component-'B'.
- (iv) Ensure project completion within the given timelines and compliance of MNRE guidelines, and standards.



- (v) Oversee installation of solar pumps, ensure timely O&M, and report periodical progress to MNRE (this will include but is not limited to online submission of monthly and quarterly progress reports).
- (vi) Inspection of installed systems and online submission of completion reports to MNRE.
- (vii) Submission of utilization certifications and audited statement of expenditure through EAT/PFMS module and disbursement of MNRE CFA.
- (viii)Maintenance of updated records and make them available for inspection at the call of MNRE.
- (ix) Maintain real time monitoring data of the installed systems through dedicated webportal.
- (x) Ensure the effective working of the Remote Monitoring System of the installed pumps, and conduct periodical inspection of the same.
- (xi) Submission of accurate, verified, and complete records in all aspects to MNRE for ensuring the timely payment of CFA.
- (xii) Performance monitoring of installed system through third party.
- (xiii)Ensure compliance of AMC by the vendors.



6. Component C

6.1. Salient Features

- 1. This Component aims at solarization of grid connected agriculture pumps including Feeder Level Solarization
- ii. The objective of this component is to provide reliable day-time solar power to farmers, enhancing their income by purchasing surplus solar power and thus incentivizing them for saving water
- iii. Under Individual Pump Solarization (IPS), solar PV capacity up to two times of pump capacity in kW is allowed, so that the farmer will be able to use the generated solar power to meet the irrigation needs and get additional income by selling surplus solar power to DISCOMs.
- iv. Under Feeder Level Solarization (FLS), States are supported for solarization of agricutlural feeder or mixed feeders.
- v. DISCOMs /GENCO/ any other Department designated by State Government will be the implementing agencies
- vi. State Implementing agencies will submit proposals to MNRE for approval.
- vii. In case of dark zones/black zones only existing grid connected pumps will be solarized.
- viii. Implementing agency will get service charges as applicable under Scheme Guidelines.

6.2. Demand Aggregation and allocation of quantities

- i. State Implementing agencies will submit proposals to MNRE for approval.
- ii. Based on demand received from SIAs, MNRE will allocate quantity of pumps to the implementation agencies in the States after approval by a Screening Committee under the chairmanship of Secretary, MNRE.
- iii. The SIA after submitting proposal to MNRE may choose to start the preparatory activities including tendering process. However, the Letter of Award/Purchase Order shall be issue after the issuance of sanction letter by MNRE.
- iv. MNRE will have the discretion to modify/cancel the allocated capacity based on the pace of progress in any particular state, or as per requirements of the scheme

6.3. Individual Pump Solarization (IPS)

6.3.1. Options for solarisation under Component C (IPS)

Followings are the two options through which grid connected pumps can be solarized; State may choose any other option, as well.

- i. **Pump to run on both Solar and Grid:** In this case the agriculture pump will continue to run at rated capacity taking power from solar panels and balance power from grid, if required, and in case solar power generation is higher than required by pump, the additional solar power would be fed to the grid. Net metering may be provided in this case.
 - a) Farmer would be able to import and export power to the grid at the rate specified by the SERC.



- b) DISCOM may introduce different drawal and injection tariff rates so that farmers are discouraged to draw power from the grid during the peak hours. Further, Time of Day (ToD) tariff is also an alternative to efficiently manage demand and supply on the solarised feeders.
- ii. **Pump to run on solar power only**: In this case the pump will only run on the solar power as in case of stand-alone solar pump and no power will be drawn from the grid for operation of pump.
 - a) The motor pump set will have to be replaced with AC/DC solar water pumping system as per MNRE specification for standalone solar pump and when the pump is not running the solar power can be fed into the grid through suitable grid-tied inverter.
 - b) The total solar PV capacity including capacity additionally allowed under the Scheme can be bifurcated into two portions, one portion of SPV capacity as per MNRE specifications required for the solar pumps of given HP, which will either run the pump or feed power to the grid when the pump is not running; and other portion of SPV capacity out of the total allowed by the State will continuously feed power to the grid during sunshine hours.

6.3.2. Selection of Feeders and load survey

- i. Feeder-wise implementation is to be carried out under this component for solarisation of grid-connected agricultural pumps.
- Feeders may be selected on the basis of load, technical and commercial losses, number of consumers, etc. Efforts should be made to solarise all agriculture pumps in a feeder, however, SIAs may impose a minimum solarisation requirement for a feeder in terms of minimum % of pumps solarized on that feeder.
- iii. GPS Survey should be carried out on eligible feeders to correct feeder coding and ensure accurate feeder loads and distribution loss calculations.
- iv. Farmers may be given option to declare his actual connected load or alternatively the SIAs may carry out survey to get details on actually connected load so that solar PV panels of required capacity is provided to the participating farmers.
- v. SIAs may also assess the actual requirement of pump capacity based on water requirement and water table

6.3.3. System Specifications

- i. MNRE will provide updated specifications (as amended from time to time) for solar water pumping system, solar modules, MMS and other balance of system.
- ii. The vendors shall provide only such pumps/panels and other equipments which are BIS certified. MNRE may also draw a list of high quality solar pumps/panels and in such a case, vendors will supply only such models which are in the approved list.



iii. It will be mandatory to use indigenously manufactured solar panels with indigenous solar cells and modules. Further, inverter/controllers and balance of system should also be manufactured indigenously. The vendor has to declare the list of imported components used manufacturing of equipment used in the solarisation system.

6.3.4. Bidding Procedure

- 1. There will be State tendering for solar water pumping system through the SIAs or any other agency nominated by the State Government. These agencies will carry out tendering process as per the guidelines, standards, & specifications issued by MNRE from time to time. Any deviation shall normally be not permissible except in specific cases with the approval of Screening Committee, MNRE.
- ii. The vendors shall use only BIS certified equipment and systems. MNRE may also draw a list of certified models and manufacturers and in that case the equipment will be from that list.
- iii. The following categories subject to meeting the technical and financial requirement of the tender are allowed to participate in the tendering:

(a) System Integrators/ Aggregators/ Any other entities fulfilling technical and financial criteria

(b) Any manufacturer of solar PV modules or manufacturer of solar pumps or manufacturer of solar pump controllers using indigenous technology.

(c) Joint venture of any of manufacturers mentioned at (b) above with system integrators/aggregators.

- iv. Selected vendors shall be responsible for design, supply, installation and commissioning of solar agriculture pumps.
- v. If the vendor upon empanelment fails to install the allocated pumps within the stipulated period, the said vendor shall be debarred for five years' period from the date of such decision of debarment.
- vi. It will be mandatory to use indigenously manufactured solar panels with indigenous solar cells and modules. Further, the motor-pump-set, controller and balance of system should also be manufactured indigenously. The vendor has to declare the list of imported components used in the manufacturing of solar water pumping system.
- vii. Thorough maintenance of selected agriculture feeders is required to maintain feeder availability during sunshine hours. This includes maintenance of 33/11 kV sub-station, 33 kV, 11 kV and LT lines and distribution transformers, etc., on regular basis in a time bound manner. Proper protection system including improved earthing of equipment shall be provided considering possibility of high voltage/current due to multiple generating sources in the feeder selected for solarisation.

6.3.5. Eligible CFA under Component C (IPS)

i. The CFA will be limited to Solar PV capacity up to two times of pump capacity in kW for pumps up to 7.5 HP. Solarisation of Pumps of capacity higher than 7.5 HP is also allowed, however, the CFA in such cases would be limited to the CFA applicable for pump of 7.5 HP in the respective State/UTs.

- ii. The solar pumps installed by the individual farmers in the North-Eastern States; UTs of Jammu & Kashmir and Ladakh; and the state of Uttarakhand and Himachal Pradesh the CFA will be available for pump capacity up to 15 HP, however the CFA for pumps up to 15 HP will be restricted to 10% of total installations
- III. CFA up to 30% (upto 50% of NER/Hilly/Islands) of the cost of solarisation will be provided for solar PV component including solar modules, module mounting structure, controller/inverter, balance of system, installation & commissioning, five Year CMC, insurance, etc., on basis of benchmark cost or cost discovered through tender whichever is less. The States are free to provide additional CFA to reduce the beneficiary farmer contribution.
- iv. There is no CFA for replacement of existing less efficient grid connected agricultural pump with energy efficient agricultural pump, however, the states are free to provide additional financial support for replacement of existing pumps with energy efficient pumps.
- v. It is also permitted to implement Component C(IPS) without state subsidy. If any State Government does not provide for subsidy by the State, the farmer may choose and implement the solarization with only Central Government subsidy and his contribution and loan from the banks. The banks may provide loans of up to 70% of the cost. From such sale of excess power from the Solar PV project, the additional income should be paid to the bank towards repayment of loan in addition to the regular installment by the farmer.
- vi. All the modalities and conditions for the release of CFA under Component C (IPS) will remain same as Component B, as provided in Para 5.5.

6.3.6. Monitoring and maintenance

- i. Vendors will mandatorily provide AMC for a period of 5 years from the date of commissioning of the systems including insurance coverage for the installed systems against natural calamities and theft.
- ii. AMC will include inspection by Vendor at least once in a quarter and submission of quarterly inspection report of the installed pumps as per prescribed format. To ensure timely maintenance of the systems the vendor shall have one authorized service centre in each operational district or cluster of districts as decided by the SNA depending upon number of installed pumps and a helpline in local language should be made available. The helpline number shall be indicated on the pump/ controller at suitable location easily visible to the user.
- iii. All solar Agriculture pumps sanctioned under the Programme shall be provided with remote monitoring system by the vendor. It will be mandatory to submit performance data of solar power plant online to MNRE in the manner and format prescribed by MNRE. The remote monitoring system may consist of smart meters, communication hardware/ IoT devices, software interface, web and mobile application and internet connection.
- iv. Monitoring of the Scheme and its implementation will be carried out during the period.

- a) The implementing agency would be responsible for monitoring the installation of the solar pumps and their functioning.
- b) Implementing agencies will submit monthly progress report for the sanctioned projects.
- c) Funds may be released by implementing agency to the vendor on submission of bank guarantee as specified in the tender by that vendor for a period of five years. Alternatively, BG may be provided initially for a period of one year which may be extended on year to year basis thereafter.
- d) The Ministry officials or designated agency may inspect the ongoing installation or installed plants.
- v. In case the installed systems are not as per standard, non-functional on account of poor quality of installation, or non-compliance of AMC, the Ministry and the State Government reserves the right to debar the bidder/vendor and any of its successors up to a period of five years from the date of such decision of debarment.
- vi. Possibilities would be explored by States for convergence of present scheme with schemes on promotion of micro irrigation system and replacement of agriculture pumps with energy efficient pumps and they may work out the modalities.

6.4. Implementation of Feeder Level Solarization

For the implementation of Feeder Level Solarization (FLS), the DISCOMs need to evaluate the agricultural load, segregate the feeder and solarize it. The States may also choose to offer virtual segregation using feeder level (Internet of Things) IoT. However, segregation would not be precondition for the solarization of mixed feeders. This component will be applicable to farmers already connected to grid. The feeder will be solarised, however, States may impose a minimum requirement of pumps on a feeder. The detailed framework for feeder level solarization will be as follows:

6.4.1. Execution Methodology

AD DIST

rpc@D

- i. The SNA /DISCOM/ SIA, will be the implementing agency for feeder level solarisation in their respective areas.
- ii. State Government may choose to appoint any other expert agency to help these implementing agencies for tendering and other related activities of installation of solar power plant for feeder level solarisation.
- Where agriculture feeders have already been separated, those feeders may be solarised under the scheme.
- iv. Where the agriculture feeders are not separated, feeders having major load for agriculture may also be considered for solarisation under the Scheme. The requirement of total annual power for an agriculture feeder will be assessed and a solar power plant of capacity that can cater to the requirement of annual power for that feeder can be installed through CAPEX/RESCO mode, which will supply solar power to that feeder.

Feeder level solar power plant may be installed to cater to the requirement of power for single feeder or for multiple agriculture feeders emanating from a distribution substation (DSS) to feed power at 11 kV or at the higher voltage level side of the DSS. There is no cap of the capacity of solar power plant for feeder level solarisation. The DISCOMs may identify land near DSS, get ownership of land or its lease rights, provide connectivity at DSS and lay sub-transmission line between DSS and solar power plant.

vi. There could be situations where for distribution substation, the solar power plant can be installed either under Component A or under FLS of Component C. In such situations, DISCOMS while bidding for installation of solar power plants under FLS of Component C, the qualification requirement may be kept at par with the guidelines of Component A. This will allow all the eligible participants under Component A to also participate under Component C (FLS).

6.4.2. Simplified step by step procedure

- i. Identifying the substation: State Nodal Agencies/Implementing Agencies/DISCOM will identify substations having downstream agriculture load along with the number of pumps connected to that substation. The annual power requirement for the agriculture feeders and the capacity of solar power plant that can cater to the power requirement for that agriculture feeder may be assessed. Preference may be given to the substations with 11/33 kV bus bars. The identification will include an assessment of the agriculture load served downstream.
- ii. Land for setting up the solar plant: If there is no Government land near the substation to set up solar plant, theland may be taken on lease from the farmers. Land lease rates would be announced by the state government. Currently, some State offer lease rates which range from Rs 25,000 to Rs 50,000 per acre per year or 6% of ready reckoner rate, whichever is higher, with a yearly 3% increment of the base rate, and Rs 1/hectare/year for public land for a period of 30 years. The States may decide similar competitive provisions. The State Government will publicise these rates along with the list of substations and request farmers to offer their land on lease in the vicinity of the substations at the offered lease rent.
- iii. Development of Online portal: A dedicated online portal can be developed by the States/SIA/DISCOM which will serve as centralized portal for land aggregation. The state should have the details of identified substations. The portal may be developed by State/SIA/DISCOM. The States/SIA/DISCOMs will advertise their intention to take land on lease from the farmers in the vicinity of the indetified substation along with a declaration of land lease rates. The farmers who wish to offer their land near the substation on lease may apply directly on the portal. They may also apply to the Tehsildar/District Magistrates. It would be the responsibility of the District Collector/District Magistrates to ensure that the offline applications received at their end from the framers to lease their land for the installation of solar power plants, are shared with DISCOM/SIAs. The DISCOM/SIA would update the same on online portal on regular basis.



iv. Modalities for leasing

- a) <u>Public Land</u>: The State may facilitate by providing unutilized land on lease at a pre-decided annual rent. The State may also include their unutilized water bodies for the development of the projects.
- b) <u>Private Land:</u> All the private land owners who are interested in providing their land for projects may be encouraged to offer their land through the above-mentioned portal. The revenue authority should also verify the authenticity of the land through the respective District Collectors/Magistrates. In order to protect the interest of land owner, the State may decide the amount of lease rent for the land along with annual escalation provisions.

Considering the land required for 1 MW is around 4-5 acres, contiguous land of a minimum of 5 acres and nearer to the substation would be given the preference.

- v. **Provision of SPV:** The DISCOMs may form an SPV which can implement the project along with arrangements for leasing the land etc.
- vi. Lessee and Lessor Agreement: The Nodal Agency/DISCOM/SPV may also become the single entity to sign land lease agreements with all landowners. In such cases, the individual land owner need not interact with SPD and face uncertainty in timely lease payment. The nodal agency should sign an agreement with individual land owner and a tripartite agreement between DISCOM, Nodal Agency and Solar Power Developers may be worked out where payments are transferred to nodal agency for the purpose of lease rent. Nodal agency will inform the District Collector/District Magistrate from time to time, about the requirement of land and get in touch with the farmers to lease their land for the purpose.
- vii. Selection of Solar Power Developers (SPD): Since the scale of an individual solar project with substation is relatively small, therefore in order to get the benefit of economy of scale, multiple substations may be grouped as one bidding group. The SPD may be permitted to quote the tariff common for all projects under one group and SPD may sign a single PPA with DISCOM.
- viii. For the project under this mechanism, the States are free to choose Renewable EnergyImplementing Agencies of Government of India to act as an intermediatory procurer.

6.4.3. Estimation of Plant Capacity and eligible CFA under FLS

- For the purpose of calculating Solar Plant Capacity and eligible CFA, the following appoach would be taken:
 - 1) The existing load of the feeder will be assessed and a solar power plant of capacity that can cater to the requirement of annual power for that agriculture feeder can be installed either through CAPEX mode or RESCO mode, which will supply solar



power to that feeder. For example, a feeder having annual power requirement of say 10 lakh units, the power can be supplied by solar power plant of capacity around 600 kW with an annual CUF of 19%.

- 2) Average of consumption of last three years shall be considered for calculating the solar plant capacity as per calculations in Para 1) above
- 3) Mixed feeders can also be solarized, however, MNRE CFA will be available based on capacity required for agriculture consumption.
- 4) Solarisation of pumps of any capacity is allowed, however, in case of pumps of capacity above 7.5 HP, the CFA will be limited to solar capacity for 7.5 HP pumps.
- 5) The solar pumps installed for feeder solarization in the North-Eastern States; UTs of Jammu & Kashmir and Ladakh; and the state of Uttarakhand and Himachal Pradesh the CFA will be available for pump capacity up to 15 HP, however the CFA for pumps up to 15 HP will be restricted to 10% of total installations
- 6) For calculating consumption for pumps of capacity higher than 7.5 HP or 15 HP, as the case may be, average consumption of 7.5 HP/15 HP pumps on the same feeder may be considered. This average consumption may be used in estimating consumption for all pumps of capacity higher than 7.5 HP/15 HP.
- 7) The state may choose to install feeder level solar power plant of capacity higher than capacity required for supplying power to agriculture feeder. However, in this case CFA will be limited for solar capacity required for supplying power to the agriculture feeder.
- Since the projects are to be commissioned near the point of consumption, distribution losses would be reduced, therefore, losses shall not be considered in plant capacity estimation.
- 9) CUF shall be taken as per the solar insolation on site, however, annual CUF of minimum 19% shall be considered for calculation of plant capacity eligible for CFA. No restrictions to be made on installing DC capacity higher than total contracted AC capacity of the plant to help achieve higher CUF.
- In case of unmetered connections, following methods may be followed for calculation of consumption:
 - a. Indexation approved by the State Electricity Regulatory Commission may be followed.
 - b. If both metered and unmetered connections are there on the same feeder, capacitywise average of consumption of the metered connections may be extended to the unmetered connections. This shall be cross-checked with the feeder level meter.
 - c. Consumption on feeder level meters may be used for estimation of plant capacity. However, if pumps of capacity higher than 7.5 HP/15 HP are there on such feeder, capacity may be calculated based on consumption reduced in proportion of the load if capacity is capped at 7.5 HP/15 HP and actual load on the feeder.
- ii. The cost of installation of solar power plant for the purpose of CFA, will be notified by MNRE from time to time. The eligible CFA would be 30% of the cost of installation of the solar power plant of capacity needed to meet the average consumption of the pumps connected on the feeder(s), as per calculations in Para (i) above.



6.4.4. Implementation under CAPEX mode and eligible CFA

- i. FLS can be implemented in CAPEX mode. The process of tendering and signing of work agreement with EPC contractor should normally be completed by SIA within six months from the date of issuance of sanction by MNRE.
- ii. For installation of feeder level solar power plant CFA of 30% (50% in case of NE States, hilly states/UTs and Island UTs) will be provided by central Government and balance will be met through loan from NABARD/PFC/REC.
- iii. Concessional financing will be available for solarisation of agriculture pumps as RBI has already included this component under priority sector lending and MoAFW has included community level solarisation under Agriculture Infrastructure Fund.
- iv. The current outlay on subsidy being presently provided for supply of electricity to agriculture pumps by State Government can be used to repay the loan in five to six years after which power will be available free of cost and outflow from State Government's exchequer on account of electricity subsidy for agriculture will come to an end.
- v. On an average power for agriculture will be required only for 150 days in a year, the electricity produced from the solar power plant in the remaining days will possibly provide an additional income to the DISCOM. If this is also used to pay off the loans taken from NABARD/PFC/REC, the loan can be repaid sooner.
- vi. The DISCOM may carry-out operation and maintenance of the solar power plant. Alternatively, the EPC contractor who install the solar power plant may also be given task for O&M of plant and supply guaranteed solar power for 25 years.
- vii. Payment for O&M of solar plant can be linked with energy production. An undertaking to this effect will be submitted by DISCOM to MNRE.

6.4.5. Implementation under RESCO mode and eligible CFA

- i. For selection of RESCO developer and PPA, the Guidelines and Model PPA issued by MNRE for implementation of Component-A of PM-KUSUM Scheme may be used, with suitable modifications.
- The maximum timeline allowed for commissioning of solar power plant by RESCO Developer will be 12 months from date of issuance of LoA. However, the implementing agency may increase this time on case to case basis, up to the sanctioned timeline given by MNRE.
- iii. The process of selection of RESCO Developer and signing of PPA should normally be completed by implementing agency within six months from the date of issuance of sanction by MNRE.
- iv. A suitable clause on PPA shall be included to ensure that RESCO developer does not sell solar power to any other buyer except the concerned DISCOM during tenure of PPA, including penal provision calling for refund of proportionate CFA granted to the project by MNRE on violation of this condition.

v. For installation of feeder level solar power plants through RESCO model, the developers

record of 25 years after taking into account the subsidy being given by Govt. of India.

- vi. In case of any liquidation of assets of the solar power plant prior to completion of PPA period, the first charge shall be towards recovery of proportionate CFA granted to the project by MNRE.
- vii. The developer will get CFA @ 30% of the estimated cost of installation of solar power plant.
- viii. The solar power supplied by RESCO developer would much cheaper than present cost of power delivered at distribution sub-station and therefore, DISCOM will save the amount equal to difference between the two.
- ix. In the RESCO model the burden of electricity subsidy for agriculture will be reduced to the extent of difference mentioned above and not become zero as in case of CAPEX model, where once the loan is repaid, subsidy support from state Government is no longer required.
- x. States may choose to provide upfront subsidy in lieu of electricity subsidy being given to agriculture consumers. This upfront subsidy from state could be in the form of VGF to RESCO developer, in addition to CFA, to supply power to farmers of an agriculture feeder at present subsidised rates or any other rate fixed by state Government.

6.4.6. System specifications and Quality Control

- i. All components used for installation of solar power plants shall confirm to applicable BIS/MNRE specifications and follow quality control guidelines issued by MNRE. It will be mandatory to use indigenously manufactured solar panels with indigenous solar cells.
- ii. Further, the balance of system should also be manufactured indigenously. The vendor must declare the list of imported components used in the solarisation system.
- iii. MNRE will draw up a list of approved models and manufacturers after carrying the quality checks. When such a list is prepared only that equipment shall be allowed which is within the approved list.
- iv. To ensure the quality, inspection shall be carried out of the solar modules, inverter/controller, MMS, etc., during the installation of system and final commissioning of the system. Officers involved in inspection should be domain experts, properly trained and equipped with necessary tools for inspection. The SIA may engage a third-party inspection agency for this purpose.
- v. Thorough maintenance of selected agriculture feeders is required to maintain feeder availability during sunshine hours. This includes maintenance of DSS, subtransmission/LT lines and distribution transformers, etc., on regular basis in a time bound manner.

6.4.7. Monitoring and maintenance

- i. Selected vendors shall be responsible for all aspects of solarisation viz., design, supply, installation, and commissioning.
- ii. Vendors will mandatorily provide AMC for a period of 5 years from the date of commissioning of the systems including insurance coverage for the installed systems against natural calamities and theft.



- iii. AMC will include submission of quarterly inspection report of the installation as per prescribed format. To ensure timely maintenance of the systems the vendor shall have one authorized service center in each operational district and a helpline in local language in each operational State.
- iv. State may choose to install watchdog transformer and devices to regulate power supply and monitor non-participating connections on the feeder concerned.
- v. MNRE has developed a central monitoring portal which will extract data from the State portals for monitoring of the scheme and analyze data generated, which is in turn be used in scheme refinement and taking measures to attract more farmers to the scheme and induce required behavioral changes.
- vi. It will be mandatory for DISCOMs to monitor solar power generation and performance of the solar power plant through online system. The online data will be integrated with central monitoring portal which will extract data from the State portals for monitoring of the scheme.
- vii. SIA may specify a minimum guaranteed generation during a year from the solar system installed for a specified period and provision of compensation in case of not achieving the same.
- viii. Monitoring of the Scheme and its implementation will be carried out during the period of implementation of the Scheme as given below:
 - a) The implementing agency would be responsible for monitoring parameters such as end-use verification and compilation of statistical information.
 - b) Implementing agencies will submit monthly progress report for the sanctioned projects.
 - c) Funds may be released by implementing agency to the vendor on submission of bank guarantee as specified in the tender by that vendor for a period of five years. Alternatively, BG may be provided initially for a period of one year which may be extended on year to year basis thereafter, up to 5 years.
- ix. The Ministry officials or designated agency may inspect the ongoing installation or installed plants. In case the installed systems are not as per standard, non-functional on account of poor quality of installation, or non-compliance of AMC, the Ministry reserves the right to debar the bidder/vendor and any of its successors up to a period of five years from the date of such decision of debarment.
- x. In case the system is installed by the RESCO operator does not work for 25 years, the entire plant will be seized and auctioned by the DISCOM/SIA to recover the CFA.

6.4.8. Release of funds for Component C (FLS)

- i. The CFA up to 30% (upto 50% of NER/Hilly/Islands) of the cost of solarisation will be provided for the projects installed under Component C (FLS).
- ii. In case of installation under CAPEX mode, the SIAs are eligible for advance upto 40% of eligible CFA after completion of tendering process and signing of work agreement with EPC contractor selected for installation of solar power plant. The balance up to 60% eligible CFA would be released at the end of commissioning of plant where plant starts supplying power to agriculture feeder(s) and on submission of PCR and other related

documents, as prescribed by MNRE.

- iii. Under the RESCO mode, CFA upto 100% of the total eligible CFA will be released to the RESCO developer through DISCOM on successful operation and performance of the solar plant for two months after the commissioning, with at least one month CUF as per minimum CUF agreed in PPA. The SIA need to submit PCR and other related documents, as prescribed by MNRE.
- iv. 2% of the eligible CFA will be provided as service charges in totality to all the agencies involved in implementation including the designated SIA. Part of service charges (to be decided by MNRE) shall be given to the central agency for centralised tendering.
- v. The MNRE will retain 33% of eligible service charges for national wide IEC activities. MNRE may release 50% of eligible service charges for the sanctioned quantities to SIAs after implementation of LOAs for preparatory activities.

6.5. Installation timeline and Penalties under Component C (IPS & FLS)

- i. Projects for installation of solar agriculture pumps and feeder level solarization under Component C shall be completed within 24 months from the date of sanction by MNRE.
- ii. Extension in project completion timelines on valid grounds, up to a maximum period of six months, will be considered at the level of Joint Secretary in MNRE. Extension beyond six months can be given by Secretary, MNRE for valid reasons.

6.6. Water saving and enhancing farmers' income

- i. Component C will enable farmers to access daytime reliable solar power for irrigation. Therefore, farmers can be incentivized for saving water and enhancing their income.
- ii. The DISCOMs shall assess the average power requirement by farmers of an area depending upon various factors. This power requirement will be treated as their benchmark consumption. The DISCOMs shall incentivize farmers for consuming power less than benchmark consumption.
- iii. Such saving of power shall be treated as surplus power injected by farmers and they will be paid by DISCOMs against this saved power at pre-determined tariff. The DISCOM will inform the farmers about the benchmark consumption and amount incentive per unit which they can avail of, in case of lower consumption. This will be an important measure for conserving groundwater level.
- iv. Some of the states have indicated that for a feeder being solarised not all farmers may be willing to install meters necessary for availing the said incentives under the scheme. Therefore, it will be the farmers choice to opt or not for this facility.
- v. In order to minimize the water usage for irrigation purpose, preference will be given to the farmers using Micro irrigation systems or covered under Micro irrigation schemes or who opt for Micro irrigation systems.



6.7. Responsibilities of Implementation Agency

The Implementing Agencies will be responsible for the following activities:

- L Carry out publicity of the scheme, in consultation with MNRE to increase awareness amongst potential beneficiaries.
- ii. Coordinate with Banks/FIs to provide the loan facilities for the required farmer share.
- iii. Prepare proposals and submit to MNRE for sanction
- iv. Conduct tendering process as per MNRE guidelines
- v. Issue additional instructions/ conditions such as minimum solarisation level of feeder for IPS.
- vi. Issue connectivity standards/regulations, if required, and facilitate connection to the grid for FLS.
- vii. Selection of feeder for solarisation and demand aggregation for solarisation of pumps under FLS.
- viii. Encourage Feeder wise committee of farmers to coordinate amongst the participating members and other agencies involved in implementation of the Scheme.
- ix. Selection of beneficiaries and implementation of scheme
- x. Oversee installation of systems.
- xi. Inspection of installed systems and online submission of completion reports to MNRE.
- xii. Disbursement of MNRE CFA and submission of utilization certificates and audited statement of expenditure through EAT/PFMS module.
- xiii. Online submission of monthly and quarterly progress reports.
- xiv. Ensure project completion within the given timelines and compliance of MNRE Guidelines and Standards.
- xv. Online and offline maintenance for records.
- xvi. Real time monitoring through dedicated web-portal
- xvii. Performance monitoring of installed system through third party
- xviii. Ensure compliance of AMC and training of locals by the vendors.
- xix. Any other activity to ensure successful implementation of the programme.

7. Installation of Innovative Standalone Solar Pumps

The scheme permits installation of innovative stand-alone solar pumps (under Component B and C) in test mode. The guidelines for installation of innovative stand-alone solar pumps are placed at Annexure III. These guidelines are applicable for all Indian innovators/ manufacturers/ service providers, who wish to install innovative stand-alone solar pumps in the country under this scheme.

8. Interpretation of the Guidelines

In case of any ambiguity in interpretation of any of the provisions of these guidelines, the decision of the Ministry of New & Renewable Energy shall be final.

The Guidelines may be reviewed by the Ministry from time to time and necessary modifications made with approval of Minister, New & Renewable Energy.



Page 25 of 26

Glossary and Definitions

i. Ministry (Ministry of New & Renewable Energy)

Nodal Ministry of the Government of India for all matters relating to new and renewable energy to deploy new and renewable energy to supplement the energy requirements of the country and is reffered to as Ministy in these guidelines.

ii. Renewable Power Generator (RPG)

Solar or other Renewable Energy based Power Plants (REPP) of capacity up to 2 MW will be setup by individual farmers/ group of farmers/ cooperatives/ panchayats/ Farmer Producer Organizations (FPO)/Water User associations (WUA) hereinafter called Renewable Power Generator (RPG).

iii. Renewable Energy based Power Plant (REPP)

Renewable Energy based Power Plants to be commissioned under PM KUSUM

iv. Renewable Power Obligation

Under Section 86(1) (e) of the Electricity Act 2003 ("EA 2003") and the National Tariff Policy 2006, Renewable purchase obligation (RPO), is a mechanism by which the obligated entities are obliged to purchase certain percentage of electricity from Renewable Energy sources, as a percentage of the total consumption of electricity.

v. Commercial Operation Date

The Commercial Operation Date (COD) shall be considered as the actual date of commissioning of the solar power plant as declared by the Commissioning Committee.

vi. Expression of Interest

DISCOM may also invite applications for Expression of Interest(EOI) from prospective RPGs to identify the interested beneficiaries.

vii. Power Purchase Agreement

A electricity purchase agreement signed by the DISCOM with the qualified RPG for purchasing the electricity for a period of 25 years at the pre-determined/transparently determined Tariff.

viii. Letter of Award

Letter of Award (LoA) will be awarded to all eligible applicants for procurement of renewable power at a pre-fixed levelized tariff on first come first serve basis.

ix. Earnest Money Deposit (EMD)

In case of bidding, the RPG shall provide Earnest Money Deposit (EMD) in the form of Bank Guarantee along with EoI.

x. Performance Bank Guarantee(PBG)

The EMD will be converted into PBG for the successful bidders, who has signed the PPA within stipulated time period.

xi. Universal Solar Pump Controller

Universal Solar Pump Controller (USPC) is an electronic device through which farmer can use solar power for other activities like operating chaff cutter, floor mill, cold storage, drier, etc., and increase his income.



Annexure-I

MODEL LEASE AGREEMENT

This AGREEMENT OF LEASE

entered into on this _____

day of

BETWEEN:

(hereinafter referred to as the "LESSOR/OWNER", which expression shall, wherever the context so requires or admits, SHALL mean and include his legal heirs, executors, administrators and assignees);

AND:

Π.

ULA DIST.

(Name of Renewable Power Generator (RPG))

Represented by

(hereinafter referred to as the "LESSEE, which expression shall, wherever the context so requires or admits, SHALL mean and include its executors, administrators and assignees successors in interest).

L WHEREAS the Lessor is the owner in possession of the Barren/ Agricultural land measuring _____ Acre _____ Kanal ____ marla ____ Share out of Hadbast Khewat No.____Khatoni No.___Khasra No.____Mustil No. No. ______ situated at Village/City Kila No. Tehsil

District which is morefully described in the Schedule hereunder and hereinafter referred to as the Schedule property.

(Note: The legal revenue terms to be changes to those prevalent in the State)

(Name of RPG) WHEREAS the being a (Details of RPG) with an object to plan, develop and operate Renewable Energy based Power Plant (REPP) under MNRE Scheme notified on 8th March 2019.

(a) WHEREAS pursuant to the request of the Lessee, the Lessor has agreed to grant the lease, the Lessee has agreed to take on lease from the Lessor the land which is morefully described in Schedule written hereunder and hereinafter referred to as "THE SCHEDULE PROPERTY" for setting up of the " Power Plant".

(b) That pursuant to the request of the Lessee, the Lessor has submitted an application under Section for the conversion of the land and on behalf of the Lessor/owner the

_____(Name of RPG) shall presume that the land is deemed to have been converted fornonagricultural purposes. (Clause to be modified as per State Policy for use of Agriculture land for generation of renewable power)

III. NOW THIS AGREEMENT OF LEASE WITNESSES THAT in consideration of the above and of the mutual covenants of the Parties hereto, the Lessor hereby grants and the Lesseehereby accepts the lease of the Schedule property on the following terms and conditions:

1. PURPOSE OF LEASE:

The grant of lease by the Lessor to the lessee in respect of the Schedule property is for the purpose of developing a______Power Plant under MNRE Scheme notified on 8th March 2019.

2. PERIOD OF THE LEASE

The period of this Lease shall be for Twenty-seven (27) years from this day which may be renewed at the option of the Lessee and Lessor for further period, on such mutually agreeable terms as may be agreed at the time of renewal, by both the parties, by executing and registeringseparate Lease Agreement.

3. RENT

The rent payable by the Lessee to the Lessor after Commercial Operation of the power plant shall be Rs._per unit of total power generated from the power



plant installed on the land of Lessor. Till the start of commercial operation of the plant, the rent shall be Rs._____/- (Rupees_____) only per annum per Acre.

(b) The annual rent shall be paid in twelve equal instalments and each instalment to bepaid by 5th day of every month, by crediting the same to the Lessor's Bank Account the details of which may be furnished by the Lessor from time to time. OR

In case of lease rent on the basis of Rs.____per unit, the monthly lease rent wouldbe calculated on the basis of monthly electricity injected in to the grid from the power plant installed on the land of Lessor.

(c) Lessor may opt for payment of lease rent directly from the Distribution company, which will sign Power Purchase Agreement with Lessee for the above-mentioned

_____Power Plant to be installed by Lessee. In such a case the Distribution company will pay the lease rent to Lessor on monthly basis from the proceeds payable to the Lessee in lieu of Power supplied by Lessee. In order to give this effect a suitable provision will be made in the PPA to be signed between Lessee and the Distribution Company.

- (d) [on mutual agreement between Lessor and Lessee] The rent hereby reserved shall be paid by enhancing the same at the end of every _____year(s), at_% on therent hereby agreed.
- (e) If the Lessee delays the payment of rent by due date of every month, for any reason, the same shall be paid by adding the interest at the rate % for the said delayed period.

4. GENERAL TERMS

i. In consideration of the rent herein agreed as payable to the Lessor being paid by the Lessee regularly and on complying other terms and conditions and covenants by the Lessee, the Lessee shall peacefully possess and enjoy the Schedule Property during the lease period without any interruption by the Lessor.



- ii. The Lessor shall allow the Lessee or its representatives to conduct survey and otherrelated work.
- iii. The Lessor has no objections for the Lessee to establish the ______ Power Plant in the Schedule property which is the purpose of the grant of this lease and to that effect the Lessee entering into any agreement/s, deeds with companies, individuals, developers/third party etc. in respect of the Schedule property.
- iv. The Lessor has no objections for the Lessee or its representatives for installation of machineries, equipments, etc. for generation of _______ power in the Schedule property and all work relating to thereto including but not limited to laying poles, wires, etc.

5. EVENT OF SALE, ACCEPTANCE OF LEASE BY THE NEW OWNER

- (a) In the event of the owners transferring their rights/interest in any manner during the existence of the lease to any other person, the same may be allowed without affecting the rights of the Lessee under the Lease Agreement in any manner and the owners/purchasers/transferees shall inform the Lessee about the acquiring of the right/interest in respect of the leased property and on receipt of such information, the Lessee shall accept such new purchaser's/transferee's ownership of the land and obtain a written confirmation from such new owner/purchaser/transferee to the effect that he will be bound by the terms of the Lease Agreement.
- (b) In the event of the owners transferring their rights/interest to any other person, the samemay be informed to the Lessee and the Lessor shall ascertain and obtain all the necessary documents from the transferee to the effect that the transferee will be boundby the terms and conditions of the Lease Agreement for the balance period of the leaseor for using the said documents for renewal of the lease for the balance period.
- (c) During the subsistence of the lease, the Lessor shall not carry any activity, in the Scheduleproperty, other than those agreed in this agreement;



- (d) The change in the legal status of the Lessee shall not affect the terms and conditions of this Agreement.
- (e) The original Lease Agreement shall be with the Lessee and the copy of the same will be with the Lessor.
- (f) In the event of any dispute in respect of the land, the Lessee shall deposit the rent in the concerned civil court. In the event of retention of the rent with the Lessee, the Lessee shall be pay the same together with interest thereon at the rate_% for such period.
- (g) The Lessee shall not offer or create any charge or encumbrance by offering the same asby way of mortgage, security, etc. in favour of any Banks or financial institutions in respect of the loans or advances or any other financial facilities that may be availed by the Lessee.
- (h) The owners shall pay the land tax/revenue in respect of the lands.

6. PAYMENT OF STAMP DUTY AND REGISTRATION CHARGES:

The stamp duty and other registration charges, as applicable for this Agreement of Leaseshall be paid by the Lessee.

7. FORCE MAJEURE:

It is also agreed and understood between the parties that in case of any mishap due to fire, earthquake, strike, floods, tempest, war, riot, civil war or civil commotions, mob violence, civil disturbance, act of God or on account of terrorist attack, the Lessor shall not be liablefor any loss or damage that may be occasioned to the Lessee/its merchandise.

8. ADDRESSES FOR CORRESPONDENCE, ETC

Any notice and/or communications between the Parties shall be deemed to be sufficient, if delivered by hand under acknowledgement or sent by registered post acknowledgementdue to the following address or the address that may be intimated in writing to the Lesseeby the Lessor from time to time:

LESSOR'S:



LESSEE'S:

9. LESSOR'S DUTIES, COVENANTS AND OBLIGATIONS

- a) The Lessor hereby covenants with the Lessee that the Lessee paying regularly the rentshereby reserved and performing and observing all the covenants of the Lessee herein contained, shall be entitled, during the subsistence of this lease to enjoy the Schedule property without let, hindrance or interference from the Lessor or any other person/s claiming through or under him; Still, in the event of the Lessee restrained from enjoyingthe peaceful possession of the Schedule property or on account of any action by the Government during the period of lease and in the event of dispossession of the Lessee from the Schedule property or any portion thereof forcibly, due to any default of the Lessor, the Lessor shall make good the reasonable loss that may be suffered by the Lessee.
- b) The Lessor shall offer necessary support and co-operation to the Lessee in its process toobtain required permission/s, approval/s, clearances, etc., from any Statutory Authority orother Local Bodies for the purpose of obtaining and licence, permissions, etc., for installation of power plant. However, obtaining such permission/s, approval/s, clearances, etc., shall be the sole responsibility of Lessee.

10. LESSEE'S COVENANT AND OBLIGATIONS

The Lessee hereby covenants with the Lessor as under:

- (a) The Schedule property shall be utilised for the purpose referred to in Clause (1) above;
- (b) The Lessee shall pay the rents (as per Clause (3)) regularly and promptly;

11. TERMINATION AND RE-ENTRY

The Lease shall be determinable under all or any of the following circumstances, namely

- i) by efflux of time;
- ii) in the event of breach by either party of the terms, conditions and covenants hereof;



- iii) if the Scheduled Premises or any part thereof is severely damaged or destroyed due to any unforeseen circumstances or civil commotion, act of God, etc., and these damages be not restored to by the LESSOR within a reasonable time or if the demised premises is acquired compulsorily by any authority;
- iv) After the expiry of lease period, the Lessee shall handover the land to the Lessor asit was existed previously at the time of this agreement (subject to normal wear andtear).

12. VARIATION:

The Lessor and the Lessee hereto acknowledge that this agreement supersedes all prior communications between them including all oral or written proposals. Any variation, addition and modifications of this agreement between the parties shall be valid only if in writing by the Lessor and Lessees authorized representative.

13. ARBITRATION:

- a) Any disputes or differences arising between the Parties hereto as to the effect, interpretation or application any of the clauses of this LEASE AGREEMENT or as to their rights, duties or liabilities thereunder, or as to any act, matter or thing arising out of, or consequent to, or in connection with this LEASE AGREEMENT shall be referred to and resolved by Arbitration by referring the same for arbitration to any retired District Judge and shall be resolved finally at his arbitration under Arbitration and Conciliation Act 1996 and its Amendments or any other Enactment. The Arbitration proceedings shall be held at and shall be in English/ Language.



IN WITNESS WHEREOF the parties hereto have executed these presents in the presence of the witnesses attesting hereunder on the day, month and year mentioned hereinabove.

LESSOR

LESSEE

WITNESSES: 1.

2.



SCHEDULE PROPERTY

	All that piece	and par	rcel of Barren/	Agricultur	al land	measuring		_Acre
Kanal_	m Khew:	arla	Share	out	of	Hadbast	No.	-
No	Khatoni Kila	No. No.	Khasra	No		_Mustil	No.	
	si	tuated a	at Village/City_		_	Tehsil		
	D	istrict a	ndbounded on t	he:				

(Note: The legal revenue terms to be changes to those prevalent in the State)

East by:	
West by	
North by	
South by	

2



Annexure-II

STANDARD

POWER PURCHASE AGREEMENT (PPA) FOR

PROCUREMENT OF MW____POWER ON LONG TERM BASIS

Between

[Name of Renewable Power Generator]

And

[Name of Distribution Company]

[month and year]



This Power Purchase Agreement is made on the _____day of ______

Between

at

[name of the Renewable Power Generator], [details of Renewable Power Generator] (hereinafter referred to as "Renewable Power Generator or RPG", which expression shall, unless repugnant to the context or meaning thereof, be deemed to include its successors and permitted assigns) as a Party of the First Part;

And

[Distribution Company], a company incorporated under the Companies Act 1956, having its registered office at _________(hereinafter referred to as "DISCOM", which expression shall, unless repugnant to the context or meaning thereof, be deemed to include its successors and assignees) as a Party of the Second Part;

The RPG and DISCOM are individually referred to as 'Party' and collectively referred to as 'Parties'.

WHEREAS:

DISTRIA

A. The Ministry of New and Renewable Energy [MNRE] has launched a scheme for farmers on 8th March 2019 and issued implementation guidelines on ______

B. The MNRE has accorded a sanction and allotted a capacity of _____MW to DISCOM for under the said scheme of 8th March 2019.

C. DISCOM had initiated a selection process for procurement of _____MW of the power generated from the Grid connected ______Power Project on the terms and conditions contained in the EoL/RfS No. ______dated ______.

D. The RPG has been selected in the Process for development, generation and supply of electricity from the _____MW ____Power Project to be established by RPG at _____[location of proposed power plant] and electricity generated to be fed to the _____[Name and location of 33/11 kV substation];

E. DISCOM has issued the Letter of Award No...... dated...... in favour of the RPG for development and establishment of the MW_____Power Project as per the terms and conditions contained in the EoI/RfS.

F. The RPG has furnished the Performance Bank Guarantee in the sum of Rs....... in favour of DISCOM as per the format prescribed by the DISCOM.

for generation and sale of electricity by the RPG to DISCOM at_____33(or 66 or 110)/11 kV S/S;

H. The parties have agreed to execute this Power Purchase Agreement in terms of the EoJ/RfS and the Letter of Award in regard to the terms and conditions for establishment of the _____Power Project at, and for generation and supply of electricity by the RPG to DISCOM.

Now therefore, in consideration of the premises and mutual agreements, covenants and conditions set forth herein, it is hereby agreed by and between the Parties as follows:

ARTICLE 1: DEFINITIONS AND INTERPRETATION

1.1 Definitions

The terms used in this Agreement, unless as defined below or repugnant to the context, shall have the same meaning as assigned to them by the Electricity Act, 2003 and the rules or regulations framed there under, including those issued/framed by the Appropriate Commission (as defined hereunder), as amended or re-enacted from time to time.

"Act" or "Electricity Act, 2003"	shall mean the Electricity Act, 2003 and include any modifications, amendments and substitution from time to time;
"Agreement" or "Power Purchase Agreement" or "PPA"	shall mean this Power Purchase Agreement including its recitals and Schedules, amended or modified from time to time in accordance with the terms hereof;
"Appropriate Commission"	Unless otherwise stated, Appropriate Commission shall be the commission of the state where DISCOM is situated;
"Bill Dispute Notice"	shall mean the notice issued by a Party raising a Dispute regarding a Monthly Bill or a Supplementary Bill issued by the other Party;
"Business Day"	shall mean with respect to RPG and DISCOM, a day other than Sunday or a statutory holiday, on which the banks remain open for business in the State;
"Capacity Utilisation Factor" or "CUF"	shall have the same meaning as provided in CERC (Terms and Conditions for Tariff determination from Renewable Energy Sources) Regulations, 2009as amended from time to time; However, for avoidance of any doubt, it is clarified that the CUF shall be calculated on the Contracted Capacity; In any Contract Year, if 'X' MWh of energy has been metered out at the Delivery Point for 'Y' MW Project capacity, CUF= (X MWh/(Y MW*8766)) X100%;
"Change in Law"	shall have the meaning ascribed thereto in Article 12 of this Agreement;



"Commercial Operation Date (COD)"	shall mean the date on which the commissioning certificate is issued upon successful commissioning (as per provisions of this Agreement) of the project;
"Competent Court of Law"	shall mean any court or tribunal or any similar judicial or quasi-judicial body in India that has jurisdiction to adjudicate upon issues relating to this Agreement;
"Consents, Clearances and	shall mean all authorizations, licenses, approvals, registrations, permits, waivers,
"Permits"	privileges, acknowledgements, agreements, or concessions required to be obtained from or provided by any concerned authority for the purpose of setting up of the generation facilities and/ or supply of power;
"Consultation	shall mean the period of ninety (90) days or such other longer period as the
Period"	Parties may agree, commencing from the date of issuance of a RPG Preliminary Default Notice or DISCOM Preliminary Default Notice as provided in Article 13 of this Agreement, for consultation between the Parties to mitigate the consequence of the relevant event having regard to all the circumstances;
"Contract Year"	 shall mean the period beginning from the Effective Date and ending on the immediately succeeding March 31 and thereafter each period of 12 months beginning on April 1 and ending on March 31 provided that: (i) in the financial year in which the COD would occur, the Contract Yea shall end on the date immediately before the COD and a new Contract Yea shall commence once again from the COD and end on the immediately succeeding March 31, and thereafter each period of twelve (12) month commencing on April 1 and ending on March 31, and (ii) provided further that the last Contract Year of this Agreement shall end on the last day of the Term of this Agreement
"Contracted Capacity"	shall mean [Insert capacity] MW contracted with DISCOM for supply by the RPG to DISCOM at the Delivery Point from the Power Project;
"Delivery Point"	"Delivery Point" shall mean the point at the voltage level of 11kV or above of the 33/11 kV Sub-station. Metering shall be done at this interconnection point where the power is injected into the 33/11 kV Sub-station. For interconnection with grid and metering, the RPG shall abide by the relevant and applicable regulations, Grid Code notified by the State Commission and Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006 as amended and revised from time to time, or orders passed thereunder by the Appropriate Commission or CEA. All charges and losses related to Transmission of power from project up to Delivery Point as notified by the Appropriate Commission shall be borne by the RPG.

VUBANESN

"Dispute"	shall mean any dispute or difference of any kind between DISCOM and the	
-	RPG, in connection with or arising out of this Agreement including but not	
	limited to any issue on the interpretation and scope of the terms of this	
	Agreement as provided in Article 16 of this Agreement;	
"Due Date"	Due Date shall mean the forty-fifth (45th) day after a Monthly Bill (including all the relevant documents) or a Supplementary Bill is received in hard copy and duly acknowledged by the DISCOM or, if such day is not a Business Day the immediately succeeding Business Day, by which date such Monthly Bill or a Supplementary Bill is payable by the DISCOM.	
"Effective Date"	shall have the meaning ascribed thereto in Article 2.1 of this Agreement;	
"Electricity Laws"		
"Event of Default"	shall mean the events as defined in Article 13 of this Agreement;	
"Expiry Date"	Shall mean the date occurring twenty five (25) years from the Comme Operation Date subject to that the supply of power shall be limited for a pe of 25 years from the COD unless extended by the Parties as per this Agreement;	
"Financing Agreements"	shall mean the agreements pursuant to which the RPG has sought financing for the Power Project including the loan agreements, security documents, notes, indentures, security agreements, letters of credit and other documents, as may be amended, modified, or replaced from time to time, but without in anyway increasing the liabilities of DISCOM;	
"Force Majeure" or "Force Majeure Event"	shall have the meaning ascribed thereto in Article 11 of this Agreement;	
'Indian	shall mean the Government of India, Governments of state of and	
Governmental Instrumentality"	any ministry, department, board, authority, agency, corporation, commission under the direct or indirect control of Government of India or the above state Government or both, any political sub-division of any of them including any court or Appropriate Commission or tribunal or judicial or quasi-judicial body in India;	
Insurances"	shall mean the insurance cover to be obtained and maintained by the RPG in accordance with Article 8 of this Agreement;	



"Interconnection	shall mean the facilities on RPG's side of the Delivery Point for scheduling			
Facilities"	transmitting and metering the electrical output in accordance with thi			
	Agreement and which shall include, without limitation, all other transmission			
	lines and associated equipment, transformers, relay and switching equipment			
	and protective devices, safety equipment and RTU, Data Transfer and			
	Acquisition facilities for transmitting data subject to Article 7, the Metering			
	System required for supply of power as per the terms of this Agreement;			
"Invoice" or "Bill"	shall mean either a Monthly Bill / Supplementary Bill or a Monthly Invoice/			
	Supplementary Invoice raised by any of the Parties;			
"Late Payment Surcharge"	shall have the meaning ascribed thereto in Article 10.3.3 of this Agreement;			
"Law"	shall mean in relation to this Agreement, all laws including Electricity Laws			
	in force in India and any statute, ordinance, regulation, notification or code,			
	rule, or any interpretation of any of them by an Indian Governmental			
	Instrumentality and having force of law and shall further include without limitation all applicable rules, regulations, orders, notifications by an Indian			
	Governmental Instrumentality pursuant to or under any of them and shall			
	include without limitation all rules, regulations, decisions and orders of the			
	Appropriate Commissions;			
"Letter of Credit" or "L/C"	shall have the meaning ascribed thereto in Article 10.4 of this Agreement;			
"Letter of Award" or "LoA"	shall mean Letter of Award issued by the DISCOM to the RPG for the project;			
"MNRE"	shall mean the Ministry of New and Renewable Energy, Government of			
MINKE	India;			
"Month"	shall mean a period of thirty (30) days from (and excluding) the date of the			
	event, where applicable, else a calendar month;			
"Party" and "Parties"	shall have the meaning ascribed thereto in the recital to this Agreement;			
"Payment Security Mechanism"	shall have the meaning ascribed thereto in Article 10.4 of this Agreement;			



"Power Project"	shall mean thepower generation facility of Contracted Capacity of
or "Project"	
"Preliminary Default Notice"	shall have the meaning ascribed thereto in Article 13 of this Agreement;
"Project Capacity"	shall mean the maximum AC capacity of the Project at the point of injection on which the Power Purchase Agreement has been signed.
"Prudent Utility Practices"	 shall mean the practices, methods and standards that are generally accepted internationally from time to time by electric utilities for the purpose of ensuring the safe, efficient and economic design, construction, commissioning, operation and maintenance of power generation equipment and which practices, methods and standards shall be adjusted as necessary, to take account of: a) operation and maintenance guidelines recommended by the manufacturers of the plant and equipment to be incorporated in the Power Project; b) the requirements of Indian Law; and the physical conditions at the site of the Power Project
"Rebate"	shall have the same meaning as ascribed thereto in Article 10.3.5 of this Agreement;
"Rupees", "Rs.", "K"	shall mean Indian rupees, the lawful currency of India;
"Scheduled Commissioning Date" or "SCD" of the Project	Shall mean
"Tariff"	Shall have the same meaning as provided for in Article 9 of this Agreement;
"Tariff Payment"	shall mean the payments to be made under Monthly Bills as referred to in Article 10 and the relevant Supplementary Bills;
"Termination Notice"	shall mean the notice given by either Parties for termination of this Agreement in accordance with Article 13 of this Agreement;



"Term of Agreement" shall have the meaning ascribed thereto in Article 2 of this Agreement;

ARTICLE 2: TERM OF AGREEMENT

2.1 Effective Date

2.1.1 This Agreement shall come into effect from ______ and such date shall be referred to as the Effective Date.

2.2 Term of Agreement

2.2.1 Subject to Article 2.3 and 2.4 of this Agreement, this Agreement shall be valid for a term from the Effective Date until the Expiry Date. This Agreement may be extended for a further period at least one hundred eighty (180) days prior to the Expiry Date, on mutually agreed terms and conditions.

2.2.2 The RPG is free to operate their plants beyond the Expiry Date if other conditions like land lease / Right to Use of Land (as applicable), permits, approvals and clearances etc. allow. In such case unless otherwise agreed by the DISCOM, DISCOM shall not be obligated to procure power beyond the Expiry Date.

2.3 Early Termination

2.3.1 This Agreement shall terminate before the Expiry Date if either DISCOM or RPG terminates the Agreement, pursuant to Article 13 of this Agreement.

2.4 Survival

2.4.1 The expiry or termination of this Agreement shall not affect any accrued rights, obligations and liabilities of the Parties under this Agreement, including the right to receive penalty as per the terms of this Agreement, nor shall it affect the survival of any continuing obligations for which this Agreement provides, either expressly or by necessary implication, which are to survive after the Expiry Date or termination including those under Article 11 (Force Majeure), Article 13 (Events of Default and Termination), Article 14 (Liability and Indemnification), Article 16 (Governing Law andDispute Resolution), Article 17 (Miscellaneous Provisions), and other Articles and Schedules of this Agreement which expressly or by their nature survive the Term or termination of this Agreement shall continue and survive any expiry or termination of this Agreement.

ARTICLE 3: CONDITIONS SUBSEQUENT

3.1 The RPG agrees and undertakes to make Project Financing Arrangements for its Project and shall provide necessary documents to DISCOM in this regard within six Months from the Date of issue of box by DISCOM for the project.

ARTICLE 4: CONSTRUCTION & DEVELOPMENT OF THE PROJECT

4.1 **RPG's Obligations**

4.1.1 The RPG undertakes to be responsible, at RPG's own cost and risk, for:

a) The RPG shall be solely responsible and make arrangements for Land & associated infrastructure for development of the Project and for Connectivity with the 33/11 kV sub-station for confirming the evacuation of power by the Scheduled Commissioning date or COD, whichever is earlier, and all clearances related thereto;

The RPG shall furnish the necessary documents to establish possession in the name of the Project Developer of the required land/ Lease Agreement;

- b) obtaining all Consents, Clearances and Permits as required and maintaining all documents.
- c) Designing, constructing, erecting, commissioning, completing and testing the Power Project in accordance with the applicable Law, the Grid Code, the terms and conditions of this Agreement and Prudent Utility Practices.
- d) the commencement of supply of power up to the Contracted Capacity to DISCOM no later than the Scheduled Commissioning Date and continuance of the supply of power throughout the term of the Agreement;
- e) Connecting the Power Project switchyard with the Interconnection Facilities at the Delivery Point. The RPG shall make adequate arrangements to connect the Power Project switchyard with the Interconnection Facilities at Interconnection / Metering / Delivery Point.
- f) owning the Power Project throughout the Term of Agreement free and clear of encumbrances, except those expressly permitted under Article 15;
- f) fulfilling all obligations undertaken by the RPG under this Agreement.
- g) The RPG shall be responsible to for directly coordinating and dealing with the DISCOM, and other authorities in all respects in regard to declaration of availability, scheduling and dispatch of Power and due compliance with deviation and settlement mechanism and the applicable Grid code/State Regulations.

4.2 Purchase and sale of Contracted Capacity

4.2.1 Subject to the terms and conditions of this Agreement, the RPG undertakes to sell to DISCOM and DISCOM undertakes to pay Tariff for all the energy supplied at the Delivery Point corresponding to the Contracted Capacity.

4.3 Right to Contracted Capacity & Energy



of grid non-availability for evacuation which is beyond the control of the RPG. This compensation shall be applied to the amount of shortfall in generation during the Contract Year. The amount of such penalty shall be as determined by the Appropriate Commission, and such penalty shall ensure that the DISCOM is offset for all potential costs associated with low generation and supply of power under the PPA. However, the minimum compensation payable to DISCOM by the RPG shall be 25%(twenty-five percent) of the cost of this shortfall in energy terms, calculated at PPA tariff. This compensation shall not be applicable in events of Force Majeure identified under PPA.

4.3.2 In case at any point of time, the peak of capacity reached is higher than the contracted capacity and causes disturbance in the system at the point where power is injected, the RPG will have to forego the excess generation and reduce the output to the contract capacity and shall also have to paythe penalty/charges (if applicable) as per applicable regulations.

4.4 Extensions of Time

4.4.1 In the event that the RPG is prevented from performing its obligations under Article 4.1 by the Scheduled Commissioning Date due to:

- a) any DISCOM Event of Default; or
- b) Force Majeure Events affecting DISCOM, or
- c) Force Majeure Events affecting the RPG,

the Scheduled Commissioning Date and the Expiry Date shall be deferred, subject to Article 4.4.5, for a reasonable period but not less than 'day for day' basis, to permit the RPG or DISCOM through the use of due diligence, to overcome the effects of the Force Majeure Events affecting the RPG or DISCOM, or till such time such Event of Default is rectified by DISCOM.

4.4.2 In case of extension due to reasons specified in Article 4.4.1(b) and (c), and if such Force Majeure Event continues even after a maximum period of three (3) months, any of the Parties may choose to terminate the Agreement as per the provisions of Article 13.5. In case neither party terminates the agreement under this clause, the agreement shall stand terminated on the expiry of twelve (12) months of the continuation of the Force majeure event unless the parties mutually agree to extend the agreement for the further period.

4.4.3 If the Parties have not agreed, within thirty (30) days after the affected Party's performance has ceased to be affected by the relevant circumstance, on the time period by which the Scheduled Commissioning Date or the Expiry Date should be deferred, any Party may raise the Dispute to be resolved in accordance with Article 16.

4.4.4 As a result of such extension, the newly determined Scheduled Commissioning Date and newly determined Expiry Date shall be deemed to be the Scheduled Commissioning Date and the Expiry Date for the purposes of this Agreement.



4.4.5 Notwithstanding anything to the contrary contained in this Agreement, any extension of the Scheduled Commissioning Date arising due to any reason envisaged in this Agreement shall not be allowed beyond the date pursuant to Article 4.5.2.

4.4.6 Delay in commissioning of the project beyond the scheduled commissioning date for reasons other than those specified in Article 4.4.1 shall be an event of default on part of the RPG and shall be subject to the consequences specified in the Article 4.5.

4.5 Liquidated Damages not amounting to penalty for delay in Commissioning

4.5.1 If the RPG is unable to commission the Project by the Scheduled Commissioning Date other than for the reasons specified in Article 4.4.1, the RPG shall pay to DISCOM, damages for the delay in such commissioning and making the Contracted Capacity available for dispatch by the Scheduled Commissioning Date as per the following:

Delay beyond the Scheduled Commissioning Date upto (& including) the date as on fifteen months from the Date of issue of LoA: The total Performance Bank Guarantee amount shall be encashed on per day basis and proportionate to the balance capacity not commissioned.

4.5.2 The maximum time period allowed for commissioning of the full Project Capacity with encashment of Performance Bank Guarantee shall be limited to 21 Months from the Date of issue of LoA. In case, the Commissioning of the Project is delayed beyond 21 Months from the Date of issue of LoA, it shall be considered as an RPG Event of Default and provisions of Article 13 shall apply and the Contracted Capacity shall stand reduced / amended to the Project Capacity Commissioned within 21 Months of the Date of issue of LoA and the PPA for the balance Capacity will stand terminated and shall be reduced from the project capacity.

4.5.3 The RPG further acknowledge that the amount of the liquidated damages fixed is genuine and reasonable pre-estimate of the damages that may be suffered by DISCOM.

4.6 Acceptance/Performance Test

4.6.1 Prior to synchronization of the Power Project, the RPG shall be required to get the Project certified for the requisite acceptance/performance test as may be laid down by respective authorities.

4.7 Third Party Verification

4.7.1 The RPG shall be further required to provide entry to the site of the Power Project free of all encumbrances at all times during the Term of the Agreement to DISCOM and a third Party nominated by any Indian Governmental Instrumentality for inspection and verification of the worksbeing carried out by the RPG at the site of the Power Project.

4.7.2 The third party may verify the construction works/operation of the Power Project being carried out by the RPG and if it is found that the construction works/operation of the Power Project is notas per the Prudent Utility Practices, it may seek clarifications from RPG or require the works tobe stopped or to comply with the instructions of such third party.



4.8 Breach of Obligations

4.8.1 The Parties herein agree that during the subsistence of this Agreement, subject to DISCOM being in compliance of its obligations & undertakings under this Agreement, the RPG would have no right to negotiate or enter into any dialogue with any third party for the sale of Contracted Capacity of power which is the subject matter of this Agreement. It is the specific understanding between the Parties that such bar will apply throughout the entire term of this Agreement.

4.9 Generation compensation for Off-take constraints

4.9.1 Generation Compensation in offtake constraints due to Grid Unavailability: During the operation of the plant, there can be some periods where the Project can generate power but due to temporary transmission unavailability, the power is not evacuated, for reasons not attributable to the RPG. In such cases, subject to the submission of documentary evidences from the competent authority, the generation compensation shall be restricted to the following and there shall be no otherclaim, directly or indirectly against DISCOM:

Duration of Grid unavailability	Provision for Generation Compensation		
Grid unavailability in a contract yearas defined in the PPA: (only period from 8 am to 6 pm to be counted):	Generation Loss = [(Average Generation per hour during the Contract Year) × (number of hours of grid unavailability during the Contract Year)] Where, Average Generation per hour during the Contract Year (kWh) = Total generation in the Contract Year (kWh) ÷ Total hours of generation in the Contract Year.		

The excess generation by the RPG equal to this generation loss shall be procured by DISCOM at the PPA tariff so as to offset this loss in the succeeding 3 (three) Contract Years.

4.9.2 Offlake constraints due to Backdown: The RPG and DISCOM shall follow the forecasting and scheduling process as per the regulations in this regard by the Appropriate Commission. In the eventuality of backdown, subject to the submission of documentary evidences from the competent authority, the RPG shall be eligible for a minimum generation compensation, from DISCOM, restricted to the following and there shall be no other claim, directly or indirectly against DISCOM:

Duration of Backdown	Provision for Generation Compensation		
Hours of Backdown during a monthly billing cycle.	Mininum Generation Compensation = 50% of [(Average Generation per hour during the month) X (number of backdown hours during the month)] X PPA tariff		
RIBUIRDA	Where, Average Generation per hour during the month (kWh) = Total generation in the month (kWh) ÷ Total hours of generation in the month		

The RPG shall not be eligible for any compensation in case the Backdown is on account of events like consideration of grid security or safety of any equipment or personnel or other such conditions. The Generation Compensation shall be paid as part of the energy bill for the successive month after JMR.

ARTICLE 5: SYNCHRONISATION, COMMISSIONING AND COMMERCIAL OPERATION

5.1 Synchronization, Commissioning and Commercial Operation

5.1.1 The RPG shall give the DISCOM at least thirty (30) days' advanced preliminary written notice and at least fifteen (15) days' advanced final written notice, of the date on which it intends to synchronize the Power Project to the Grid System.

5.1.2 Subject to Article 5.1.1, the Power Project may be synchronized by the RPG to the Grid System when it meets all the connection conditions prescribed in applicable Grid Code then in effect and otherwise meets all other Indian legal requirements for synchronization to the Grid System.

5.1.3 The synchronization equipment and all necessary arrangements / equipment including RTU for scheduling of power generated from the Project and transmission of data to the concerned authority as per applicable regulation shall be installed by the RPG at its generation facility of the Power Project at its own cost. The RPG shall synchronize its system with the Grid System only after the approval of synchronization scheme is granted by the head of the concerned substation/ and checking/verification is made by the concerned authorities of the DISCOM.

5.1.4 The RPG shall immediately after each synchronization/tripping of generator, inform the substation of the Grid System to which the Power Project is electrically connected in accordance with applicable Grid Code. In addition, the RPG will inject in-firm power to grid time to time to carry out operational/ functional test prior to commercial operation. For avoidance of doubt, it is clarified that Synchronization / Connectivity of the Project with the grid shall not to be considered as Commissioning of the Project.

5.1.5 The RPG shall commission the Project within fifteen (15) Months from the Date of issue of LoA. Declaration of COD shall only be done upon the successful visit by the Commissioning Committee.

5.1.6 The Parties agree that for the purpose of commencement of the supply of electricity by RPG to DISCOM, liquidated damages for delay etc., the Scheduled Commissioning Date as defined in this Agreement shall be the relevant date.



ARTICLE 6: DISPATCH AND SCHEDULING

6.1 Dispatch and Scheduling

6.1.1 The RPG shall be required to schedule its power as per the applicable regulations of SERC /SLDC or any other competent agency and same being recognized by the SLDC or any other competent authority / agency as per applicable regulation/ law / direction and maintain compliance to the applicable Codes/ Grid Code requirements and directions, if any, as specified by concerned SLDC from time to time. Any deviation from the Schedule will attract the provisions of applicable regulation / guidelines / directions and any financial implication on account of this shall be on the account of the RPG.

6.1.2 The RPG shall be responsible for directly coordinating and dealing with the DISCOM, State Load Dispatch Centers, and other authorities in all respects in regard to declaration of availability, scheduling and despatch of Power and due compliance with deviation and settlement mechanism and the applicable Grid code Regulations.

6.1.3 The RPG shall be responsible for any deviation from scheduling and for any resultant liabilities on account of charges for deviation as per applicable regulations. UI charges on this account shall be directly paid by the RPG.

6.1.4 Auxiliary power consumption will be treated as per the concerned state regulations.

ARTICLE 7: METERING

7.1 Meters

7.1.1 For installation of Meters, Meter testing, Meter calibration and Meter reading and all matters incidental thereto, the RPG and DISCOM shall follow and be bound by the Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006, the Grid Code, as amended and revised from time to time.

7.1.2 The RPG shall bear all costs pertaining to installation, testing, calibration, maintenance, renewal and repair of meters at RPG's side of Delivery Point.

7.1.3 In addition to ensuring compliance of the applicable codes, the RPG shall install Main & Check meters at the Delivery Point, along with Stand-by meter(s) as per the applicable regulations of the State where the Project is located.

7.2 Reporting of Metered Data and Parameters

7.2.1 The grid connected renewable power plants will install necessary equipment for regular monitoring of required data and simultaneously for monitoring of the electric power generated from the Project.



7.2.2 Online arrangement would have to be made by the RPG for submission of above data regularly for the entire period of this Power Purchase Agreement to the DISCOM, to central portal of MNRE and concerned agency as per applicable regulation / directions.

7.2.3 Reports on above parameters on monthly basis (or as required by regulation / guidelines) shall be submitted by the RPG to Ministry of New and Renewable Energy/National Institute of Solar Energy through DISCOM for entire period of PPA.

ARTICLE 8: INSURANCES

8.1 Insurance

8.1.1 The RPG shall effect and maintain or cause to be effected and maintained, at its own cost and expense, throughout the Term of PPA, Insurances against such risks to keep the Project in good condition and shall take Industrial All Risk insurance policy covering risks against any loss or damage, with such deductibles and with such endorsements and co-insured(s), which the Prudent Utility Practices would ordinarily merit maintenance of and as required under the Financing Agreements, and under the applicable laws.

8.2 Application of Insurance Proceeds

8.2.1 In case of the Project not being implemented through Financing Agreement(s), save as expressly provided in this Agreement or the Insurances, the proceeds of any insurance claim made due to loss or damage to the Power Project or any part of the Power Project shall be first applied to reinstatement, replacement or renewal of such loss or damage.

In case of the Project being financed through Financing Agreement(s), save as expressly provided in this Agreement or the Insurances, the proceeds of any insurance claim made due to loss or damage to the Power Project or any part of the Power Project shall be applied as per such Financing Agreements.

8.2.2 If a Force Majeure Event renders the Power Project no longer economically and technically viable and the insurers under the Insurances make payment on a "total loss" or equivalent basis, DISCOM shall have claim on such proceeds of such Insurance limited to outstanding dues of DISCOM against RPG.

8.3 Effect on liability of DISCOM

8.3.1 Notwithstanding any liability or obligation that may arise under this Agreement, any loss, damage, liability, payment, obligation or expense which is insured or not or for which the RPG can claim compensation, under any Insurance shall not be charged to or payable by DISCOM. It is for the RPG to ensure that appropriate insurance coverage is taken for payment by the insurer for the entire loss and there is no under insurance or short adjustment etc.



ARTICLE 9 : APPLICABLE TARIFF

9.1 The RPG shall be entitled to receive the Tariff of Rs...... / kWh, fixed for the entire term of this Agreement, with effect from the COD, for the power sold to the DISCOM as reflected in the Energy Accounts.

ARTICLE 10: BILLING AND PAYMENT

10.1 General

10.1.1 From the commencement of supply of power, DISCOM shall pay to the RPG the monthly Tariff Payments subject to the adjustments as per provisions of this Agreement including Article 6, in accordance with Article 9. All Tariff Payments by DISCOM shall be in Indian Rupees.

10.1.2 The RPG shall be required to make arrangements and payments for import of energy (if any) as per applicable regulations.

10.2 Delivery and Content of Monthly Bills/Supplementary Bills

10.2.1 The RPG shall issue to DISCOM hard copy of a signed Monthly Bill for the immediately preceding Month based on the JMR/Energy Account along with all relevant documents (payments made by RPG for drawl of power, payment of reactive energy charges, Metering charges or any other charges as per regulations of SERC/SLDC, if applicable.)

Each Monthly Bill shall include all charges as per this Agreement for the energy supplied for the relevant Month based on JMR/Energy Accounts. The Monthly Bill amount shall be the product of the energy as per Energy Accounts and the Applicable Tariff. Energy drawn from the grid will be regulated as per the regulations of respective State the Project is located in.

10.3 Payment of Monthly Bills

10.3.1 DISCOM shall pay the amount payable under the Monthly Bill by the Due Date to such account of the RPG, as shall have been previously notified by the RPG.

10.3.2 All payments required to be made under this Agreement shall also include any deduction or set off for:

- i) deductions required by the Law; and
- ii) Amount claimed by DISCOM, if any, from the RPG, will be adjusted from the monthly energy payment.

The RPG shall open a bank account (the "RPG's Designated Account") for all Tariff Payments to be made by DISCOM to the RPG, and notify DISCOM of the details of such account at least sixty (60) Days before the dispatch of the first Monthly Bill.

19.3.3 Late Payment Surcharge

In the event of delay in payment of a Monthly Bill by DISCOM beyond thirty (30) days of its Due

Date, a Late Payment Surcharge shall be payable to the RPG at the rate of 1.25% per month on the outstanding amount calculated on a day to day basis. The Late Payment Surcharge shall be claimed by the RPG through the Supplementary Bill.

10.3.5 Rebate

For payment of any Bill on or before Due Date, the following Rebate shall be paid by the RPG to DISCOM in the following manner and the RPG shall not raise any objections to the payments made under this article.

- a) A Rebate of 2% shall be payable to the DISCOM for the payments made within a period of seven clear working days of the presentation of hard copy of Bill along with required supporting documents at DISCOM office.
- b) Any payments made after seven clear working days of the date of presentation of hard copy of the Bill along with the required supporting documents at DISCOM office up to the Due Date shall be allowed a rebate of 1 %.
- c) For the above purpose, the date of presentation of Bill shall be the next Business Day of delivery of the physical copy of the Bill at DISCOM.
- d) No Rebate shall be payable on the Bills raised on account of Change in Law relating to taxes, duties, cess etc. and on Supplementary Bill.

For the above purpose date of presentation of bill shall be the same day of delivery in hard copy. However, for consideration of rebate, next business day shall be considered.

10.4 Payment Security Mechanism

Letter of Credit (LC):

10.4.1 DISCOM shall provide to the RPG, in respect of payment of its Monthly Bills and/or Supplementary Bills, a monthly unconditional, revolving and irrevocable letter of credit ("Letter of Credit"), opened and maintained which may be drawn upon by the RPG in accordance with this Article.

10.4.2 Not later than one (1) Month before the start of supply, DISCOM through a scheduled bank open a Letter of Credit in favour of the RPG, to be made operative from a date prior to the Due Date of its first Monthly Bill under this Agreement. The Letter of Credit shall have a term of twelve (12) Months and shall be renewed annually, for an amount equal to:

- i) for the first Contract Year, equal to the estimated average monthly billing;
- ii) for each subsequent Contract Year, equal to the average of the monthly billing of the previous Contract Year.

10.4.3 Provided that the RPG shall not draw upon such Letter of Credit prior to the Due Date of the relevant Monthly Bill and/or Supplementary Bill, and shall not make more than one drawal in a Month.

IO291

10.4.4 Provided further that if at any time, such Letter of Credit amount falls short of the amount specified in Article 10.4.2 due to any reason whatsoever, DISCOM shall restore such shortfall within fifteen (15) days.

10.4.5 DISCOM shall cause the scheduled bank issuing the Letter of Credit to intimate the RPG, in writing regarding establishing of such irrevocable Letter of Credit.

10.4.6 DISCOM shall ensure that the Letter of Credit shall be renewed not later than its expiry.

10.4.7 All costs relating to opening, maintenance of the Letter of Credit shall be borne by DISCOM.

10.4.8 If DISCOM fails to pay undisputed Monthly Bill or Supplementary Bill or a part thereof within and including the Due Date, then, subject to Article 10.4.6 & 10.5.2, the RPG may draw upon the Letter of Credit, and accordingly the bank shall pay without any reference or instructions from DISCOM, an amount equal to such Monthly Bill or Supplementary Bill or part thereof, in accordance with Article 10.4.3 above, by presenting to the scheduled bank issuing the Letter of Credit, the following documents:

- i) a copy of the Monthly Bill or Supplementary Bill which has remained unpaid to RPG and;
- ii) a certificate from the RPG to the effect that the bill at item (i) above, or specified part thereof, is in accordance with the Agreement and has remained unpaid beyond the Due Date;

10.5 Disputed Bill

10.5.1 If the DISCOM does not dispute a Monthly Bill or a Supplementary Bill raised by the RPG within fifteen (15) days of receiving such Bill shall be taken as conclusive.

10.5.2 If the DISCOM disputes the amount payable under a Monthly Bill or a Supplementary Bill, as the case may be, it shall pay undisputed amount of the invoice amount and it shall within fifteen (15) days of receiving such Bill, issue a notice (the "Bill Dispute Notice") to the invoicing Party setting out:

- i) the details of the disputed amount;
- ii) its estimate of what the correct amount should be; and iii) all written material in support of its claim.

10.5.3 If the RPG agrees to the claim raised in the Bill Dispute Notice issued pursuant to Article 10.5.2, the RPG shall revise such Bill and present along with the next Monthly Bill. In such a case excess amount shall be refunded along with interest at the same rate as Late Payment Surcharge, which shall be applied from the date on which such excess payment was made by the disputing Partyto the invoicing Party and up to and including the date on which such such payment has been received as refund.



10.5.4 If the RPG does not agree to the claim raised in the Bill Dispute Notice issued pursuant to Article 10.5.2, it shall, within fifteen (15) days of receiving the Bill Dispute Notice, furnish a notice (Bill Disagreement Notice) to the DISCOM providing:

- i) reasons for its disagreement;
- ii) its estimate of what the correct amount should be; and iii) all written material in support of its counter-claim.

10.5.5 Upon receipt of the Bill Disagreement Notice by the DISCOM under Article 10.5.4, authorized representative(s) or a director of the board of directors/ member of board of the DISCOM and RPG shall meet and make best endeavours to amicably resolve such dispute within fifteen (15) days of receipt of the Bill Disagreement Notice.

10.5.6 If the Parties do not amicably resolve the Dispute within fifteen (15) days of receipt of Bill Disagreement Notice pursuant to Article 10.5.4, the matter shall be referred to Dispute resolution in accordance with Article 16.

10.5.7 For the avoidance of doubt, it is clarified the despite a Dispute regarding an invoice, DISCOM shall, without prejudice to its right to Dispute, be under an obligation to make payment of undisputed amount of the invoice amount in the Monthly Bill.

10.6 Quarterly and Annual Reconciliation

10.6.1 The Parties acknowledge that all payments made against Monthly Bills and Supplementary Bills shall be subject to quarterly reconciliation within 30 days of the end of the quarter at the beginning of the following quarter of each Contract Year and annual reconciliation at the end of each Contract Year within 30 days to take into account the Energy Accounts, Tariff adjustment payments, Tariff Rebate, Late Payment Surcharge, or any other reasonable circumstanceprovided under this Agreement.

10.6.2 The Parties, therefore, agree that as soon as all such data in respect of any quarter of a Contract Year or a full Contract Year as the case may be has been finally verified and adjusted, the RPG and DISCOM shall jointly sign such reconciliation statement. Within fifteen (15) days of signing of a reconciliation statement, the RPG shall make appropriate adjustments in the next Monthly Bill. Late Payment Surcharge/ interest shall be payable in such a case from the date on which such payment had been made to the invoicing Party or the date on which any payment was originally due, as may be applicable. Any Dispute with regard to the above reconciliation shall be dealt with in accordance with the provisions of Article 16.

10.7 Payment of Supplementary Bill

10.7.1 RPG may raise a ("Supplementary Bill") for payment on account of:

i) Adjustments required by the Energy Accounts (if applicable); or

ii) Change in Law as provided in Article 12

And such Supplementary Bill shall be paid by the other Party.



10.7.2 DISCOM shall remit all amounts due under a Supplementary Bill raised by the RPG to the RPG's Designated Account by the Due Date, except open access charges, RLDC or scheduling charges and transmission charges (if applicable). For Supplementary Bill on account of adjustment required by energy account, Rebate as applicable to Monthly Bills pursuant to Article 10.3.5 shall equally apply. No surcharge will be applicable other than that on the monthly energy payment and associated debit and credit note.

10.7.3 In the event of delay in payment of a Supplementary Bill by either Party beyond its Due Date, a Late Payment Surcharge shall be payable at the same terms applicable to the Monthly Bill in Article 10.3.3.

ARTICLE 11: FORCE MAJEURE

11.1 Definitions

11.1.1 In this Article, the following terms shall have the following meanings:

11.2 Affected Party

11.2.1 An affected Party means DISCOM or the RPG whose performance has been affected by an event of Force Majeure.

11.3 Force Majeure

11.3.1 A 'Force Majeure' means any event or circumstance or combination of events those stated below that wholly or partly prevents or unavoidably delays an Affected Party in the performance of its obligations under this Agreement, but only if and to the extent that such events or circumstances are not within the reasonable control, directly or indirectly, of the Affected Party and could not have been avoided if the Affected Party had taken reasonable care or complied with Prudent Utility Practices:

- a) Act of God, including, but not limited to lightning, drought, fire and explosion (to the extent originating from a source external to the site), earthquake, volcanic eruption, landslide, flood, cyclone, typhoon or tornado if and only if it is declared / notified by the competent state / central authority / agency (as applicable);
- b) any act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, terrorist or military action if and only if it is declared / notified by the competent state / central authority / agency (as applicable); or
- c) radioactive contamination or ionising radiation originating from a source in India or resulting from another Force Majeure Event mentioned above excluding circumstances where the source or cause of contamination or radiation is brought or has been brought into or near the Power Project by the Affected Party or those employed or engaged by the Affected Party.



11.4 Force Majeure Exclusions

11.4.1 Force Majeure shall not include (i) any event or circumstance which is within the reasonable control of the Parties and (ii) the following conditions, except to the extent that they are consequences of an event of Force Majeure:

- a. Unavailability, late delivery, or changes in cost of the plant, machinery, equipment, materials, spare parts or consumables for the Power Project;
- b. Delay in the performance of any contractor, sub-contractor or their agents ;
- Non-performance resulting from normal wear and tear typically experienced in power generation materials and equipment;
- d. Strikes at the facilities of the Affected Party;
- e. Insufficiency of finances or funds or the agreement becoming onerous to perform; and
- f. Non-performance caused by, or connected with, the Affected Party's:
 - i. Negligent or intentional acts, errors or omissions;
 - ii. Failure to comply with an Indian Law; or
 - iii. Breach of, or default under this Agreement.

11.5 Notification of Force Majeure Event

11.5.1 The Affected Party shall give notice to the other Party of any event of Force Majeure as soon as reasonably practicable, but not later than seven (7) days after the date on which such Party knew or should reasonably have known of the commencement of the event of Force Majeure. If an event of Force Majeure results in a breakdown of communications rendering it unreasonable to give notice within the applicable time limit specified herein, then the Party claiming Force Majeure shall give such notice as soon as reasonably practicable after reinstatement of communications, but not later than one (1) day after such reinstatement.

11.5.2 Provided that such notice shall be a pre-condition to the Affected Party's entitlement to claim relief under this Agreement. Such notice shall include full particulars of the event of Force Majeure, its effects on the Party claiming relief and the remedial measures proposed. The Affected Party shall give the other Party regular (and not less than monthly) reports on the progress of those remedial measures and such other information as the other Party may reasonably request about the Force Majeure Event.

11.5.3 The Affected Party shall give notice to the other Party of (i) the cessation of the relevant event of Force Majeure; and (ii) the cessation of the effects of such event of Force Majeure on the performance of its rights or obligations under this Agreement, as soon as practicable after becoming aware of each of these cessations.



11.6 Duty to Perform and Duty to Mitigate

11.6.1 To the extent not prevented by a Force Majeure Event pursuant to Article 11.3, the Affected Party shall continue to perform its obligations pursuant to this Agreement. The Affected Party shall use its reasonable efforts to mitigate the effect of any Force Majeure Event as soon as practicable.

11.7 Available Relief for a Force Majeure Event

- 11.7.1 Subject to this Article 11:
 - (a) no Party shall be in breach of its obligations pursuant to this Agreement except to the extent that the performance of its obligations was prevented, hindered or delayed due to a Force Majeure Event;
 - (b) every Party shall be entitled to claim relief in relation to a Force Majeure Event in regard to its obligations;
 - (c) For avoidance of doubt, neither Party's obligation to make payments of money due and payable prior to occurrence of Force Majeure events under this Agreement shall be suspended or excused due to the occurrence of a Force Majeure Event in respect of such Party.
 - (d) Provided that no payments shall be made by either Party affected by a Force Majeure Event for the period of such event on account of its inability to perform its obligations due to such Force Majeure Event.

ARTICLE 12: CHANGE IN LAW

12.1 Definitions

In this Article 12, the term Change in Law shall refer to the occurrence of any of the following events pertaining to this project only after the last date of the bid submission, including

- (i) the enactment of any new law; or
- (ii) an amendment, modification or repeal of an existing law; or

(iii) the requirement to obtain a new consent, permit or license; or

(iv) any modification to the prevailing conditions prescribed for obtaining an consent, permit or license, not owing to any default of the RPG; or (v) any change in the rates of any Taxes including any duties and cess or Introduction of any new tax made applicable for setting up the power project and supply of power from the Power project by the RPG Which have a direct effect on the Project. However, Change in Law shall not include (i) any change in taxes on corporate income or (ii) any change in any withholding tax on income or dividends distributed to the shareholders of the RPG, or (iii) any change on account of regulatory measures by the Appropriate Commission.

In the event a Change in Law results in any adverse financial loss/ gain to the RPG then, in order to ensure that the RPG is placed in the same financial position as it would have been had it not been for the occurrence of the Change in Law, the RPG/ DISCOM shall be entitled to compensation by the other party, as the case may be, subject to the condition that the quantum and mechanism of compensation payment shall be determined and shall be effective from such date as may be decided by the Appropriate Commission.



In the event of any decrease in the recurring/ nonrecurring expenditure by the RPG or any income to the RPG on account of any of the events as indicated above, RPG shall file an application to the Appropriate Commission no later than sixty (60) days from the occurrence of such event, for seeking approval of Change in Law. In the event of the RPG failing to comply with the above requirement, in case of any gain to the RPG, DISCOM shall withhold the monthly tariff payments on immediate basis, until compliance of the above requirement by the RPG.

12.2 Relief for Change in Law

12.2.1 The aggrieved Party shall be required to approach the Appropriate Commission forseeking approval of Change in Law.

12.2.2 The decision of the Appropriate Commission to acknowledge a Change in Law and the date from which it will become effective, provide relief for the same, shall be final and governing on both the Parties.

ARTICLE 13: EVENTS OF DEFAULT AND TERMINATION

13.1 RPG Event of Default

13.1.1 The occurrence and/or continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by DISCOM of its obligations under this Agreement, shall constitute an RPG Event of Default:

(i) the failure to commence supply of power to DISCOM up to the Contracted Capacity, by the end of the period specified in Article 4, or failure to continue supply of Contracted Capacity to DISCOM after Commercial Operation Date throughout the term of this Agreement, or

if

- a) the RPG assigns, mortgages or charges or purports to assign, mortgage or charge any of its assets or rights related to the Power Project in contravention of the provisions of this Agreement; or
- b) the RPG transfers or novates any of its rights and/ or obligations under this agreement, in a manner contrary to the provisions of this Agreement; except where such transfer
 - is in pursuance of a Law; and does not affect the ability of the transferee to perform, and such transferee has the financial capability to perform, its obligations under this Agreement or
 - is to a transferee who assumes such obligations under this Agreement and the Agreement remains effective with respect to the transferee;
- (ii) if (a) the RPG becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding up proceedings and such proceedings remain uncontested for a period of thirty (30) days, or (b) any winding up or bankruptcy or insolvency order is passed against the RPG, or (c) the RPG goes into liquidation or dissolution or has a receiver or any similar officer appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, provided that a dissolution or liquidation of the RPG will not be a RPG Event of Default if such dissolution or liquidation is for the purpose of a merger, consolidation or



•

reorganization and where the resulting company retains creditworthiness similar to the RPG and expressly assumes all obligations of the RPG under this Agreement and is in a position to perform them; or

- (iii) the RPG repudiates this Agreement and does not rectify such breach within a period of thirty
 (30) days from a notice from DISCOM in this regard; or
- (iv) except where due to any DISCOM's failure to comply with its material obligations, the RPG is in breach of any of its material obligations pursuant to this Agreement, and such material breach is not rectified by the RPG within thirty (30) days of receipt of first notice in this regard given by DISCOM.
- (v) occurrence of any other event which is specified in this Agreement to be a material breach/ default of the RPG.
- (vi) except where due to any DISCOM's failure to comply with its material obligations, the RPG is in breach of any of its material obligations pursuant to this Agreement, and such material breach is not rectified by the RPG within thirty (30) days of receipt of first notice in this regard given by DISCOM.

13.2 DISCOM Event of Default

13.2.1 The occurrence and the continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by the RPG of its obligations under this Agreement, shall constitute the Event of Default on the part of defaulting DISCOM:

- (i) DISCOM fails to pay (with respect to a Monthly Bill or a Supplementary Bill), subject to Article 10.5, for a period of ninety (90) days after the Due Date and the RPG is unable to recover the amount outstanding to the RPG through the Letter of Credit,
- (ii) DISCOM repudiates this Agreement and does not rectify such breach even within a period of sixty (60) days from a notice from the RPG in this regard; or
- (iii) except where due to any RPG's failure to comply with its obligations, DISCOM is in material breach of any of its obligations pursuant to this Agreement, and such material breach is not rectified by DISCOM within sixty (60) days of receipt of notice in this regard from the RPG to DISCOM; or

if

A DISTA

- DISCOM becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding up proceedings and such proceedings remain uncontested for a period of sixty (60) days, or
- any winding up or bankruptcy or insolvency order is passed against DISCOM, or
- DISCOM goes into liquidation or dissolution or a receiver or any similar officer is appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, provided that it shall not constitute a DISCOM Event of Default, where such dissolution or liquidation of DISCOM or DISCOM is for the purpose of a merger, consolidation or reorganization and where the resulting entity has the financial standing to perform its obligations under this Agreement and has creditworthiness similar to DISCOM and expressly assumes all obligations of DISCOM and is in a position to perform them; or;

(iv) Occurrence of any other event which is specified in this Agreement to be a material breach or default of DISCOM.

13.3 Procedure for cases of RPG Event of Default

13.3.1 Upon the occurrence and continuation of any RPG Event of Default under Article 13.1, DISCOM shall have the right to deliver to the RPG, with a copy to the representative of the lenders to the RPG with whom the RPG has executed the Financing Agreements, a notice stating its intention to terminate this Agreement (DISCOM Preliminary Default Notice), which shall specify in reasonable detail, the circumstances giving rise to the issue of such notice.

13.3.2 Following the issue of a DISCOM Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply and it shall be the responsibility of the Parties to discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Event of Default having regard to all the circumstances.

13.3.3 During the Consultation Period, the Parties shall continue to perform their respective obligations under this Agreement.

13.3.4 Within a period of seven (7) days following the expiry of the Consultation Period unless the Parties shall have otherwise agreed to the contrary or the RPG Event of Default giving rise to the Consultation Period shall have ceased to exist or shall have been remedied, DISCOM may terminate this Agreement by giving a written Termination Notice of sixty (60) days to the RPG.

13.3.5 Subject to the terms of this Agreement, upon occurrence of a RPG Event of Default under this Agreement, the lenders in concurrence with the DISCOM, may exercise their rights, if any, under Financing Agreements, to seek substitution of the RPG by a selectee for the residual period of the Agreement, for the purpose of securing the payments of the total debt amount from the RPG and performing the obligations of the RPG. However, in the event the lenders are unable to substitute the defaulting RPG within the stipulated period, DISCOM may terminate the PPA and may acquire the Project assets for an amount equivalent to 90% of the debt due or less as mutually agreed, failing which, the lenders may exercise their mortgage rights and liquidate the Project assets.

Provided that any substitution under this Agreement can only be made with the prior consent of DISCOM including the condition that the selectee meets the eligibility requirements of Request for Selection (RfS) issued by DISCOM and accepts the terms and conditions of this Agreement.

13.3.6 The lenders in concurrence with DISCOM, may seek to exercise right of substitution under Article 13.3.5 by an amendment or novation of the PPA in favour of the selectee. The RPG shall cooperate with DISCOM to carry out such substitution and shall have the duty and obligation to continue to operate the Power Project in accordance with this PPA till such time as the substitution is finalized. In the event of Change in Shareholding/Substitution of Promoters triggered by the Financial Institutions leading to signing of fresh PPA with a new entity, an amount of Rs. 1 Lakh per





MW +18% GST per transaction as facilitation fee (non-refundable) shall be deposited by the RPG to DISCOM.

13.3.7 In the event the lenders are unable to substitute the defaulting RPG within the stipulated period, DISCOM may terminate the PPA and may acquire the Project assets for an amount equivalent to 90% of the debt due, failing which, the lenders may exercise their mortgage rights and liquidate the Project assets.

13.4 Procedure for cases of DISCOM Event of Default

13.4.1 Upon the occurrence and continuation of any DISCOM Event of Default specified in Article 13.2, the RPG shall have the right to deliver to DISCOM, a RPG Preliminary Default Notice, which notice shall specify in reasonable detail the circumstances giving rise to its issue.

13.4.2 Following the issue of a RPG Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply and it shall be the responsibility of the Parties to discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Event of Default having regard to all the circumstances.

13.4.3 During the Consultation Period, the Parties shall continue to perform their respective obligations under this Agreement.

13.4.4 After a period of two hundred ten (210) days following the expiry of the Consultation Period and unless the Parties shall have otherwise agreed to the contrary or DISCOM Event of Default giving rise to the Consultation Period shall have ceased to exist or shall have been remedied, DISCOM under intimation to RPG shall, subject to the prior consent of the RPG, novate its part of the PPA to any third party, including its Affiliates within the stipulated period. In the event the aforesaid novation is not acceptable to the RPG, or if no offer of novation is made by DISCOM within the stipulated period, then the RPG may terminate the PPA and at its discretion require DISCOM to either (i) takeover the Project assets by making a payment of the termination compensation equivalent to the amount of the debt due and 150% (one hundred and fifty per cent) of the adjusted equity or, (ii) pay to the RPG, damages, equivalent to 6 (six) months, or balance PPA period whichever is less, of charges for its contracted capacity, with the Project assets being retainedby the RPG.

Provided further that at the end of three (3) months period from the period mentioned in this Article 13.4.4, this Agreement may be terminated by the RPG.

13.5 Termination due to Force Majeure

13.5.1 If the Force Majeure Event or its effects continue to be present beyond a period as specified in Article 4.4.2, either Party shall have the right to cause termination of the Agreement. In such an event this Agreement shall terminate on the date of such Termination Notice without any further liability to either Party from the date of such termination.



ARTICLE 14: LIABILITY AND INDEMNIFICATION

14.1 Indemnity

14.1.1 The RPG shall indemnify, defend and hold DISCOM harmless against:

- a) any and all third party claims against DISCOM for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by the RPG of any of its obligations under this Agreement; and
- b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest actually suffered or incurred by DISCOM from third party claims arising by reason of a breach by the RPG of any of its obligations under this Agreement, (provided that this Article 14 shall not apply to such breaches by the RPG, for which specific remedies have been provided for under this Agreement).

14.1.2 DISCOM shall indemnify, defend and hold the RPG harmless against:

- any and all third party claims against the RPG, for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by DISCOM of any of their obligations under this Agreement; and
- b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest ('Indemnifiable Losses') actually suffered or incurred by the RPG from third party claimsarising by reason of a breach by DISCOM of any of its obligations.

14.2 Procedure for claiming Indemnity

14.2.1 Third party claims

a. Where the Indemnified Party is entitled to indemnification from the Indemnifying Party pursuant to Article 14.1.1(a) or 14.1.2(a), the Indemnified Party shall promptly notify the Indemnifying Party of such claim referred to in Article 14.1.1(a) or 14.1.2(a) in respect of which it is entitled to be indemnified. Such notice shall be given as soon as reasonably practicable after the Indemnified Party becomes aware of such claim. The Indemnifying Party shall be liable to settle the indemnification claim within thirty (30) days of receipt of the above notice. Provided however that, if:

- i) the Parties choose to refer the dispute before the Arbitrator in accordance with Article 16.3.2; and
- ii) the claim amount is not required to be paid/ deposited to such third party pending the resolution of the Dispute,

the Indemnifying Party shall become liable to pay the claim amount to the Indemnified Party or to the third party, as the case may be, promptly following the resolution of the Dispute, if such Dispute is not settled in favour of the Indemnified Party.



b. The Indemnified Party may contest the claim by referring to the Arbitrator for which it is entitled to be Indemnified under Article 14.1.1(a) or 14.1.2(a) and the Indemnifying Party shall reimburse to the Indemnified Party all reasonable costs and expenses incurred by the Indemnified party. However, such Indemnified Party shall not settle or compromise such claim without first getting the consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed.

An Indemnifying Party may, at its own expense, assume control of the defence of any proceedings brought against the Indemnified Party if it acknowledges its obligation to indemnify such Indemnified Party, gives such Indemnified Party prompt notice of its intention to assume control of the defence, and employs an independent legal counsel at its own cost that is reasonably satisfactory to the Indemnified Party.

14.3 Indemnifiable Losses

14.3.1 Where an Indemnified Party is entitled to Indemnifiable Losses from the Indemnifying Party pursuant to Article 14.1.1(b) or 14.1.2(b), the Indemnified Party shall promptly notify the Indemnifying Party of the Indemnifiable Losses actually incurred by the Indemnified Party. The Indemnifiable Losses shall be reimbursed by the Indemnifying Party within thirty (30) days of receipt of the notice seeking Indemnifiable Losses by the Indemnified Party. In case of nonpaymentof such losses after a valid notice under this Article 14.3, such event shall constitute a payment default under Article 13.

14.4 Limitation on Liability

14.4.1 Except as expressly provided in this Agreement, neither the RPG nor its/ their respective officers, directors, agents, employees or affiliates (or their officers, directors, agents or employees), shall be liable or responsible to the other Party or its affiliates, officers, directors, agents, employees, successors or permitted assigns or their respective insurers for incidental, indirect or consequential damages, connected with or resulting from performance or non-performance of this Agreement, or anything done in connection herewith, including claims in the nature of lost revenues, income or profits (other than payments expressly required and properly due under this Agreement), any increased expense of, reduction in or loss of power generation or equipment used therefore, irrespective of whether such claims are based upon breach of warranty, tort (including negligence, whether of DISCOM, the RPG or others), strict liability, contract, breach of statutory duty, operationof law or otherwise.

14.4.2 DISCOM shall have no recourse against any officer, director or shareholder of the RPG or any Affiliate of the RPG or any of its officers, directors or shareholders for such claims excluded under this Article. The RPG shall have no recourse against any officer, director or shareholder of DISCOM, or any affiliate of DISCOM or any of its officers, directors or shareholders for such claims excluded under this Article.



14.5 Duty to Mitigate

14.5.1 The Parties shall endevour to take all reasonable steps so as mitigate any loss or damage which has occurred under this Article 14.

ARTICLE 15: ASSIGNMENTS AND CHARGES

15.1 Assignments

This Agreement shall be binding upon, and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement shall not be assigned by any Party, except to the Project Lenders or Lender's Representative as security for their debt under the Financing Agreements, other than by mutual consent between the Parties to be evidenced in writing. Such assignment shall be agreed to by DISCOM subject to the compliance of provisions contained in this Agreement and more specifically to the provisions of Article 4.1.1 of this Agreement. In no case, such assignment shall be permissible prior to the declaration of COD.

Provided that, DISCOM shall permit assignment of any of RPG's rights and obligations under this Agreement in favour of the lenders to the RPG, if required under the Financing Agreements. Provided that, such consent shall not be withheld if DISCOM seeks to transfer to any transferee all ofits rights and obligations under this Agreement.

The enforcement of the rights and obligation between the RPG and the DISCOM provided in this Agreement shall not be treated as an assignment but an enforcement of the terms agreed under this Agreement.

Provided further that any successor(s) or permitted assign(s) identified after mutual agreement between the Parties may be required to execute a new agreement on the same terms and conditions as are included in this Agreement. An amount of Rs. 1 Lakh per Transaction as Facilitation Fee (nonrefundable) shall be deposited by the RPG to DISCOM. Provided further that, such consent shall not be withheld by the RPG if DISCOM seeks to transfer to any affiliate all of its rights and obligations under this Agreement.

In the event of Change in Shareholding/Substitution of Promoters triggered by the Financial Institutions leading to signing of fresh PPA with a New Entity, an amount of Rs. 1 Lakh per Transaction as Facilitation Fee (non-refundable) shall be deposited by the RPG to DISCOM.

15.2 Permitted Charges

15.2.1 RPG shall not create or permit to subsist any encumbrance over all or any of its rights and benefits under this Agreement, other than as set forth in Article 15.1 and the Guidelines. ARTICLE 16: GOVERNING LAW AND DISPUTE RESOLUTION

16.1 Governing Law

16.1.1 This Agreement shall be governed by and construed in accordance with the Laws of India. Any legal proceedings in respect of any matters, claims or disputes under this Agreement shall be under the jurisdiction of appropriate courts in_____.



16.2 Amicable Settlement and Dispute Resolution

16.2.1 Amicable Settlement

- i. Either Party is entitled to raise any claim, dispute or difference of whatever nature arising under, out of or in connection with this Agreement ("Dispute") by giving a written notice (Dispute Notice) to the other Party, which shall contain:
 - (a) a description of the Dispute;
 - (b) the grounds for such Dispute; and
 - (c) all written material in support of its claim.
- ii. The other Party shall, within thirty (30) days of issue of Dispute Notice issued under Article 16.2.1(i), furnish:
 - (a) counter-claim and defences, if any, regarding the Dispute; and
 - (b) all written material in support of its defences and counter-claim.
- iii. Within thirty (30) days of issue of Dispute Notice by any Party pursuant to Article 16
 - (i) if the other Party does not furnish any counter claim or defence under Article 16
 - (ii) or thirty (30) days from the date of furnishing counter claims or defence by the other Party, both the Parties to the Dispute shall meet to settle such Dispute amicably. If the Parties fail to resolve the Dispute amicably within thirty (30) days from the later of the dates mentioned in this Article 16.2.1.
 - (iii) the Dispute shall be referred for dispute resolution in accordance with Article 16.3.

16.3 Dispute Resolution

16.3.1 Dispute Resolution by the Appropriate Commission

- i) Where any Dispute or differences arises in relation to this agreement of any nature whatsoever including the construction, interpretation or implementation of the provisions of this agreement as well as claim made by any Party for any change in or determination of the Tariff or any matter related to Tariff or claims made by any Party which partly or wholly relate to any change in the Tariff or determination of any of such claims could result in change in the Tariff, and relates to any matter agreed to be referred to the Appropriate Commission, shall be submitted to adjudication by the Appropriate Commission. Appeal against the decisions of the Appropriate Commission shall be made only as per the provisions of the Electricity Act, 2003, as amended from time to time.
- ii) DISCOM shall be entitled to co-opt the lenders (if any) as a supporting party in such proceedings before the Appropriate Commission.

16.3.2 Dispute Resolution through Arbitration

PCODI

i) If the Dispute arising as per Article 16.2.1 is not amicably resolved & such dispute is not covered in Article 16.3.1(i), such Dispute shall be resolved by arbitration under the provisions of the Electricity Act, 2003 (as amended from time to time) as under: Proceedings as well as appointment of the arbitrator(s) shall be carried out by the Appropriate Commissions under the Electricity Act 2003 as amended from time to time. As stipulated by the said Electricity Act 2003, the said arbitration will take place as per the provisions of the Arbitration and ConciliationAct 1996 as amended from time to time.

- ii) The place of arbitration shall be the _____(City where head quarter of DISCOM is located). The language of the arbitration shall be English.
- iii) The Arbitration Tribunal's award shall be substantiated in writing. The Arbitration Tribunal shall also decide on the costs of the arbitration proceedings and the allocation thereof.
- iv) The provisions of this Article shall survive the termination of this PPA for any reason whatsoever.
- v) The award shall be of majority decision.
- vi) DISCOM shall be entitled to co-opt the lenders (if any) as a supporting party in such arbitration proceedings.

16.4 Parties to Perform Obligations

16.4.1 Notwithstanding the existence of any Dispute and difference referred to the Appropriate Commission and save as the Appropriate Commission may otherwise direct by a final or interim order, the Parties hereto shall continue to perform their respective obligations (which are not in dispute) under this Agreement.

ARTICLE 17: MISCELLANEOUS PROVISIONS

17.1 Amendment

17.1.1 This Agreement may only be amended or supplemented by a written agreement between the Parties.

17.2 Third Party Beneficiaries

17.2.1 This Agreement is solely for the benefit of the Parties and their respective successors and permitted assigns and shall not be construed as creating any duty, standard of care or any liability to, any person not a party to this Agreement.

17.3 Waiver

17.3.1 No waiver by either Party of any default or breach by the other Party in the performance of any of the provisions of this Agreement shall be effective unless in writing duly executed by an authorised representative of such Party.

17.3.2 Neither the failure by either Party to insist on any occasion upon the performance of the terms, conditions and provisions of this Agreement nor time or other indulgence granted by one Party to the other Parties shall act as a waiver of such breach or acceptance of any variation or the relinquishment of any such right or any other right under this Agreement, which shall remain in full force and effect.



17.4 Confidentiality

17.4.1 The Parties undertake to hold in confidence this Agreement and not to disclose the terms and conditions of the transaction contemplated hereby to third parties, except:

a) to their professional advisors;

b) to their officers, contractors, employees, agents or representatives, financiers, who need to have access to such information for the proper performance of their activities; or

c) disclosures required under Law, without the prior written consent of the other Party.

17.5 Severability

17.5.1 The invalidity or unenforceability, for any reason, of any part of this Agreement shall not prejudice or affect the validity or enforceability of the remainder of this Agreement, unless the part held invalid or unenforceable is fundamental to this Agreement.

17.6 Notices

17.6.1 All notices or other communications which are required to be given under this Agreement shall be in writing and in the English language.

17.6.2 If to the RPG, all notices or other communications which are required must be delivered personally or by registered post or facsimile or any other method duly acknowledged to the addresses below:

Address: Attention: Email: Fax. No. : Telephone No. :

17.6.3 If to DISCOM, all notices or communications must be delivered personally or by registered post or facsimile or any other mode duly acknowledged to the address(es) below: Address:

Attention: Email: Fax. No. : Telephone No. :

17.6.4 All notices or communications given by facsimile shall be confirmed by sending a copy of the same via post office in an envelope properly addressed to the appropriate Party for delivery by registered mail. All notices shall be deemed validly delivered upon receipt evidenced by an acknowledgement of the recipient, unless the Party delivering the notice can prove in case of delivery through the registered post that the recipient refused to acknowledge the receipt of the notice despite efforts of the postal authorities.



17.6.5 Any Party may by notice of at least fifteen (15) days to the other Party change the address and/or addresses to which such notices and communications to it are to be delivered or mailed.

17.7 Language

17.7.1 All agreements, correspondence and communications between the Parties relating to this Agreement and all other documentation to be prepared and supplied under the Agreement shall be written in English, and the Agreement shall be construed and interpreted in accordance with English language.

17.7.2 If any of the agreements, correspondence, communications or documents are prepared in any language other than English, the English translation of such agreements, correspondence, communications or documents shall prevail in matters of interpretation.

17.8 Restriction of Shareholders / Owners' Liability

17.8.1 Parties expressly agree and acknowledge that none of the shareholders of the Parties hereto shall be liable to the other Parties for any of the contractual obligations of the concerned Party under this Agreement. Further, the financial liabilities of the shareholder/s of each Party to this Agreement, shall be restricted to the extent provided in the Indian Companies Act, 2013.

17.9 Taxes and Duties

17.9.1 The RPG shall bear and promptly pay all statutory taxes, duties, levies and cess, assessed/ levied on the RPG, contractors or their employees that are required to be paid by the RPG as per the Law in relation to the execution of the Agreement and for supplying power as per the terms of this Agreement.

17.9.2 DISCOM shall be indemnified and held harmless by the RPG against any claims that may be made against DISCOM in relation to the matters set out in Article 17.9.1.

17.9.3 DISCOM shall not be liable for any payment of, taxes, duties, levies, cess whatsoever for discharging any obligation of the RPG by DISCOM on behalf of RPG.

17.10 Independent Entity

17.10.1 The RPG shall be an independent entity performing its obligations pursuant to the Agreement.

17.10.2 Subject to the provisions of the Agreement, the RPG shall be solely responsible for the manner in which its obligations under this Agreement are to be performed. All employees and representatives of the RPG or contractors engaged by the RPG in connection with the performance of the Agreement shall be under the complete control of the RPG and shall not be deemed to be employees, representatives, contractors of DISCOM and nothing contained in the Agreement or in any agreement or contract awarded by the RPG shall be construed to create any contractual relationship between any such employees, representatives or contractors and DISCOM.



17.11 Compliance with Law

Despite anything contained in this Agreement but without prejudice to this Article, if any provision of this Agreement shall be in deviation or inconsistent with or repugnant to the provisions contained in the Electricity Act, 2003, or any rules and regulations made there under, such provision of this Agreement shall be deemed to be amended to the extent required to bring it into compliance with the aforesaid relevant provisions as amended from time to time.

17.12 Breach of Obligations

The Parties acknowledge that a breach of any of the obligations contained herein would result in injuries. The Parties further acknowledge that the amount of the liquidated damages or the method of calculating the liquidated damages specified in this Agreement is a genuine and reasonable preestimate of the damages that may be suffered by the non-defaulting party in each case specified under this Agreement.

IN WITNESS WHEREOF the Parties have caused the Agreement to be executed through their duly authorized representatives as of the date and place set forth above.

For and on behalf of [DISCOM]

For and on behalf of [RPG]

Name, Designation and Address	Name, Designation and Address
Signature with seal	Signature with seal
Witness:	Witness:
1.	I.
2.	2.



Guidelines for Installation of Innovative Stand-alone Solar Pumps

These guidelines are applicable for all Indian innovators/manufacturers/service providers, who wish to install innovative stand-alone solar pumps in the country under Schemes operated by the MNRE.

I. Call of Expression of Interest (EoI)

a) The Ministry will call EoI from time to time for inviting application from solar pump innovators claiming better performance on account of improved efficiency, cost effectiveness, monitoring, other value addition, etc. Only those innovative products which are available for testing and field trial will be eligible for participation in the EoI.

b) New technologies for which patent/IP related filings have been done (patent may not have been awarded) will also be eligible to participate in the EoI. Organization which has filed the patent should be the sole applicant or lead partner in case of a consortium/JV.

c) The elements of innovation in the context of improved performance of solar pump needs to be mentioned very clearly by the applicants in their response to EoI. A cost benefit analysis shall also be enclosed with the proposal.

II. Evaluation of Applications

Applications will be evaluated by an Evaluation Committee constituted by MNRE having technical and financial experts of the sector including academic experts. Test report for the proposed solar pump will be submitted by the innovator along with EoI, however, if deemed necessary, the Committee may recommend re-testing of performance of the pump at NISE or any other NABL accredited lab before allowing installation of such pumps in the field. In cases, where the technology is in its nascent stage, special method may be adopted to test the pumps in consultation with the innovator and NISE. The expenditure incurred on testing will be borne by the applicant.



III. Demonstration of the Innovation

a) On recommendations of the Committee, the innovative technology will be allowed for demonstration in the field after getting consent from respective state implementing agency (SIA) and the beneficiary farmers by the innovator. The innovator will be allowed to install up to 50 solar pumps in different parts of the State/UT for demonstration purpose.

b) For payment (subsidy and beneficiary share), AMC, performance guarantee, etc., the innovators will be treated at par with other vendors selected through competitive bidding process and awarded work for installation of solar water pumps in the respective state/UT and the innovators will not get any additional cost over and above the cost discovered through tendering process. Further, the innovators is required to deposit bank guarantee equivalent to payment made to him against the installation of solar pump, which will be encashed, if the pump is failed to meet at least the existing MNRE performance standards during demonstration period of one year. In addition, the innovators have to fill-up indemnity bond against damages caused to farmer due to under performance of pump during the period of demonstration. On successful trail of pump, the indemnity bond and bank guarantee would be released, however, the performance guarantee as applicable during the period of AMC will be retained by the SIA.

c) The field demonstration will continue for at least one year from date of installation of such pumps and during that period the performance of the pump will be monitored on real time basis. Access to remote monitoring facility for the installed pumps shall be provided to NISE so that field data may be accumulated and evaluated.

d) Innovator will submit on monthly basis a detailed performance report of the pump including a comparative analysis with similar capacity pump specified by the MNRE, along with feedback from SIA & beneficiary farmers to the Evaluation Committee. MNRE/NISE will also conduct field inspection to check the performance as and when required.

e) Installed innovative pumps shall meet at least the existing MNRE performance standards. However, only the innovations showing substantial improvement from the MNRE specifications may be considered for adoption. A comparative analysis of performance of innovative technology vis-à-vis MNRE specifications will also be done by the Evaluation Committee.



f) After detailed examination and deliberations, the Evaluation Committee may recommend for adoption of specifications of innovative pump, with certain modifications, if required.

IV. Adoption of technology by MNRE

On recommendations of the Evaluation Committee, the MNRE may adopt the innovative technology and update the existing specification after having detailed stakeholders' consultation on the same. For innovations with outstanding performance the innovators will get certificate of appreciation from MNRE.



		Annexure-8	
Date : 30/03/2024	TPCODL	Tender Ref No	
	Minutes of Meeting TPO	CODL/P&S/1000000494/2023-2	
Background information in brief	Request for Proposal (RFP) for Selection of Solar Power Generators (SPG for Solarisation of Agricultural Feeders in Odisha under PM-KUSUM (Component-C) in RESCO mode.		
	 BA agreed to complete understanding terms and conditions of the Tender. 	; of Scope of Work and all th	
	 Derived Capacity will only be allocat Qualification Criteria. Back up calculation of the discovered U The discovered unit rate consists of all additional cost of UU 	nit Rate is available with BA.	
Summary of Discussions	Qualification Criteria. 3. Back up calculation of the discovered U	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible.	
Summary of Discussions	 Qualification Criteria. Back up calculation of the discovered U The discovered unit rate consists of all additional cost shall be entertained. The BA informed that further reduction 	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible.	
Summary of Discussions	 Qualification Criteria. Back up calculation of the discovered U The discovered unit rate consists of all additional cost shall be entertained. The BA informed that further reduction Final Discovered L1 rates is tabulated be Particulars 	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible. elow: (INR/kWH)	
Summary of Discussions	 Qualification Criteria. Back up calculation of the discovered U The discovered unit rate consists of all additional cost shall be entertained. The BA informed that further reduction Final Discovered L1 rates is tabulated be Particulars WITH DCR If land is purchased /taken on lease by the Developer mutually agreed lease rent 	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible. elow: (INR/kWH) 3.75	
Summary of Discussions	Qualification Criteria. 3. Back up calculation of the discovered U 4. The discovered unit rate consists of all additional cost shall be entertained. 5. The BA informed that further reduction 6. Final Discovered L1 rates is tabulated be Particulars WITH DCR If land is purchased /taken on lease by the	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible. elow: (INR/kWH) 3.75	
Summary of Discussions	Qualification Criteria. 3. Back up calculation of the discovered U 4. The discovered unit rate consists of all additional cost shall be entertained. 5. The BA informed that further reduction 6. Final Discovered L1 rates is tabulated be WITH DCR If land is purchased /taken on lease by the Developer mutually agreed lease rent If land is provided by Government on nom	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible. elow: (INR/kWH) 3.75 inal 3.65	
Summary of Discussions	Qualification Criteria. 3. Back up calculation of the discovered U 4. The discovered unit rate consists of all additional cost shall be entertained. 5. The BA informed that further reduction 6. Final Discovered L1 rates is tabulated be Particulars WITH DCR If land is purchased /taken on lease by the Developer mutually agreed lease rent If land is provided by Government on nom lease rent	nit Rate is available with BA. the cost (SITC, Taxes, etc.). No in unit rate is not possible. elow: (INR/kWH) 3.75 inal 3.65 R 3.35	

O'ishow

SCM SOLAR ENERGY (OPC) PRIVATE LIMITED 135, Soubhagya Nagar, Siripur, BHUBANESWAR-751 003. Cell: 9676058005



Dat	e : 30/03/2024	TPCODL	Tender Ref No
		Minutes of Meeting	TPCODL/P&S/1000000494/2023-;
Subject Venue MOM Prep	: Online N		P&5/1000000494/2023-24 for KUSUSN
		Committee Member	
Sr. No.	Name		Signature
1.	Mr. Ashok Choudhury (Renewable)		
2.	Ms. Sony Jha (Procur	ement)	
3.	Ms. Pratima Singh (P	re-Audit)	
4.	Mr. Arijeet Choudhu	y (Procurement)	
	Business Associate	M/s SCM Solar Energy(OPC) Pvt. Ltd	
1.	Mr. Umesh Mishra		Oithree.

SCM SOLAR ENERGY (OPC) PRIVATE LIMITED 135, Soubhagya Nagar, Siripur BHUBANESWAR-751 003. Cell 9676058005

Points of Discussion:

A DISTRI AUGANES

32/645/2017-SPV Division Government of India Ministry of New and Renewable Energy

Annexure-6

(Dr. Nikhil Gakkhar)

Scientist D 011-20849115

Atal Akshay Urja Bhavan Lodhi Road, New Delhi 110003

Date: 14th February 2024

Office Memorandum

Subject: Clarification regarding new Comprehensive guidelines of PM KUSUM -reg.

In continuation of this Ministry's Office Memorandum of even number dated 17.01.2024 regarding comprehensive guidelines for implementation of PM-KUSUM Scheme, and Ministry's OM dated 29.01.2024, it clarified that prevailing CFA under this component would be valid till any update is announced by MNRE.

2. In view of the representation received from SIAs it is again clarified that

a) CFA of Rs 1.05 Cr/MW would be valid for all those LoAs, till any further update is announced by MNRE.

b) The DCR exemption would be valid for all those Letter of Awards which have been issued on or before 31.03.2024.

3. All the SIA's are requested to make a note of the above clarification and are requested to expediate the implementation of PM KUSUM Scheme. Further the progress made under each of the PM KUSUM scheme components shall be submitted, duly signed, to this Ministry on a monthly basis (Formats enclosed)

This is with the approval of Competent Authority.

All the concerned SIAs

To

5 7 202



GOVERNMENT OF ODISHA DEPARTMENT OF ENERGY

Dated.

No. _____/RE&ET. ENG-RE&ET-RE-0005-2024

From

Sri Sambit Parija, OFS (SS), FA-cum-Special Secretary to Government

То

The Additional Secretary & Chief Executive Officer, OREDA Ltd., / EIC(Elect.)-cum-PCEI, Odiska, Bhubaneswar/ CPM & HOD, REC, Bhubaneswar/ Managing Director, GRIDCO / Director (F&CA), GRIDCO/ Managing Director, OHPC / Dy. General Manager, GEDCOL/ Chief Regulatory, TATA Power/ Chief Executive Officers, TPCODL/TPWODL/TPNODL/TPSODL

Sub: Approved minutes of the meeting on PM-Surya Ghar : Muft Bijli Yojana scheme and PM-KUSUM scheme held under the Chairmanship of Principal Secretary to Government in the Finance Dept. Confernce Hall on 27.06.2024 at 12.00 Noon.

1.0

Sir,

I am directed to enclose herewith a copy of the approved minutes of the meeting on PM-Surya Ghar: Muft Bijli Yojana scheme and PM-KUSUM scheme held under the Chairmanship of the Principal Secretary to Government, Energy Department in the Finance Department Confernce Hall for favour of your kind information and necessary action.

Encl- As above

Yours faithfully, FA-cum-Specia to Govt. Sec DISTRI (\cdot)

Minutes of the review meeting on PM-Surya Ghar: Muft Bijli Yojana and PM-KUSUM scheme held under the Chairmanship of Principal Secretary to Government, Energy Department on 27.06.2024 at 12.00 Noon in the Conference Hall of Finance Department.

Members present in the meeting is at Annexure-1

At the outset the Principal Secretary to Government, Energy Department welcomed all participants, present in the meeting and requested the TPDISCOMs to present the action taken by the DISCOMs on PM-Surya Ghar. Muft Bijli Yojana.

Principal Secretary, Energy Department observed that out of more than 1 lakh applications, only 30,282 are active applications and only 881 have deposited necessary fee with the DISCOMs for net metering. He expressed his displeasure on the dismal rate of conversion of active applications to executable projects. He stressed on adoption of extensive outreach efforts by the TPDISCOMs to connect with all applicants and guide for depositing net metering fees, select vendors and go ahead with the installation of their RTS systems in Mission Mode.

Sri K.C. Nanda, G.M., TPWODL gave a presentation on the action taken and challenges faced while implementing PM-Surya Ghar: Muft Bijli Yojana. CPM, REC was informed about the slow response on the national portal for which the applicants are getting demotivated to get themselves registered.

After detailed discussions the following decisions were taken:

A. PM-Surya Ghar: Muft Bijli Yojona

- Single PBG for all DISCOMs of the State in respect of PM-Surya Ghar: Muft Bijli Yojana will be collected and the TPCODL is nominated to collect the same on behalf of all DISCOMs. Accordingly, a Guideline / SoP to be prepared by TPCODL in line with MNRE guidelines.
- TPDISCOMs to conduct extensive outreach programmes and submit a detailed action plan in the next meeting. Mo SEVA Kendras may be roped in to assist the consumers in registration. The Call Centres maintained by TPDISCOMs be activated. IEC activities taken up by MNRE may also be adopted and customised in Odia. Meeting with E&IT Department to be taken up for taking up such outreach programmes through social media.
- TPCODL to prepare a draft advertisement both for print media and electronic media to create awareness
 among the public about the benefits as well as on the procedures for adopting PM-Surya Ghar. Muft Bijli
 Yojona.
- 4. TPDISCOM to issue mass emails to all Vendors registered in the National Portal to take up works in Odisha. They may be asked to register in this State.
- 5. OERC to be followed up for expediting issuance of necessary notifications for waiver of TFR.
- 6. REC to initiate necessary actions to speed up the dedicated National Portal on PM-Surya Ghar: Muft Bijli Yojona for easy registration by the applicants of the State as well as for vendors registration.
- REC is to take action on integration (Community Solarisation of Rural Households Mechanism) of the provision for State Subsidy of Rs. 20,000 per kw available under Green City Mission for Bhubaneswar in the National Portal.
- REC to expedite the release of CFA to the consumers who have installed & Inspection and PCR has been completed by DISCOMs on their rooftops; this will motivate other consumers to go for the scheme. REC to immediately clear the pending CFAs in this regard.
- 9. Efforts be taken to get the MoU signed between Energy Department and MNRE at the earliest.
- 10. Energy Department to take up a meeting with the Principal Secretary to Govt., Skill Development and Technical Education Department and Director of Technical Education and Training (DTET). Principals of ITI & Diploma colleges may also be requested to encourage the final year students to come forward for vendor registration in respect of PM-Surya Ghar. Muft Bijli Yojana. Further, ITI technicians be roped in as Vendors.
- Energy Dept, will organise a meeting with Industry Associations and TATA Channel partners for extensive coverage of the scheme. TPCODL to prepare a draft notification/ advertisement which is to be circulated among the Industry Associations.
- The Energy Clubs under EIC(Elecy)-cum-PCEI may also be utilised to disseminate the efficacy of the Distance scheme.

PCOD

- Energy Department will take up the issue of approval of proposed community solarisation programme with MNRE.
- 14. Standard Checklist to be developed by REC and TPDISCOMs for easy inspection of installations by DISCOMs.
- A dedicated RE Cell to be constituted comprising of two officials each from EIC and TPDISCOMs for overall monitoring of this scheme. The RE Cell for PMSG-MBY may function in the office of EIC(Elecy)-cum-PCEL
- The progress of PM SG-MBY shall be reviewed once in a month and the next review meeting will be held around 15th July, 2024.

B. PM-KUSUM (A, C1 and C2)

CEO, TPWODL gave a presentation on the progress of PM KUSUM and stated about the challenges faced in implementation of the scheme.

After detailed deliberations, the followings were decided upon:

- DISCOMs to file petition before OERC for approval of the discovered tariffs under PM-KUSUM-A and Feeder Level Solarisation schemes (C2).
- Progress under Individual Pump Solarisation (PM KUSUM C1) is held up due to farmers unwillingness to contribute their share of 40%. DISCOMs suggested, with 30% of CFA if SFA of 60% on Discovered price will be extended then farmers may be interested. World Bank be requested in this regard to submit alternative financing proposals for implementation of Individual Pump Solarisation Programme in line with the Kerala model.
- Energy Department to take up with IPICOL and other relevant Government Departments for fixation of uniform Long term land lease rent across the State for installation of solar power plants under KUSUM schemes.
- 4. Energy Department to explore the possibility of acquiring farmers' land on lease and on-ward subleasing the same to developers for development of solar power systems under KUSUM schemes.
- 5. DISCOMs to intimate the number of agriculture feeder and their respective loads by 14th July, 2024.
- DISCOMs to develop cluster level feeder solarisation projects of sizable capacities. TPWODL to explore the possibility of implementing the scheme on a pilot basis in Bargarh.

The meeting ended with vote of theanks to the Chair.

Principal Secretary to Govt. Energy Department.



Annexure-I

List of participants

- 1. Vishal Kumar Dev, IAS Principal Secretary, Energy Department
- 2. Sambit Parija, OFS FA-cum-Special Secretary, Energy Department
- S.B.K. Pradhan Additional Secretary and Chief Executive Officer, OREDA Ltd.
- Trilochan Panda Managing Director, GRIDCO
- 5. G.B. Swain Director(F&CA), GRIDCO
- Shri Amresh Kumar Managing Director, OHPC
- Subrat Kumar Senapati Dy. General Manager, GEDCOL
- 8. T. Sridhar CPM & HOD, REC-8BSR
- 9. Shri Arvind Singh CEO, TPCODL
- 10. Shri Dwijadas Basak CEO, TPNODL
- 11. Shri Praveen Verma CEO, TPWODL
- 12. Bharat Bhadawat Chief Regulatory, TATA Power
- Shri K.C. Nanda G.M (Regulatory Affairs& Strategy), TPNODL
 Shri R.K. Nayak
 - GM(Solarization), TPNODL
- 15. Shri Binod Bihari Nayak AGM(Renewable), TPSODL
- Shri Soumya Ranjan Aich ACE DIR(Tech), OREDA
- 17. Shri Shadab Usmani Officer(Engg.) REC limited
- 18. Shri Trivendra Kumar Officer(Engg.) REC limited



Annexure-8



MODEL POWER PURCHASE AGREEMENT (PPA)

FOR PROCUREMENT OF [Project capacity] kW SOLAR POWER ON LONG TERM

BASIS, UNDER COMPONENT- C OF "PMKUSUM SCHEME"

Between

[Name of Solar Power Generator (SPG)]

And

GRIDCO Limited

On

[DD MMM YYYY]



This Power Purchase Agreement is made on the [Date] day of [Month] of [Year] at Bhubaneswar, Odisha

Between

[Name of the Solar Power Generator], [a/an] [individual farmer/ group of farmers/ cooperatives/ panchayats/ farmer producer organizations/ water user associations/ Developer (sole proprietor/ firm/ company)] incorporated under [relevant act in India. Ex. The Indian Companies Act, 2013], having its registered office at [registered office address] (hereinafter referred to as "**Solar Power Generator or SPG**", which expression shall, unless repugnant to the context or meaning thereof, be deemed to include its successors and permitted assigns) as a Party of the **First Part**;

And

The SPG and the GRIDCO are individually referred to as a 'Party' and collectively referred to as 'Parties'.

WHEREAS:

- A. The Ministry of New and Renewable Energy (MNRE) has launched a scheme for farmers on 8th March 2019 and issued implementation guidelines on 22nd July 2019.
- B. MNRE has sanctioned a 10,000 pumps to Odisha vide office memorandum No. 32/54/2018-SPV Division dated 18th May 2022 & 5th August 2022, under the said scheme of 8th March 2019. Further, Energy Department, Government of Odisha has distributed the above target amongst the four DISCOMs vide Letter No 8686 dated 12th September 2022. The DISCOMs have authorized TP Central Odisha Distribution Limited (TPCODL) to initiate a selection process of the SPGs based on which TPCODL decided to go ahead with the procurement process for a capacity of 10,000 pumps with aggregate capacity of 74.10 MW in this phase as per the target given by MNRE,Gol.
- C. TPCODL had completed a selection process for the procurement of energy generated from the grid-connected solar Project based on the terms and conditions contained in the RFP No. [RFP No.] dated [DD MMM YYYY] and its Amendment (if any) along with the Draft Power Purchase Agreement and Draft, Model Lease Agreement and any other documents published or referred by TPCODL related to the selection of SPG.
- D. The SPG has been selected in the process for development, generation and supply of electricity from the [Project Capacity] kW solar Project to be established by SPG at [village] Village, [Revenue Inspector Circle] Revenue Inspector (RI) Circle in [district] District of Odisha State and electricity generated to be fed to the [Name and location of 33/11 kV DISCOM substation, exactly as per the list in the RFP] Sub-station;
- E. _____ has issued the Letter of Award no. [LoA no.] dated [DD MMM YYYY] in favour of the SPG for the development and establishment of [Project Capacity] kW solar Project as per the terms and conditions contained in the RFP.
- F. The SPG has furnished the Performance Security in the form of a Bank Guarantee in the sum of INR [Insert amount] (Indian Rupees [in words]) only <u>in favour of TPCODL</u> as per the format Annexure 3 (Bank Guarantee), prescribed by GRIDCO.
- G. The SPG has fulfilled the terms and conditions for signing this Power Purchase Agreement as a definitive agreement for establishing the Project;
- H. The Parties have agreed to execute this Power Purchase Agreement, and for generation and supply of electricity by the SPG to GRIDCO.

Now therefore, in consideration of the premises and mutual agreements, covenants and conditions set forth herein, it is hereby agreed by and between the Parties as follows:



1. ARTICLE 1: DEFINITIONS AND INTERPRETATION

1.1. Definitions

The terms used in this Agreement, unless as defined below or repugnant to the context, shall have the same meaning as assigned to them by the Electricity Act, 2003 and the rules or regulations framed thereunder, including those issued/ framed by the Appropriate Commission (as defined hereunder), as amended or re-enacted from time to time.

"Act" or "Electricity Act, 2003"	shall mean the Electricity Act, 2003 and include any modifications, amendments, and substitution from time to time;	
"Agreement" or "Power Purchase Agreement" or "PPA"	shall mean this power purchase agreement including its recitals and schedules, amended or modified from time to time in accordance with the terms hereof;	
"Appropriate Commission"	shall mean Odisha Electricity Regulatory Commission (OERC);	
" DISCOM"	 shall mean the DISCOM having the Sub-station is located within its jurisdiction and the name of the various DISCOMs in Odisha are given below: TP Central Odisha Distribution Limited (TPCODL) TP Western Odisha Distribution Limited (TPWODL) TP Southern Odisha Distribution Limited (TPSODL) TP Northern Odisha Distribution Limited (TPNODL) 	
"GRIDCO"	M/s GRIDCO Ltd.	
"Bill"	shall mean either a Monthly Bill/ Supplementary Bill raised by the SPG;	
"Bill Dispute Notice"	shall mean the notice issued by raising a Dispute regarding a Monthly Bill or a Supplementary Bill issued by the SPG;	
"Business Day"	shall mean with respect to SPG and DISCOM, a day other than Sunday or a statutory holiday, on which the banks remain open for business in the State;	
"Capacity Utilization Factor" or "CUF"	shall have the same meaning as provided in CERC (Terms and Conditions for Tariff determination from Renewable Energy Sources) Regulations, 2009 as amended from time to time; However, for avoidance of any doubt, it is clarified that the CUF shall be calculated on the Contracted Capacity;	
	In any Contract Year, if 'X' MWh of energy has been metered out at the Delivery Point for 'Y' MW Project's Contracted Capacity,	
	CUF= (X MWh/ (Y MW*8,766)) *100%;	
"Change in Law"	shall have the meaning ascribed thereto in Article 12 of this Agreement;	
"Commercial Operation Date (COD)"	shall mean the date mentioned in the commissioning certificate as issued upon successful achievement of commercial operation date (as per provisions of this PPA) of the Project;	
"Commissioning Committee"	shall consist of representations from Principal Chief Electrical Inspector, DISCOM, SPG, and SPG's representative, as required;	
"Competent Court of Law"	shall mean any court or tribunal or any similar judicial or quasi-judicial body in India that has jurisdiction to adjudicate upon issues relating to this Agreement;	



"Consents, Clearances and Permits"	s shall mean all authorizations, licenses, approvals, registrations, permit waivers, privileges, acknowledgements, agreements, or concessions require to be obtained from or provided by any concerned authority for the purpose setting up of the generation facilities and/ or supply of energy;	
"Consultation Period"	shall mean the period of ninety (90) days or such other longer period as the Parties may agree, commencing from the date of issuance of an SPG Preliminary Default Notice or GRIDCO Preliminary Default Notice as provided in Article 13 of this Agreement, for consultation between the Parties to mitigate the consequence of the relevant event having regard to all the circumstances;	
"Contract Year"	shall mean the period beginning from the Effective Date and ending on the immediately succeeding March 31 and thereafter each period of twelve (12) months beginning on April 1 and ending on March 31 provided that:	
	i. in the financial year in which the COD would occur, the Contract Year shall end on the date immediately before the COD and a new Contract Year shall commence once again from the COD and end on the immediately succeeding March 31, and thereafter each period of twelve (12) months commencing on April 1 and ending on March 31, and	
	 provided further that the last Contract Year of this Agreement shall end on the last day of the Term of this Agreement 	
"Contracted Capacity"	shall mean [Insert capacity] kW contracted with GRIDCO for supply of energy from the Project by the SPG to DISCOM/GRIDCO at the Delivery Point;	
"Contracted Energy"	shall have the meaning ascribed thereto in Article 4.3.1 of this Agreement;	
"Delivery Point"	shall mean the point at the voltage level of 11kV or above of the [Name and location of 33/11 kV DISCOM substation, exactly as per the list in the RFP] Sub-station. Metering shall be done at this interconnection point where the power is injected into the [Name and location of 33 /11 kV DISCOM substation, exactly as per the list in the RFP] Sub-station. For interconnection with grid and metering, the SPG shall abide by the relevant and applicable regulations, Grid Code notified by the State Commission and Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006 as amended and revised from time to time, or orders passed thereunder by the Appropriate Commission or CEA. All charges and losses related to Transmission of power from project up to Delivery Point as notified by the Appropriate Commission shall be borne by the SPG;	
"Dispute"	shall mean any dispute or difference of any kind between GRIDCO/DISCOM and the SPG, in connection with or arising out of this Agreement including but not limited to any issue on the interpretation and scope of the terms of this Agreement as provided in Article 16 of this Agreement;	
"Due Date"	shall mean the thirtieth (30 th) day after a Monthly Bill (including all the relevant documents) or a Supplementary Bill is received in hard copy and duly acknowledged by GRIDCO or, if such day is not a Business Day, the immediately succeeding Business Day, by which date such Monthly Bill or a Supplementary Bill is payable by GRIDCO;	
'Effective Date"	shall have the meaning ascribed thereto in Article 2.1.1 of this Agreement;	
'Electricity Laws*	shall mean the Electricity Act, 2003 and the rules and regulations made thereunder from time to time along with amendments thereto and replacements thereof and any other Law pertaining to electricity including regulations framed by the Appropriate Commission;	
Event of Default"	shall mean the events as defined in Article 13 of this Agreement;	

CONTRAL OD

SHAUCAMESHAR

"Expiry Date"	Shall mean the date occurring twenty-five (25) years from the Commercial Operation Date subject to that the supply of power shall be limited for a period of twenty-five (25) years from the COD unless extended by the Parties as per this Agreement;	
"Financing Agreements"	shall mean the agreements pursuant to which the SPG has sought financing for the Project including the loan agreements, security documents, notes, indentures, security agreements, letters of credit and other documents, as may be amended, modified, or replaced from time to time, but without in anyway increasing the liabilities of DISCOM/GRIDCO;	
"Force Majeure" or "Force Majeure Event"	shall have the meaning ascribed thereto in Article 11 of this Agreement;	
"Indian Governmental Instrumentality"	shall mean the Government of India, Governments of Odisha and any ministry, department, board, authority, agency, corporation, commission under the direct or indirect control of Government of India or the above state Government or both, any political subdivision of any of them including any court or Appropriate Commission or tribunal or judicial or quasi-judicial body in India;	
"Insurances"	shall mean the insurance cover to be obtained and maintained by the SPG in accordance with Article 8 of this Agreement;	
"Interconnection Facilities"	shall mean the facilities on SPG's side of the Delivery Point for scheduling, transmitting and metering the electrical output in accordance with this Agreement and which shall include, without limitation, all other transmission lines and associated equipment, transformers, relay and switching equipment and protective devices, safety equipment and RTU, Data Transfer and Acquisition facilities for transmitting data subject to Article 7, the Metering System required for supply of power as per the terms of this Agreement;	
"Indian Rupees ", "INR"	shall mean Indian rupees, the lawful currency of India;	
'Joint Meter Reading" or "JMR" or "EBC energy export statement"	shall mean the monthly joint meter reading or Energy Billing Center of GRIDCO/DISCOM's energy export statement;	
"Land Lease Agreement"	Shall mean the land lease agreement signed between the 'Individual farmer/ Group of farmers/ Cooperatives/ Panchayats/ Farmer Producer Organizations (FPO)/ Water User associations (WUA)' and the 'SPG', in case it is applicable.	
"Late Payment Surcharge"	shall have the meaning ascribed thereto in Article 10.3.3 of this Agreement;	
"Law"	shall mean in relation to this Agreement, all laws including Electricity Laws in force in India and any statute, ordinance, regulation, notification or code, rule, or any interpretation of any of them by an Indian Governmental Instrumentality and having force of law and shall further include without limitation all applicable rules, regulations, orders, notifications by an Indian Governmental Instrumentality pursuant to or under any of them and shall include without limitation all rules, regulations, decisions and orders of the Appropriate Commissions;	
Letter of Credit"	shall have the meaning ascribed thereto in Article 10.4 of this Agreement;	
'Letter of Award" or 'LoA"	shall mean Letter of Award issued by GRIDCO/DISCOM to the SPG for the project;	
MNRE"	shall mean the Ministry of New and Renewable Energy, Government of India;	

DISTRIA TPCODL UNANES

"Month"	shall mean a period of thirty (30) days from (and excluding) the date of the event, where applicable, else a calendar month;	
"Monthly Bill"	shall have the meaning ascribed thereto in Article 10.2.1 of this Agreement;	
"Party" and "Parties"	shall have the meaning ascribed thereto in the recital to this Agreement;	
"Payment Security Mechanism"	shall have the meaning ascribed thereto in Article 10.4 of this Agreement;	
"Project"	shall mean the solar power generation facility of Contracted Capacity of [Inser capacity] kW, located at [village] Village, [Revenue Inspector Circle Revenue Inspector (RI) Circle [district] District of Odisha State having a separate control system, metering and separate points of injection into the grid at Delivery point of [Name and location of 33 /11 kV DISCOM substation exactly as per the list in the RFP] Sub-station.	
	The Project shall include all units and auxiliaries such as water supply treatment or storage facilities, bay(s) for transmission system in the switchyard dedicated transmission line up to the Delivery Point and all the other assets buildings/structures, equipment, plant and machinery, facilities and related assets required for the efficient and economic operation of the powe generation facility, whether completed or at any stage of development and construction or intended to be developed and constructed for the purpose of supply of power as per this Agreement;	
"Preliminary Default Notice"	shall have the meaning ascribed thereto in Article 13 of this Agreement;	
"Project Capacity"	shall mean the maximum AC capacity of the Project at the point of injection on which the Power Purchase Agreement has been signed.	
"Prudent Utility Practices"	shall mean the practices, methods and standards that are generally accepted internationally from time to time by electric utilities for the purpose of ensuring the safe, efficient and economic design, construction, commissioning operation and maintenance of power generation equipment and which practices, methods and standards shall be adjusted as necessary, to take account of:	
	 a) operation and maintenance guidelines recommended by the manufacturers of the plant and equipment to be incorporated in the Project, 	
	b) the requirements of Indian Law; and the physical conditions at the site of the Project;	
Rebate	shall have the same meaning as ascribed thereto in Article 10.3.4 of this Agreement;	
"Scheduled Commissioning Date" or "SCD" of the Project	shall mean [Insert date that is nine (9) Months from the Effective Date];	
"Sub-station"	shall mean [Name and location of 33/11 kV DISCOM substation, exactly as per the list in the RFP]	
"Supplementary Bill"	shall have the meaning ascribed thereto in Article 10.7.1 of this Agreement;	
'Tariff'	shall have the same meaning as provided for in Article 9 of this Agreement;	
"Tariff Payment"	shall mean the payments to be made under Monthly Bills as referred to in Article 10 and the relevant Supplementary Bills;	

4HARPH

"Termination Notice" shall mean	the notice given by either Parties for termination of this Agreement
in accorda	nce with Article 13 of this Agreement;
	ONSHA DIST
	and the second se

2. ARTICLE 1: TERM OF AGREEMENT

2.1. Effective Date

2.1.1. This Agreement shall come into effect from the date of signing of this Agreement and such date shall be referred to as the Effective Date.

2.2. Term of Agreement

- 2.2.1 Subject to Article 2.3 and Article 2.4 of this Agreement, this Agreement shall be valid for a 'Term' from the Effective Date until the Expiry Date. This Agreement may be extended for a further period at least one hundred eighty (180) days prior to the Expiry Date, on mutually agreed terms and conditions.
- 2.2.2 The SPG is free to operate the Project beyond the Expiry Date if other conditions like Land Lease Agreement (as applicable), Consents, Clearances and Permits etc. allow. In such case, unless otherwise agreed by GRIDCO, GRIDCO shall not be obligated to procure power beyond the Expiry Date.

2.3. Early Termination

2.3.1 This Agreement shall terminate before the Expiry Date if either GRIDCO or SPG terminates the Agreement, pursuant to Article 13 of this Agreement.

2.4. Survival

2.4.1. The expiry or termination of this Agreement shall not affect any accrued rights, obligations and liabilities of the Parties under this Agreement, including the right to receive penalty as per the terms of this Agreement, nor shall it affect the survival of any continuing obligations for which this Agreement provides, either expressly or by necessary implication, which are to survive after the Expiry Date or termination including those under Article 11 (Force Majeure), Article 13 (Events of Default and Termination), Article 14 (Liability and Indemnification), Article 16 (Governing Law and Dispute Resolution), Article 17 (Miscellaneous Provisions), and other Articles and Schedules of this Agreement which expressly or by their nature survive the Term or termination of this Agreement shall continue and survive any expiry or termination of this Agreement.

3. CONDITIONS SUBSEQUENT

3.1. The SPG agrees and undertakes to make Financing Agreements for its Project, execute the Land Lease Agreement (as applicable) and shall provide necessary documents to GRIDCO in this regard within six (6) Months from the Effective Date.

4. CONSTRUCTION & DEVELOPMENT OF THE PROJECT

4.1. SPG's Obligations

A 0157

144

- 4.1.1. The SPG undertakes to be responsible, at SPG's own cost and risk, for:
- 4.1.1.1. The SPG shall be solely responsible and make arrangements for land, execution of Financing Agreements and associated infrastructure for development of the Project and for connectivity with the Sub-station for confirming the evacuation of power by the Scheduled Commissioning Date, and all clearances related thereto;
- 4.1.1.2. The SPG shall furnish the necessary documents to establish possession in the name of the SPG of the required land/ Land Lease Agreement (as applicable).

In case of a Land Lease Agreement, the SPG shall make payment of monthly rent directly in advance, by the 5th day of every month, by crediting the same to the lessor's (Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA) bank account as per the Land Lease Agreement until achieving the COD and for the first month of the PPA Term. In this regard, the SPG shall submit the bank receipts to GRIDCO by 10th of every month citing that the rent payments are made on or before 5th of every month;

In case of a Land Lease Agreement, the SPG shall make payment of monthly rent through GRIDCO./ DISCOM in advance, by the 5th day of every month, by crediting the same to the lessor's (Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA) bank account as per the Land Lease Agreement during the PPA Term, except for the first month of PPA Term. In such a case, GRIDCO/DISCOM will pay the rent to Lessor on monthly basis from the proceeds payable to the Lessee for the energy supplied by Lessee as per the PPA. In case there is a shortfall of payment for the payment to be made to the Lessor, then the Bidder shall be responsible to make a payment to GRIDCO/DISCOM for the equivalent amount thirty (30) Days prior to the due date of payment;

- i. The SPG shall make payment of monthly rent directly in advance, by the 5th day of every month, by crediting the same to the lessor's (Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA) bank account as per the Land Lease Agreement from the date of expiry of PPA term of 25 years until the expiry of lease period (as per the LLA terms and conditions). In this regard, the SPG shall submit the bank receipts to GRIDCO/DISCOM by 10th of every month citing that the rent payments are made on or before 5th of every month and this obligation will survive post expiry of the PPA. The provisions of this Article shall survive the termination of this PPA for any reason whatsoever.;
- 4.1.1.3. Obtaining all Consents, Clearances and Permits as required and maintaining all documents throughout the Term of this Agreement;
- 4.1.1.4. Designing, constructing, erecting, commissioning, completing and testing the Project in accordance with the applicable Law, the Grid Code, the terms and conditions of this Agreement and Prudent Utility Practices;
- 4.1.1.5. Commencement of supply of power up to the Contracted Capacity to GRIDCO no later than the Scheduled Commissioning Date and continuance of the supply of power throughout the Term of the Agreement;
- 4.1.1.6. Connecting the Project switchyard with the Interconnection Facilities at the Delivery Point. The SPG shall make adequate arrangements to connect the Project switchyard with the Interconnection Facilities at the Delivery Point;
- 4.1.1.7. Owning the Project throughout the Term of Agreement free and clear of encumbrances, except those expressly permitted under Article 15;
- 4.1.1.8. Fulfilling all obligations undertaken by the SPG under this Agreement;
- 4.1.1.9. The SPG shall be responsible for directly coordinating and dealing with GRIDCO/DISCOM, Indian Governmental Instrumentality and other authorities in all respects in regard to declaration of availability, scheduling and dispatch of power and due compliance with deviation and settlement mechanism and the applicable Law, Grid code, State Regulations.

4.2. Purchase and sale of Contracted Energy

4.2.1. Subject to the terms and conditions of this Agreement, the SPG undertakes to sell a minimum Contracted Energy [insert energy equivalent to target CUF of 15%] and a maximum Contracted Energy [insert energy equivalent to target CUF of 19%] to GRIDCO and GRIDCO undertakes to pay Tariff for all the energy supplied at the Delivery Point up to corresponding to the Contracted Capacity as per Article 4.3.1.

4.3. Right to Contracted Energy

4.3.1 GRIDCO, in any Contract Year shall not be obliged to purchase any additional energy from the SPG beyond what is stated in article 4.2.1. If for any Contract Year, except for the first Contract Year and last Contract Year of operation, it is found that the SPG has not been able to generate minimum energy of [insert energy equivalent to target CUF of 15%] kWh, on account of reasons solely attributable to the SPG, the non-compliance by SPG shall make the SPG liable to pay the compensation. For the first Contract Year and last Contract Year of operation, the above limits shall be considered on a pro-rata basis. The lower limit will, however, be relaxable by GRIDCO to the extent of grid non-availability for evacuation which is beyond the control of the SPG. This compensation shall be applied to the amount of shortfall in generation during the Contract Year. The amount of such compensation shall be as determined as per the APPC tariff or the Tariff, whichever is minimum. This compensation shall not be applicable in events of Force Majeure identified under PPA.



For generation in excess of [insert energy equivalent to target CUF of 18.7%] in a particular year shall be first adjusted towards generation compensation against offtake constraints as per article 4.9 and any remaining excess generation will be settled at 75% of applicable tariff.

4.3.2. In case at any point of time, the peak of capacity reached is higher than the Contracted Capacity and causes disturbance in the system at the point where power is injected, the SPG will have to forego the excess generation and reduce the output to the Contracted Capacity and shall also have to pay the charges (if applicable) as per applicable regulations.

4.4. Extensions of Time

- 4.4.1 In the event that the SPG is prevented from performing its obligations under Article 4.1 by the Scheduled Commissioning Date due to:
 - a. any GRIDCO Event of Default; or
 - b. Force Majeure Events affecting GRIDCO, or
 - c. Force Majeure Events affecting the SPG,

the Scheduled Commissioning Date and the Expiry Date shall be deferred, subject to Article 4.4.5, for a reasonable period but not less than 'day for day' basis, to permit the SPG or GRIDCO through the use of due diligence, to overcome the effects of the Force Majeure Events affecting the SPG or GRIDCO, or till such time such Event of Default is rectified by GRIDCO.

- 4.4.2. In case of extension due to reasons specified in Article 4.4.1(b) and (c), and if such Force Majeure Event continues even after a maximum period of three (3) months, any of the Parties may choose to terminate the Agreement as per the provisions of Article 13.5. In case neither Party terminates the Agreement under this clause, the Agreement shall stand terminated on the expiry of twelve (12) months of the continuation of the Force majeure event unless the Parties mutually agree to extend the Agreement for the further period.
- 4.4.3. If the Parties have not agreed, within thirty (30) days after the affected Party's performance has ceased to be affected by the relevant circumstance, on the time period by which the Scheduled Commissioning Date or the Expiry Date should be deferred, any Party may raise the Dispute to be resolved in accordance with Article 16.
- 4.4.4. As a result of such extension, the newly determined Scheduled Commissioning Date and newly determined Expiry Date shall be deemed to be the Scheduled Commissioning Date and the Expiry Date for the purposes of this Agreement.
- 4.4.5. Notwithstanding anything to the contrary contained in this Agreement, any extension of the Scheduled Commissioning Date arising due to any reason envisaged in this Agreement shall not be allowed beyond the date pursuant to Article 4.5.2.
- 4.4.6. Delay in achieving the COD of the Project beyond the Scheduled Commissioning Date for reasons other than those specified in Article 4.4.1 shall be an Event of Default on part of the SPG and shall be subject to the consequences specified in Article 4.5.

4.5. Liquidated Damages not amounting to penalty for delay in Commissioning

4.5.1. If the SPG is unable to achieve the COD of the Project by the Scheduled Commissioning Date other than for the reasons specified in Article 4.4.1, the SPG shall pay to GRIDCO, liquidated damages for the delay in such commissioning and making the Contracted Capacity available for dispatch by the Scheduled Commissioning Date as per the following:

Delay beyond the Scheduled Commissioning Date up to (& including) the date as on nine (9) months from the Effective Date: The total Performance Bank Guarantee amount shall be encashed on per day basis and proportionate to the balance capacity not achieved COD.

4.5.2. The maximum time period allowed for achieving the COD of the full Project Capacity with encashment of Performance Bank Guarantee shall be limited to 11 Months from the Effective Date. In case, the COD of the Project is delayed beyond 11 Months from the Effective Date, it shall be considered as an SPG Event of Default and provisions of Article 13 shall apply and the Contracted Capacity shall stand reduced/ amended to the Project Capacity Commissioned within 11 Months from the Effective Date and the PPA for the balance Capacity will stand terminated and shall be reduced from the Project Capacity.

The SPG further acknowledge that the amount of the liquidated damages fixed is genuine

and reasonable pre-estimate of the damages that may be suffered by GRIDCO.

4.6. Acceptance/ Performance Test

4.6.1. Prior to synchronization of the Project, the SPG shall be required to get the Project certified for the requisite acceptance/ performance test as may be laid down by respective authorities.

4.7. Third-party Verification

- 4.7.1. The SPG shall be further required to provide entry to the site of the Project free of all encumbrances at all times during the Term of the Agreement to GRIDCO and a third party nominated by any Indian Governmental Instrumentality for inspection and verification of the works being carried out by the SPG at the site of the Project.
- 4.7.2. The third-party may verify the construction works/ operation of the Project being carried out by the SPG and if it is found that the construction works/ operation of the Project is not as per the Prudent Utility Practices, it may seek clarifications from SPG or require the works to be stopped or to comply with the instructions of such third-party.

4.8. Breach of Obligations

4.8.1. The Parties herein agree that during the subsistence of this Agreement, subject to GRIDCO being in compliance of its obligations & undertakings under this Agreement, the SPG would have no right to negotiate or enter into any dialogue with any third party for the sale of Contracted Capacity of power which is the subject matter of this Agreement. It is the specific understanding between the Parties that such bar will apply throughout the entire term of this Agreement.

4.9. Generation compensation for Off-take constraints

4.9.1. Generation Compensation in offtake constraints due to Grid Unavailability: During the operation of the Project, there can be some periods where the Project can generate power but due to temporary transmission unavailability, the power is not evacuated, for reasons not attributable to the SPG. In such cases, subject to the submission of documentary evidence from the competent authority, the generation compensation shall be restricted to the following and there shall be no other claim, directly or indirectly against GRIDCO:

Duration of Grid unavailability	Provision for Generation Compensation
Grid unavailability in a Contract Year as defined in the PPA: (only period from 8 am to 6 pm to be counted):	Generation Loss = [(Average Generation per hour during the Contract Year) x (number of hours of grid unavailability during the Contract Year)]
	Where, Average Generation per hour during the Contract Year (kWh) = Total generation in the Contract Year (kWh) ÷ Total hours of generation in the Contract Year (hours)

The excess generation by the SPG equal to this generation loss shall be procured by GRIDCO at the Tariff so as to offset this loss in the succeeding 3 (three) Contract Years.

4.9.2. Offtake constraints due to Backdown: The SPG and GRIDCO shall follow the forecasting and scheduling process as per the regulations in this regard by OERC. The Government of India, as per Clause 5.2(u) of the Indian Electricity Grid Code (IEGC), provides for status of "must-run" to solar power projects. Accordingly, no solar power plant, duly commissioned, should be directed to back down by a GRIDCO/DISCOM/ Load Dispatch Centre (LDC). In case such eventuality of Backdown arises, including non-dispatch of power due to non-compliance with "Order No. 23/22/2019-R&R dated 28.06.2019 of Ministry of Power regarding opening and maintaining of adequate Letter of Credit (LC) as payment security mechanism under the Power Purchase Agreements by Distribution Licensees" and any clarifications or amendment thereto, except for the cases where the Backdown is on account of events like consideration of grid security or safety of any equipment or personnel or other such conditions, the SPG shall be eligible for a Minimum Generation Compensation, from GRIDOC, in the manner detailed below:

Duration of Backdown

Provision for Generation Compensation

CUSTRIN.

Hours of Backdown	Minimum Generation Compensation = 100% of [(Average Generation
during a monthly billing	per hour during the month) × (number of backdown hours during the
cycle.	month) × PPA Tariff]
	Where, Average Generation per hour during the month (kWh) = Total
	generation in the month (kWh) + Total hours of generation in the month

No back-down / curtailment to be ordered without giving formal/ written instruction for the same. The details of back-down / curtailment, including justifications for such curtailment, to be made public by the concerned Sate Load Dispatch Centre.

The SPG shall not be eligible for any compensation in case the Backdown is on account of events like consideration of grid security or safety of any equipment or personnel or other such conditions. The Generation Compensation shall be paid as part of the energy bill for the successive month as per JMR/EBC energy export statement.

5. SYNCHRONISATION, COMMISSIONING AND COMMERCIAL OPERATION

5.1. Synchronization, Commissioning and Commercial Operation

- 5.1.1 The SPG shall give the GRIDCO at least thirty (30) days' advanced preliminary written notice and at least fifteen (15) days' advanced final written notice, of the date on which it intends to synchronize the Project to the Interconnection Facilities.
- 5.1.2. Subject to Article 5.1.1, the Project may be synchronized by the SPG to the Interconnection Facilities when it meets all the connection conditions prescribed in applicable Grid Code then in effect and otherwise meets all other Indian Governmental Instrumentality for synchronization to the Interconnection Facilities.
- 5.1.3. The synchronization equipment and all necessary arrangements/ equipment including Remote Terminal Unit (RTU) for scheduling of power generated from the Project and transmission of data to the concerned authority as per applicable regulation shall be installed by the SPG at its generation facility of the Project at its own cost. The SPG shall synchronize its system with the Interconnection Facilities only after the approval of the synchronization scheme is granted by the head of the concerned substation/ and checking/verification is made by the concerned authorities of DISCOM/GRIDCO.
- 5.1.4. The SPG shall immediately after each synchronization/ tripping of generator, inform the Substation of the Interconnection Facilities to which the Project is electrically connected in accordance with the applicable Grid Code. In addition, the SPG will inject infirm power to grid time to time to carry out operational/ functional tests prior to COD. For avoidance of doubt, it is clarified that synchronization/ connectivity of the Project with the Interconnection Facilities shall not be considered as COD of the Project.
- 5.1.5. The SPG shall achieve COD of the Project within nine (9) Months from the <u>Effective Date</u>. Declaration of COD shall only be done upon the successful visit by the Commissioning Committee.
- 5.1.6. The Parties agree that for the purpose of commencement of the supply of electricity by SPG to GRIDCO, liquidated damages for delay, etc., the Scheduled Commissioning Date as defined in this Agreement shall be the relevant date.

6. DISPATCH AND SCHEDULING

6.1. Dispatch and Scheduling

PCOD

 6.1.1. The SPG shall be required to schedule its power as per the applicable regulations of SERC/ SLDC or any other competent agency and same being recognized by the SLDC or any other competent authority/ agency as per applicable regulation/ law/ direction and maintain compliance to the Indian Governmental Instrumentality, applicable Distribution Codes/ Odisha Grid Code requirements and directions, if any, as specified by concerned SLDC from time to time. Any deviation from the Schedule will attract the provisions of applicable regulation/ guidelines/ directions and any financial implication on account of this shall be on the account of the SPG. The SPG shall be responsible for directly coordinating and dealing with GRIDCO/ DISCOM, State Load Dispatch Center (SLDC), and other authorities in all respects in regard to declaration availability, scheduling and dispatch of power and due compliance with deviation and settlement mechanism and the applicable Odisha Grid Code regulations.

- 6.1.2 The SPG shall be responsible for any deviation from scheduling and for any resultant liabilities on account of charges for deviation as per applicable regulations. Deviation Settlement Management (DSM) charges on this account shall be directly paid by the SPG.
- 6.1.3. Auxiliary power consumption will be treated as per the concerned state regulations.

7. METERING

7.1. Meters

- 7.1.1. For installation of meters, meter testing, meter calibration, meter reading and all matters incidental thereto, the SPG and GRIDCO/DISCOM shall follow and be bound by the Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006, the Odisha Grid Code, as amended and revised from time to time.
- 7.1.2. The SPG shall bear all costs pertaining to installation, testing, calibration, maintenance, renewal and repair of meters at <u>SPG's side of</u> the Delivery Point.
- 7.1.3. In addition to ensure compliance of the applicable codes, the SPG shall install main and check meters at the Delivery Point, along with stand-by meter(s) as per the applicable regulations.

7.2. Reporting of Metered Data and Parameters

- 7.2.1. The Project will install necessary equipment for regular monitoring of required data and simultaneously for monitoring of the electric power generated from the Project.
- 7.2.2. Online arrangement would have to be made by the SPG for submission of above data regularly for the entire period of this Power Purchase Agreement to GRIDCO, MNRE and concerned agency as per applicable regulation/ directions.
- 7.2.3. Reports on the above parameters on monthly basis (or as required by regulation/ guidelines) shall be submitted by the SPG to the Ministry of New and Renewable Energy/ National Institute of Solar Energy through GRIDCO for entire period of PPA.

8. INSURANCES

8.1. Insurance

8.1.1 The SPG shall effect and maintain or cause to be effected and maintained, at its own cost and expense, throughout the Term of PPA, Insurances against such risks to keep the Project in good condition and shall take 'Industrial All Risk Insurance policy' covering risks against any loss or damage, with such deductibles and with such endorsements and co-insured(s), which the Prudent Utility Practices would ordinarily merit maintenance of and as required under the Financing Agreements, and under the applicable laws.

8.2. Application of Insurance Proceeds

8.2.1. In case of the Project not being implemented through Financing Agreement(s), save as expressly provided in this Agreement or the Insurances, the proceeds of any Insurance claim made due to loss or damage to the Project or any part of the Project shall be first applied to reinstatement, replacement or renewal of such loss or damage.

In case of the Project being financed through Financing Agreement(s), save as expressly provided in this Agreement or the Insurances, the proceeds of any Insurance claim made due to loss or damage to the Project or any part of the Project shall be applied as per such Financing Agreements.

8.2.2. If a Force Majeure Event renders the Project no longer economically and technically viable and the insurers under the Insurances make payment on a "total loss" or equivalent basis, DISCOM/GRIDCO shall have claim on such proceeds of such Insurance limited to outstanding dues of DISCOM/GRIDCO against SPG.

8.3. Effect on liability of GRIDCO / DISCOM

8.3.1. Notwithstanding any liability or obligation that may arise under this Agreement, any loss, damage, liability, payment, obligation or expense which is insured or not or for which the SPG can claim compensation, under any Insurance shall not be charged to or payable by GRIDCO/DISCOM. It is for the SPG to ensure that appropriate Insurance coverage is taken for payment by the insurer for the entire loss and there is no under Insurance or short adjustment etc.



9. APPLICABLE TARIFF

9.1 The SPG shall be entitled to receive the lowest discovered tariff through this tender, with effect from the COD, for the power sold to GRIDCO as reflected in the JMR/ EBC Energy export statement.

10. BILLING AND PAYMENT

10.1 General

- 10.1.1 From the commencement of supply of power, GRIDCO shall pay to the SPG, the Monthly Tariff Payments subject to the adjustments as per provisions of this Agreement including Article 6, in accordance with Article 9. All Tariff Payments by GRIDCO shall be in Indian Rupees.
- 10.1.2. The SPG shall be required to make arrangements and payments for import of energy (if any) as per applicable regulations.

10.2. Delivery and Content of Monthly Bills/ Supplementary Bills

10.2.1. The SPG shall issue to GRIDCO, a hard copy of the signed Monthly Bill, for the immediately preceding Month based on the JMR/ EBC energy export statement along with all relevant documents (payments made by SPG for drawl of power, payment of reactive energy charges, Metering charges or any other charges as per regulations of SERC/SLDC, if applicable.)

The Monthly Bill amount shall be the product of the energy as per JMR/ EBC energy export statement and the Applicable Tariff. Energy drawn from the grid will be regulated as per the regulations of Odisha.

10.3. Payment of Monthly Bills

- 10.3.1. GRIDCO shall pay the amount payable under the Monthly Bill by the Due Date to such account of the SPG, as shall have been previously notified by the SPG.
- 10.3.2. All payments required to be made under this Agreement shall also include any deduction or set off for:
 - i. deductions required by the Law; and
 - Amount claimed by GRIDCO, if any, from the SPG, will be adjusted from the monthly energy payment.

The SPG shall open a bank account (the "SPG's Designated Account") for all Tariff Payments to be made by GRIDCOto the SPG and notify GRIDCO of the details of such account at least sixty (60) days before the dispatch of the first Monthly Bill.

10.3.3. Late Payment Surcharge

In the event of delay in payment of a Monthly Bill by GRIDCO beyond thirty (30) days of its Due Date, a Late Payment Surcharge shall be payable to the SPG at the rate of 1.25% per month on the outstanding amount calculated on a day to day basis. The Late Payment Surcharge shall be claimed by the SPG through the Supplementary Bill.

10.3.4. Rebate

10:4

10.4.1

60 DL

M

For payment of any Bill on or before Due Date, the following Rebate shall be paid by the SPG to GRIDCOin the following manner and the SPG shall not raise any objections to the payments made under this article.

- a. A Rebate of two percent (2%) shall be payable to GRIDCO for the payments made within a period of seven working days of the presentation of Bill along with required supporting documents at the office of Chief General Manager (PP), GRiDCO.
- b. Any payments made within the Due Date shall be allowed a rebate of one percent (1%).
- c. For the billing purpose, if bill is received by 12:00 Hours, the present date is considered or else billing date.
- d. No Rebate shall be payable on the Bills raised on account of Change in Law relating to taxes, duties, cess, etc. and on Supplementary Bill.

Payment Security Mechanism Letter of Credit

CARIDCO shall provide to the SPG, in respect of payment of its Monthly Bills and/or

Supplementary Bills, a Monthly unconditional, revolving and irrevocable letter of credit ("Letter of Credit"), opened and maintained which may be drawn upon by the SPG in accordance with this Article.

- 10.4.2. Not later than one (1) Month before the start of supply, GRIDCO through a scheduled bank open a Letter of Credit in favour of the SPG, to be made operative from a date prior to the Due Date of its first Monthly Bill under this Agreement. The Letter of Credit shall have a term of twelve (12) Months and shall be renewed annually, for an amount equal to:
 - a. for the first Contract Year, equal to the estimated average monthly billing;
 - b. for each subsequent Contract Year, equal to the average of the monthly billing of the previous Contract Year.
- 10.4.3. Provided that the SPG shall not draw upon such Letter of Credit prior to the Due Date of the relevant Monthly Bill and/or Supplementary Bill and shall not make more than one drawl in a Month.
- 10.4.4. Provided further that if at any time, such Letter of Credit amount falls short of the amount specified in Article 10.4.2 due to any reason whatsoever, GRIDCO shall restore such shortfall within fifteen (15) days.
- 10.4.5. GRIDCO shall cause the scheduled bank issuing the Letter of Credit to intimate the SPG, in writing regarding the establishing of such irrevocable Letter of Credit.
- 10.4.6. GRIDCO shall ensure that the Letter of Credit shall be renewed not later than its expiry.
- 10.4.7. All costs relating to opening, maintenance of the Letter of Credit shall be borne by GRIDCO
- 10.4.8. If GRIDCO fails to pay undisputed Monthly Bill or Supplementary Bill or a part thereof within and including the Due Date, then, subject to Article 10.4.6 & 10.5.2, the SPG may draw upon the Letter of Credit, and accordingly the bank shall pay without any reference or instructions from GRIDCO, an amount equal to such Monthly Bill or Supplementary Bill or part thereof, in accordance with Article 10.4.3 above, by presenting to the scheduled bank issuing the Letter of Credit, the following documents:
 - a. a copy of the Monthly Bill or Supplementary Bill which has remained unpaid to SPG and;
 - a certificate from the SPG to the effect that the bill at item (i) above, or specified part thereof, is in accordance with the Agreement and has remained unpaid beyond the Due Date;

10.5. Disputed Bill

- 10.5.1. If GRIDCOdoes not dispute a Monthly Bill or a Supplementary Bill raised by the SPG within fifteen (15) days of receiving such Bill shall be taken as conclusive.
- 10.5.2. If GRIDCO disputes the amount payable under a Monthly Bill or a Supplementary Bill, as the case may be, it shall pay undisputed amount of the invoice amount and it shall within fifteen (15) days of receiving such Bill, issue a notice (the "Bill Dispute Notice") to the invoicing Party setting out:
 - i. the details of the disputed amount;
 - ii. its estimate of what the correct amount should be; and iii) all written material in support of its claim.
- 10.5.3. If the SPG agrees to the claim raised in the Bill Dispute Notice issued pursuant to Article 10.5.2, the SPG shall revise such Bill and present along with the next Monthly Bill. In such a case excess amount shall be refunded along with interest at the same rate as Late Payment Surcharge, which shall be applied from the date on which such excess payment was made by the disputing Party to the invoicing Party, including the date on which such payment has been received a refund.
- 10.5.4. If the SPG does not agree to the claim raised in the Bill Dispute Notice issued pursuant to Article 10.5.2, it shall, within fifteen (15) days of receiving the Bill Dispute Notice, furnish a notice (Bill Disagreement Notice) to the GRIDCO providing:
 - i. reasons for its disagreement;
 - its estimate of what the correct amount should be; and iii) all written material in support of its counter-claim.

- 10.5.5. Upon receipt of the Bill Disagreement Notice by the GRIDCO under Article 10.5.4, authorized representative(s) or a director of the Board of Directors/ member of the Board of GRIDCO and SPG shall meet and make best endeavours to amicably resolve such dispute within fifteen (15) days of receipt of the Bill Disagreement Notice.
- 10.5.6. If the Parties do not amicably resolve the Dispute within fifteen (15) days of receipt of Bill Disagreement Notice pursuant to Article 10.5.4, the matter shall be referred to Dispute resolution in accordance with Article 16.
- 10.5.7. For the avoidance of doubt, it is clarified that despite a Dispute regarding an invoice, GRIDCO shall, without prejudice to its right to Dispute, be under an obligation to make payment of the undisputed amount of the invoice amount in the Monthly Bill.

10.6. Quarterly and Annual Reconciliation

- 10.6.1. The Parties acknowledge that all payments made against the Monthly Bills and Supplementary Bills shall be subject to quarterly reconciliation and annual reconciliation at the end of each Contract Year within 30 days to take into account the JMR/ EBC energy export statement, Tariff adjustment payments, Rebate, Late Payment Surcharge, or any other reasonable circumstance provided under this Agreement.
- 10.6.2. The Parties, therefore, agree that as soon as all such data in respect of any quarter of a Contract Year or a full Contract Year as the case may be has been finally verified and adjusted, the SPG and GRIDCO shall jointly sign such reconciliation statement. Within fifteen (15) days of signing of a reconciliation statement, the SPG shall make appropriate adjustments in the next Monthly Bill. Late Payment Surcharge/ interest shall be payable in such a case from the date on which such payment had been made to the invoicing Party or the date on which any payment was originally due, as may be applicable. Any Dispute with regard to the above reconciliation shall be dealt with in accordance with the provisions of Article 16.

10.7. Payment of Supplementary Bill

- 10.7.1. SPG may raise a ("Supplementary Bill") for payment on account of:
 - i. Adjustments required in the JMR/ EBC energy export statement (if applicable); or
 - ii. Change in Law as provided in Article 12

And such Supplementary Bill shall be paid by the other Party.

10.7.2. GRIDCO shall remit all amounts due under a Supplementary Bill raised by the SPG to the SPG's Designated Account by the Due Date, except open access charges, RLDC or scheduling charges and transmission charges (if applicable). For Supplementary Bill on account of adjustment required in the JMR/ EBC energy export statement, Rebate as applicable to Monthly Bills pursuant to Article 10.3.5 shall equally apply. No surcharge will be applicable other than that on the monthly energy payment and associated debit and credit note.

11. FORCE MAJEURE

11.1. Definitions

11.1.1 In this Article, the following terms Affected Party, and Force Majeure, and shall have the following meanings.

11.2. Affected Party

11.2.1. An affected Party means GRIDCO/ DISCOM or the SPG whose performance has been affected by an event of Force Majeure.

11.3. Force Majeure

A DIST

6)Fi

11.3.1 A 'Force Majeure' means any event or circumstance or combination of events those stated below that wholly or partly prevents or unavoidable delays an Affected Party in the performance of its obligations under this Agreement, but only if and to the extent that such events or circumstances are not within the reasonable control, directly or indirectly, of the Affected Party and could not have been avoided if the Affected Party had taken reasonable care or complied with Prudent Utility Practices:

Act of God, including, but not limited to lightning, drought, fire and explosion (to the extent originating from a source external to the site), earthquake, volcanic eruption, landslide, flood, cyclone, typhoon or tomado if and only if it is declared/ notified by the competent state

/ central authority/ agency (as applicable);

- any act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, terrorist or military action if and only if it is declared/notified by the competent state/ central authority/ agency (as applicable); or
- c. radioactive contamination or ionizing radiation originating from a source in India or resulting from another Force Majeure Event mentioned above excluding circumstances where the source or cause of contamination or radiation is brought or has been brought into or near the Project by the Affected Party or those employed or engaged by the Affected Party.

11.4. Force Majeure Exclusions

- 11.4.1. Force Majeure shall not include (i) any event or circumstance which is within the reasonable control of the Parties and (ii) the following conditions, except to the extent that they are consequences of an event of Force Majeure:
 - a. Unavailability, late delivery, or changes in cost of the plant, machinery, equipment, materials, spare parts or consumables for the Project;
 - b. Delay in the performance of any contractor, sub-contractor or their agents;
 - C. Non-performance resulting from normal wear and tear typically experienced in power generation materials and equipment;
 - d. Strikes at the facilities of the Affected Party;
 - e. Insufficiency of finances or funds or the agreement becoming onerous to perform; and
 - f. Non-performance caused by, or connected with, the Affected Party's:
 - Negligent or intentional acts, errors or omissions;
 - ii. Failure to comply with an Indian Law; or
 - iii. Breach of, or default under this Agreement.

11.5. Notification of Force Majeure Event

- 11.5.1. The Affected Party shall give notice to the other Party of any event of Force Majeure as soon as reasonably practicable, but not later than fifteen (15) days after the date on which such Party knew or should reasonably have known of the commencement of the event of Force Majeure. If an event of Force Majeure results in a breakdown of communications rendering it unreasonable to give notice within the applicable time limit specified herein, then the Party claiming Force Majeure shall give such notice as soon as reasonably practicable after reinstatement of communications, but not later than one (1) day after such reinstatement.
- 11.5.2. Provided that such notice shall be a pre-condition to the Affected Party's entitlement to claim relief under this Agreement. Such notice shall include full particulars of the event of Force Majeure, its effects on the Party claiming relief and the remedial measures proposed. The Affected Party shall give the other Party regular (and not less than monthly) reports on the progress of those remedial measures and such other information as the other Party may reasonably request about the Force Majeure Event.
- 11.5.3. The Affected Party shall give notice to the other Party of (i) the cessation of the relevant event of Force Majeure; and (ii) the cessation of the effects of such event of Force Majeure on the performance of its rights or obligations under this Agreement, as soon as practicable after becoming aware of each of these cessations.

11.6. Duty to Perform and Duty to Mitigate

11.6.1. To the extent not prevented by a Force Majeure Event pursuant to Article 11.3, the Affected Party shall continue to perform its obligations pursuant to this Agreement. The Affected Party shall use its reasonable efforts to mitigate the effect of any Force Majeure Event as soon as practicable.

11.7. Available Relief for a Force Majeure Event

11.7.1. Subject to this Article 11:



- no Party shall be in breach of its obligations pursuant to this Agreement except to the extent that the performance of its obligations was prevented, hindered or delayed due to a Force Majeure Event;
- every Party shall be entitled to claim relief in relation to a Force Majeure Event in regard to its obligations;
- c. For avoidance of doubt, neither Party's obligation to make payments of money due and payable prior to occurrence of Force Majeure events under this Agreement shall be suspended or excused due to the occurrence of a Force Majeure Event in respect of such Party.
- d. Provided that no payments shall be made by either Party affected by a Force Majeure Event for the period of such event on account of its inability to perform its obligations due to such Force Majeure Event.

12. CHANGE IN LAW

12.1 Definitions

In Article 12, the term Change in Law shall refer to the occurrence of any of the following events pertaining to this project only after the last date of the bid submission, including

- i. the enactment of any new law; or
- ii. an amendment, modification or repeal of an existing law; or
- iii. the requirement to obtain a new consent, permit or license; or
- iv. any modification to the prevailing conditions prescribed for obtaining consent, permit or license, not owing to any default of the SPG; or
- any change in the rates of any Taxes including any duties and cess or introduction of any new tax made applicable for setting up the Power Project and supply of power from the Power Project by the SPG which have a direct effect on the Project.

However, Change in Law shall not include

- i. any change in taxes on corporate income or
- ii. any change in any withholding tax on income or dividends distributed to the shareholders of the SPG, or
- iii. any change on account of regulatory measures by the Appropriate Commission.

In the event a Change in Law results in any adverse financial loss/ gain to the SPG then, in order to ensure that the SPG is placed in the same financial position as it would have been had it not been for the occurrence of the Change in Law, the SPG/ GRIDCO shall be entitled to compensation by the other Party, as the case may be, subject to the condition that the quantum and mechanism of compensation payment shall be determined and shall be effective from such date as may be decided by the Appropriate Commission.

In the event of any decrease in the recurring/ nonrecurring expenditure by the SPG or any income to the SPG on account of any of the events as indicated above, SPG shall file an application to the Appropriate Commission no later than sixty (60) days from the occurrence of such event, for seeking approval of Change in Law. In the event of the SPG failing to comply with the above requirement, in case of any gain to the SPG, GRIDCO shall withhold the monthly tariff payments on an immediate basis, until compliance of the above requirement by the SPG.

12.2. Relief for Change in Law

- 12.2.1. The aggrieved Party shall be required to approach the Appropriate Commission for seeking approval of Change in Law.
- 12.2.2. The decision of the Appropriate Commission to acknowledge a Change in Law and the date from which it will become effective, provide relief for the same, shall be final and governing on both the Parties.

13. EVENTS OF DEFAULT AND TERMINATION

SPG Event of Default

DISTRAS

- 13.1.1. The occurrence and/or continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by GRIDCO of its obligations under this Agreement, shall constitute an SPG Event of Default:
 - the failure to commence supply of power to GRIDCO up to the Contracted Capacity, by the end of the period specified in Article 4, or failure to continue supply of Contracted Capacity to GRIDCO/DISCOM after Commercial Operation Date throughout the term of this Agreement, or

if

İ.

- a) the SPG assigns, mortgages or charges or purports to assign, mortgage or charge any of its assets or rights related to the Project in contravention of the provisions of this Agreement; or
- b) the SPG transfers or novates any of its rights and/ or obligations under this Agreement, in a manner contrary to the provisions of this Agreement; except where such transfer
 - is in pursuance of a Law; and does not affect the ability of the transferee to perform, and such transferee has the financial capability to perform, its obligations under this Agreement or
 - is to a transferee who assumes such obligations under this Agreement and the Agreement remains effective with respect to the transferee;
- ii. if (a) the SPG becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding-up proceedings and such proceedings remain uncontested for a period of thirty (30) days, or (b) any winding up or bankruptcy or insolvency order is passed against the SPG, or (c) the SPG goes into liquidation or dissolution or has a receiver or any similar officer appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, provided that a dissolution or liquidation of the SPG will not be a SPG Event of Default if such dissolution or liquidation is for the purpose of a merger, consolidation or reorganization and where the resulting company retains creditworthiness similar to the SPG and expressly assumes all obligations of the SPG under this Agreement and is in a position to perform them; or
- iii. the SPG repudiates this Agreement and does not rectify such breach within a period of thirty (30) days from a notice from GRIDCO in this regard; or
- iv. except where due to any GRIDCO's failure to comply with its material obligations, the SPG is in breach of any of its material obligations pursuant to this Agreement, and such material breach is not rectified by the SPG within thirty (30) days of receipt of first notice in this regard given by GRIDCO; or
- occurrence of any other event which is specified in this Agreement to be a material breach/ default of the SPG;

13.2. GRIDCO Event of Default

- 13.2.1. The occurrence and the continuation of any of the following events, unless any such event occurs as a result of a Force Majeure Event or a breach by the SPG of its obligations under this Agreement, shall constitute the Event of Default on the part of defaulting GRIDCO:
 - GRIDCO fails to pay (with respect to a Monthly Bill or a Supplementary Bill), subject to Article 10.5, for a period of ninety (90) days after the Due Date and the SPG is unable to recover the amount outstanding to the SPG through the Letter of Credit,
 - ii. GRIDCO repudiates this Agreement and does not rectify such breach even within a period of sixty (60) days from a notice from the SPG in this regard; or
 - II. except where due to any SPG's failure to comply with its obligations, GRIDCO is in material breach of any of its obligations pursuant to this Agreement, and such material breach is not rectified by GRIDCO within sixty (60) days of receipt of notice in this regard from the SPG to GRIDCO; or



if

- GRIDCO becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding-up proceedings and such proceedings remain uncontested for a period of sixty (60) days, or
- any winding up or bankruptcy or insolvency order is passed against GRIDCO, or
- GRIDCO goes into liquidation or dissolution or a receiver or any similar officer is appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, provided that it shall not constitute a GRIDCO Event of Default, where such dissolution or liquidation of GRIDCO or DISCOM is for the purpose of a merger, consolidation or reorganization and where the resulting entity has the financial standing to perform its obligations under this Agreement and has creditworthiness similar to DISCOM and expressly assumes all obligations of GRIDCO and is in a position to perform them; or;
- Occurrence of any other event which is specified in this Agreement to be a material breach or default of GRIDCO.

13.3. Procedure for cases of SPG Event of Default

- 13.3.1. Upon the occurrence and continuation of any SPG Event of Default under Article 13.1, GRIDCO shall have the right to deliver to the SPG, with a copy to the representative of the lenders to the SPG with whom the SPG has executed the Financing Agreements, a notice stating its intention to terminate this Agreement (GRIDCO Preliminary Default Notice), which shall specify in reasonable detail, the circumstances giving rise to the issue of such notice.
- 13.3.2. Following the issue of a GRIDCO Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply and it shall be the responsibility of the Parties to discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Event of Default having regard to all the circumstances.
- 13.3.3. During the Consultation Period, the Parties shall continue to perform their respective obligations under this Agreement.
- 13.3.4. Within a period of seven (7) days following the expiry of the Consultation Period unless the Parties shall have otherwise agreed to the contrary or the SPG Event of Default giving rise to the Consultation Period shall have ceased to exist or shall have been remedied, GRIDCO may terminate this Agreement by giving a written Termination Notice of sixty (60) days to the SPG.
- 13.3.5. Subject to the terms of this Agreement, upon occurrence of a SPG Event of Default under this Agreement, the lenders in concurrence with the GRIDCO, may exercise their rights, if any, under Financing Agreements, to seek substitution of the SPG by a selectee for the residual period of the Agreement, for the purpose of securing the payments of the total debt amount from the SPG and performing the obligations of the SPG. However, in the event the lenders are unable to substitute the defaulting SPG within the stipulated period, GRIDCO may terminate the PPA and may acquire the Project assets for an amount equivalent to 90% of the debt due or less as mutually agreed, failing which, the lenders may exercise their mortgage rights and liquidate the Project assets.

Provided that any substitution under this Agreement can only be made with the prior consent of GRIDCO including the condition that the selectee meets the eligibility requirements of Request for Selection (RfS) issued by GRIDCO and accepts the terms and conditions of this Agreement.

- 13.3.6. The lenders in concurrence with GRIDCO may seek to exercise the right of substitution under Article 13.3.5 by an amendment or novation of the PPA in favour of the selectee. The SPG shall cooperate with GRIDCO to carry out such substitution and shall have the duty and obligation to continue to operate the Project in accordance with this PPA till such time as the is finalized. In the event of Change in Shareholding/Substitution of Promoters triggered by the Financial Institutions leading to signing of fresh PPA with a new entity, an amount of Rs. 1 Lakh per MW +18% GST per transaction as facilitation fee (non-refundable) shall be deposited by the SPG to GRIDCO.
- 13.3.7. In the event the lenders are unable to substitute the defaulting SPG with in the stipulated period, GRIDCO may terminate the PPA and may acquire the Project assets for an amount equivalent to 90% of the debt due, failing which, the lenders may exercise their mortgage rights and liquidate

the Project assets.

13.4. Procedure for cases of GRIDCO Event of Default

- 13.4.1. Upon the occurrence and continuation of any GRIDCO Event of Default specified in Article 13.2, the SPG shall have the right to deliver to GRIDCO, an SPG Preliminary Default Notice, which notice shall specify in reasonable detail the circumstances giving rise to its issue.
- 13.4.2. Following the issue of an SPG Preliminary Default Notice, the Consultation Period of ninety (90) days or such longer period as the Parties may agree, shall apply and it shall be the responsibility of the Parties to discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Event of Default having regard to all the circumstances.
- 13.4.3. During the Consultation Period, the Parties shall continue to perform their respective obligations under this Agreement.
- 13.4.4. After a period of two hundred ten (210) days following the expiry of the Consultation Period and unless the Parties shall have otherwise agreed to the contrary or GRIDCO Event of Default giving rise to the Consultation Period shall have ceased to exist or shall have been remedied, GRIDCO under intimation to SPG shall, subject to the prior consent of the SPG, novate its part of the PPA to any third party, including its Affiliates within the stipulated period. In the event the aforesaid novation is not acceptable to the SPG may terminate the PPA and at its discretion require GRIDCO to either (i) takeover the Project assets by making a payment of the termination compensation equivalent to the amount of the debt due and 150% (one hundred and fifty per cent) of the adjusted equity or, (ii) pay to the SPG, damages, equivalent to 6 (six) months, or balance PPA Term whichever is less, of charges for its contracted capacity, with the Project assets being retained by the SPG.

Provided further that at the end of three (3) months period from the period mentioned in this Article 13.4.4, this Agreement may be terminated by the SPG.

13.5. Termination due to Force Majeure

13.5.1. If the Force Majeure Event or its effects continue to be present beyond a period as specified in Article 4.4.2, either Party shall have the right to cause termination of the Agreement. In such an event this Agreement shall terminate on the date of such Termination Notice without any further liability to either Party from the date of such termination.

14. LIABILITY AND INDEMNIFICATION

14.1. Indemnity

- 14.1.1. The SPG shall indemnify, defend and hold GRIDCO harmless against:
 - any and all third-party claims against GRIDCO for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by the SPG of any of its obligations under this Agreement; and
 - b. any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest actually suffered or incurred by GRIDCO from third party claims arising by reason of a breach by the SPG of any of its obligations under this Agreement, (provided that this Article 14 shall not apply to such breaches by the SPG, for which specific remedies have been provided for under this Agreement).
- 14.1.2. GRIDCO shall indemnify, defend and hold the SPG harmless against:
 - any and all third-party claims against the SPG, for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by GRIDCO of any of their obligations under this Agreement; and
 - b. any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest ('Indemnifiable Losses') actually suffered or incurred by the SPG from third party claims arising by reason of a breach by GRIDCO of any of its obligations.

14.2. Procedure for claiming Indemnity

14.2.1. Third-party claims

a. Where the Indemnified Party is entitled to indemnification from the Indemnifying Party pursuant to Article 14.1.1(a) or 14.1.2(a), the Indemnified Party shall promptly notify the Indemnifying

(PCOD

Party of such claim referred to in Article 14.1.1(a) or 14.1.2(a) in respect of which it is entitled to be indemnified. Such notice shall be given as soon as reasonably practicable after the Indemnified Party becomes aware of such a claim. The Indemnifying Party shall be liable to settle the indemnification claim within thirty (30) days of receipt of the above notice. Provided however that, if:

- the Parties choose to refer the dispute before the Arbitrator in accordance with Article 16.3.2;
- and

t

ii. the claim amount is not required to be paid/ deposited to such third party pending the resolution of the Dispute,

the Indemnifying Party shall become liable to pay the claim amount to the Indemnified Party or to the third party, as the case may be, promptly following the resolution of the Dispute, if such Dispute is not settled in favour of the Indemnified Party.

b. The Indemnified Party may contest the claim by referring to the Arbitrator for which it is entitled to be Indemnified under Article 14.1.1(a) or 14.1.2(a) and the Indemnifying Party shall reimburse to the Indemnified Party all reasonable costs and expenses incurred by the Indemnified Party. However, such Indemnified Party shall not settle or compromise such claim without first getting the consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed.

An Indemnifying Party may, at its own expense, assume control of the defence of any proceedings brought against the indemnified Party if it acknowledges its obligation to indemnify such Indemnified Party, gives such Indemnified Party prompt notice of its intention to assume control of the defence, and employs an independent legal counsel at its own cost that is reasonably satisfactory to the Indemnified Party.

14.3. Indemnifiable Losses

- 14.3.1. Where an Indemnified Party is entitled to Indemnifiable Losses from the Indemnifying Party pursuant to Article 14.1.1(b) or 14.1.2(b), the Indemnified Party shall promptly notify the Indemnifying Party of the Indemnifiable Losses actually incurred by the Indemnified Party. The Indemnifiable Losses shall be reimbursed by the Indemnifying Party within thirty (30) days of receipt of the notice seeking Indemnifiable Losses by the Indemnified Party. In case of nonpayment of such losses after a valid notice under this Article 14.3, such event shall constitute a payment default under Article 13.
- 14.4. Limitation of Liability
- 14.4.1. Except as expressly provided in this Agreement, neither the SPG nor its/ their respective officers, directors, agents, employees or affiliates (or their officers, directors, agents or employees), shall be liable or responsible to the other Party or its affiliates, officers, directors, agents, employees, successors or permitted assigns or their respective insurers for incidental, indirect or consequential damages, connected with or resulting from performance or non-performance of this Agreement, or anything done in connection herewith, including claims in the nature of lost revenues, income or profits (other than payments expressly required and properly due under this Agreement), any increased expense of, reduction in or loss of power generation or equipment used therefore, irrespective of whether such claims are based upon breach of warranty, tort (including negligence, whether of GRIDCO, the SPG or others), strict liability, contract, breach of statutory duty, operation of law or otherwise.
- 14.4.2. GRIDCO shall have no recourse against any officer, director or shareholder of the SPG or any Affiliate of the SPG or any of its officers, directors or shareholders for such claims excluded under this Article. The SPG shall have no recourse against any officer, director or shareholder of GRIDCO, or any affiliate of GRIDCO or any of its officers, directors or shareholders for such claims excluded under this Article.

14.5. Duty to Mitigate

14.5.1. The Parties shall endeavor to take all reasonable steps so as mitigate any loss or damage which has occurred under this Article 14.

15. ASSIGNMENTS AND CHARGES

151 Assignments

PCODI

This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement shall not be assigned by any Party, except to the Project lender or lender's representative as security for their debt under the Financing Agreements, other than by mutual consent between the Parties to be evidenced in writing. Such assignment shall be agreed to by GRIDCO subject to the compliance of provisions contained in this Agreement and more specifically to the provisions of Article 4.1.1 of this Agreement. In no case, such assignment shall be permissible prior to the declaration of COD.

Provided that, GRIDCO shall permit assignment of any of SPG's rights and obligations under this Agreement in favour of the lenders to the SPG, if required under the Financing Agreements. Provided that, such consent shall not be withheld if GRIDCO seeks to transfer to any transferee all of its rights and obligations under this Agreement.

The enforcement of the rights and obligation between the SPG and the GRIDCO provided in this Agreement shall not be treated as an assignment but an enforcement of the terms agreed under this Agreement.

Provided further that any successor(s) or permitted assign(s) identified after mutual agreement between the Parties may be required to execute a new agreement on the same terms and conditions as are included in this Agreement. An amount of Rs. 1 Lakh per Transaction as Facilitation Fee (non-refundable) shall be deposited by the SPG to GRIDCO. Provided further that, such consent shall not be withheld by the SPG if GRIDCO seeks to transfer to any affiliate all of its rights and obligations under this Agreement.

In the event of Change in Shareholding/Substitution of Promoters triggered by the Financial Institutions leading to signing of fresh PPA with a New Entity, an amount of INR 1 Lakh per Transaction as Facilitation Fee (non-refundable) shall be deposited by the SPG to GRIDCO.

15.2. Permitted Charges

15.2.1. SPG shall not create or permit to subsist any encumbrance over all or any of its rights and benefits under this Agreement, other than as set forth in Article 15.1 and the Guidelines.

16. GOVERNING LAW AND DISPUTE RESOLUTION

16.1. Governing Law

16.1.1. This Agreement shall be governed by and construed in accordance with the Laws of India. Any legal proceedings in respect of any matters, claims or disputes under this Agreement shall be under the jurisdiction of appropriate courts in Bhubaneswar/ Odisha.

16.2. Amicable Settlement and Dispute Resolution

- i. Either Party is entitled to raise any claim, dispute or difference of whatever nature arising under, out of or in connection with this Agreement ("Dispute") by giving a written notice (Dispute Notice) to the other Party, which shall contain:
 - a. a description of the Dispute;
 - b. the grounds for such Dispute; and
 - c. all written material in support of its claim.
- ii. The other Party shall, within thirty (30) days of issue of Dispute Notice issued under Article 16.2.1(i), furnish:
 - a. counter-claim and defenses, if any, regarding the Dispute; and
 - b. all written material in support of its defenses and counter-claim.
- iii. Within thirty (30) days of issue of Dispute Notice by any Party pursuant to Article 16
 - a. if the other Party does not furnish any counterclaim or defense under Article 16
 - b. or thirty (30) days from the date of furnishing counterclaims or defense by the other Party, both the Parties to the Dispute shall meet to settle such Dispute amicably. If the Parties fail to resolve the Dispute amicably within thirty (30) days from the later of the dates mentioned in this Article 16.2.1.
 - c. the Dispute shall be referred for dispute resolution in accordance with Article 16.3.

16.3. Dispute Resolution

16.3.1. Dispute Resolution by the Appropriate Commission

I. Where any Dispute or differences arises in relation to this agreement of any nature whatsoever including the construction, interpretation or implementation of the provisions of this agreement as well as claim made by any Party for any change in or determination of the Tariff or any matter related to Tariff or claims made by any Party which partly or wholly relate to any change in the Tariff or determination of any of such claims could result in change in the Tariff, and relates to any matter agreed to be referred to the Appropriate Commission, shall be submitted to adjudication by the Appropriate Commission. Appeal against the decisions of the Appropriate Commission shall be made only as per the provisions of the Electricity Act, 2003, as amended from time to time.

ii. GRIDCO shall be entitled to co-opt the lenders (if any) as a supporting party in such proceedings before the Appropriate Commission.

16.3.2 Dispute Resolution through Arbitration

- II. If the Dispute arising as per Article 16.2.1 is not amicably resolved & such dispute is not covered in Article 16.3.1(i), such Dispute shall be resolved by arbitration under the provisions of the Electricity Act, 2003 (as amended from time to time) as under: Proceedings as well as appointment of the arbitrator(s) shall be carried out by the Appropriate Commissions under the Electricity Act 2003 as amended from time to time. As stipulated by the said Electricity Act 2003, the said arbitration will take place as per the provisions of the Arbitration and Conciliation Act 1996 as amended from time to time.
- III. The place of arbitration shall be at Bhubaneswar. The language of the arbitration shall be English.
- iv. The Arbitration Tribunal's award shall be substantiated in writing. The Arbitration Tribunal shall also decide on the costs of the arbitration proceedings and the allocation thereof.
- v. The provisions of this Article shall survive the termination of this PPA for any reason whatsoever.
- vi. The award shall be of majority decision.
- vii. GRIDCO shall be entitled to co-opt the lenders (if any) as a supporting party in such arbitration proceedings.

16.4. Parties to Perform Obligations

16.4.1. Notwithstanding the existence of any Dispute and difference referred to the Appropriate Commission and save as the Appropriate Commission may otherwise direct by a final or interim order, the Parties hereto shall continue to perform their respective obligations (which are not in dispute) under this Agreement.

17. MISCELLANEOUS PROVISIONS

17.1. Amendment

17.1.1. This Agreement may only be amended or supplemented by a written agreement between the Parties.

17.2. Third-party Beneficiaries

17.2.1. This Agreement is solely for the benefit of the Parties and their respective successors and permitted assigns and shall not be construed as creating any duty, standard of care or any liability to, any person not a party to this Agreement.

17.3. Waiver

- 17.3.1. No waiver by either Party of any default or breach by the other Party in the performance of any of the provisions of this Agreement shall be effective unless in writing duly executed by an authorized representative of such Party.
- 17.3.2. Neither the failure by either Party to insist on any occasion upon the performance of the terms, conditions and provisions of this Agreement nor time or other indulgence granted by one Party to the other Parties shall act as a waiver of such breach or acceptance of any variation or the relinquishment of any such right or any other right under this Agreement, which shall remain in full force and effect.

17.4. Confidentiality

- 17.4.1. The Parties undertake to hold in confidence this Agreement and not to disclose the terms and conditions of the transaction contemplated hereby to third-parties, except:
 - a. to their professional advisors;
 - b. to their officers, contractors, employees, agents or representatives, financiers, who need
 - to have access to such information for the proper performance of their activities; or
 - c. disclosures required under Law, without the prior written consent of the other Party.

17.5. Severability

ran

LANKSN

17.5.1. The invalidity or unenforceability, for any reason, of any part of this Agreement shall not prejudice

or affect the validity or enforceability of the remainder of this Agreement, unless the part held invalid or unenforceable is fundamental to this Agreement.

17.6. Notices

- 17.6.1. All notices or other communications which are required to be given under this Agreement shall be in writing and in the English language.
- 17.6.2. If to the SPG, all notices or other communications which are required must be delivered personally or by registered post or facsimile or any other method duly acknowledged to the addresses below:

Address: [•] Attention: [•] Email: [•] Fax. No.: [•] Telephone No.: [•]

- 17.6.3. If to GRIDCO, all notices or communications must be delivered personally or by registered post or facsimile or any other mode duly acknowledged to the address(es) below:
- 17.6.4. All notices or communications given by facsimile shall be confirmed by sending a copy of the same via post office in an envelope properly addressed to the appropriate Party for delivery by registered mail. All notices shall be deemed validly delivered upon receipt evidenced by an acknowledgment of the recipient, unless the Party delivering the notice can prove in case of delivery through the registered post that the recipient refused to acknowledge the receipt of the notice despite efforts of the postal authorities.
- 17.6.5. Any Party may by notice of at least fifteen (15) days to the other Party change the address and/or addresses to which such notices and communications to it are to be delivered or mailed.

17.7. Language

- 17.7.1. All agreements, correspondence and communications between the Parties relating to this Agreement and all other documentation to be prepared and supplied under the Agreement shall be written in English, and the Agreement shall be construed and interpreted in accordance with English language.
- 17.7.2. If any of the agreements, correspondence, communications or documents are prepared in any language other than English, the English translation of such agreements, correspondence, communications or documents shall prevail in matters of interpretation.

17.8. Restriction of Shareholders / Owners' Liability

17.8.1. Parties expressly agree and acknowledge that none of the shareholders of the Parties hereto shall be liable to the other Parties for any of the contractual obligations of the concerned Party under this Agreement. Further, the financial liabilities of the shareholder/s of each Party to this Agreement, shall be restricted to the extent provided in the relevant act in India.

17.9. Restriction of Shareholders / Owners' Liability

17.9.1. Parties expressly agree and acknowledge that none of the shareholders of the Parties hereto shall be liable to the other Parties for any of the contractual obligations of the concerned Party under this Agreement. Further, the financial liabilities of the shareholder/s of each Party to this Agreement, shall be restricted to the extent provided in the relevant act in India.

17.10. Taxes and Duties

- 17.10.1 The SPG shall bear and promptly pay all statutory taxes, duties, levies and cess, assessed/ levied on the SPG, contractors or their employees that are required to be paid by the SPG as per the Law in relation to the execution of the Agreement and for supplying power as per the terms of this Agreement.
- 17.10.2. GRIDCO shall be indemnified and held harmless by the SPG against any claims that may be made against GRIDCO in relation to the matters set out in Article 17.9.1.

17.10.3. GRIDCO shall not be liable for any payment of, taxes, duties, levies, cess whatsoever for discharging any obligation of the SPG by GRIDCO on behalf of SPG.

17.11. Independent Entity

- 17.11.1. The SPG shall be an independent entity performing its obligations pursuant to the Agreement.
- 17.11.2. Subject to the provisions of the Agreement, the SPG shall be solely responsible for the manner in which its obligations under this Agreement are to be performed. All employees and representatives of the SPG or contractors engaged by the SPG in connection with the performance of the Agreement shall be under the complete control of the SPG and shall not be deemed to be employees, representatives, contractors of GRIDCO and nothing contained in the Agreement or in any agreement or contract awarded by the SPG shall be construed to create any contractual relationship between any such employees, representatives or contractors and GRIDCO.

17.12. Compliance with Law

17.12.1. Despite anything contained in this Agreement but without prejudice to this Article, if any provision of this Agreement shall be in deviation or inconsistent with or repugnant to the provisions contained in the Electricity Act, 2003, or any rules and regulations made thereunder, such provision of this Agreement shall be deemed to be amended to the extent required to bring it into compliance with the aforesaid relevant provisions as amended from time to time.

17.13. Breach of Obligations

17.13.1. The Parties acknowledge that a breach of any of the obligations contained herein would result in injuries. The Parties further acknowledge that the amount of the liquidated damages or the method of calculating the liquidated damages specified in this Agreement is a genuine and reasonable pre-estimate of the damages that may be suffered by the non-defaulting party in each case specified under this Agreement.

IN WITNESS WHEREOF the Parties have caused the Agreement to be executed through their duly authorized representatives as of the date and place set forth above. For and on behalf of GRIDCO

For and on behalf of [SPG]

Name: Designation: Address: Signature with seal:

Name: Designation: Address: Signature with seal:

Witness: 1.

Witness: 1

2.

2.



Annexure 1: Details of land

SCHEDULE PROPERTY

All that piece and parcel of Schedule Property is mentioned below:

- Name: [name of Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA]
- Mobile no.: [Mobile no. of Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA]
- AADHAAR no.: [AADHAAR no. (applicable for Individual Farmers or Group of Farmers)]
- Registration details: [Registration no., date of registration, registered under [insert the act under which it is registered] (appliable for Cooperatives or Panchayats or FPO or WUA)]
- Mouja: [mouja name]
- Thana, thana no.: [thana name], [thana no.]
- Tehsil, tehsil no.: [tehsil name], [tehsil no.]
- Panchayat: [panchayat name]
- Village: [village name]
- RI circle: [RI circle name]
- District: [district name]
- Pin code: [6-digit pin code]
- Googie coordinate: [for example 20.135172, 85.084869 format]

Khata no.	Plot no.	Kissam	Extent/ rakba of land (in acres)	

Note: Please note that this Annexure is a sample format and the Bidder shall provide the above information for each Project applied for an individual substation. In case the Bidder has land spread across 2 different administrative units, for example, in 2 different tehsil or village, then the information shall be provided under separate Annexures.

Annexure 2: Schedule for minimum Contracted Energy generation for the Project

The year-wise schedule for minimum Contracted Energy generation for the Project is mentioned below:

Contract Year	Minimum Contracted Energy (kWh)@19% CUF
1	
2	
3	
4	
5	
6	
7	
8	
9	
11	
12	
<u>10</u> <u>11</u> <u>12</u> <u>13</u>	
14	
15	
15 16	
17	
18	
<u>19</u>	
20	
21	
22 23	
23	
24	
<u>25</u>	



Annexure 3: Performance Bank Guarantee format

Format for Performance Security

(To be on non-judicial stamp paper of appropriate value as per Stamp Act relevant to place of execution.)

(Performance Security shall be submitted in 2 Nos. of Bank Guarantees in the ratio of 20% and 80% value.)

In consideration of the M/s [Insert name of the Successful Bidder] (hereinafter referred to as 'Solar Project Generator or SPG') submitting the response to RFP No. [RFP] dated [DD MMM YYYY] issued byTata Power Central Odisha Distribution Limited (TPCODL) inter alia for selection of the Project having the capacity of [Capacity] kWp, at [Insert name of the SPP location] for supply of energy therefrom on long term basis to Grid Corporation of Odisha Limited (hereinafter referred to as "DISCOM") and TPCODL considering such response to the RFP (which expression shall unless repugnant to the context or meaning thereof include its executers, administrators, successors and assignees), TPCODL issuing an intimation Letter of Award no. [intimation no.] dated [DD MMM YYYY] to [name of SPG] as per terms of RFP and the same having been accepted by the SPG resulting in a Power Purchase Agreement (PPA) to be entered into with DISCOM, for purchase of Delivered Energy from the SPG. As per the terms of the RFP, the [insert name & address of bank] hereby agrees unequivocally, irrevocably and unconditionally to pay to TPCODL at [TPCODL's address] forthwith on demand in writing from TPCODL or any Officer authorized by it in this behalf, any amount up to and not exceeding INR [Amount] (Indian Rupees [in words]) only, on behalf of the SPG.

This Bank Guarantee shall be valid and binding on this Bank up to and including [insert expiry date in DD MMM YYYY] along with claim date up to and including [insert claim date in DD MMM YYYY] and shall not be terminable by any notice or any change in the constitution of the Bank or the term of contract or by any other reasons whatsoever and our liability hereunder shall not be impaired or discharged by any extension of time or variations or alternations made, given, or agreed with or without our knowledge or consent, by or between parties to the respective agreement.

Our liability under this Bank Guarantee is restricted to INR [Amount] (Indian Rupees [in words]) only. Our Bank Guarantee shall remain in force until [insert expiry date in DD MMM YYYY] TPCODL shall be entitled to invoke this Bank Guarantee till [insert claim date in DD MMM YYYY].

The Guarantor Bank hereby agrees and acknowledges that TPCODL shall have the right to invoke this Bank Guarantee in part or in full, as it may deem fit at its branch at Bhubaneswar.

The Guarantor Bank hereby expressly agrees that it shall not require any proof in addition to the written demand by TPCODL, made in any format, raised at the above-mentioned address of the Guarantor Bank, in order to make the said payment to TPCODL.

The Guarantor Bank shall make payment hereunder on first demand without restriction or conditions and notwithstanding any objection by the SPG and/ or any other person. The Guarantor Bank shall neither require TPCODL to justify the invocation of this Bank Guarantee, nor shall the Guarantor Bank have any recourse against TPCODL in respect of any payment made hereunder.

This Bank Guarantee shall be interpreted in accordance with the laws of India and the courts at Bhubaneswar shall have exclusive jurisdiction.

The Guarantor Bank represents that this Bank Guarantee has been established in such form and with such content that it is fully enforceable in accordance with its terms as against the Guarantor Bank in the manner provided herein.

This Bank Guarantee shall not be affected in any manner by reason of merger, amalgamation, restructuring or any other change in the constitution of the Guarantor Bank or whatsoever.

This Bank Guarantee shall be a primary obligation of the Guarantor Bank and accordingly TPCODL shall not be obliged before enforcing this Bank Guarantee to take any action in any court or arbitral proceedings against the SPG, to make any claim against or any demand on the SPG or to give any notice to the SPG or to enforce any security held by TPCODL or to exercise, levy or enforce any distress, diligence or other process against the SPG.

DISTRIA

The Guarantor Bank acknowledges that this Bank Guarantee is not personal to TPCODL and may be assigned, in whole or in part, (whether absolutely or by way of security) by TPCODL to any entity to whom TPCODL is entitled to assign its rights and obligations under the PPA.

Signature:

Name: [Name] Power of Attorney No.: [PoA no.]

For

[Insert name of the Bank] Email ID of the Bank: [Email id] Banker's Stamp and Full Address: [Address] Dated this [DD] day of [MMM], [YYYY]

Witness

Signature
Name and Address

2
Signature
Name and Address

Note:

- 1. The stamp paper shall be in the name of the executing bank and of appropriate value of stamp paper.
- The Performance Security shall be executed by any of the nationalized or scheduled bank in India.



LEASE AGREEMENT

This Lease Agreement entered into on this [Day] day of [Month] month [Year] year ("Effective Date") at [Location].

BETWEEN:

[Name of the Lessor - Individual Farmer/ Group of Farmers/ Cooperatives/ Panchayats/ Farmer Producer Organizations/ Water User Associations] (hereinafter referred to as the "Lessor", which expression shall, wherever the context so requires or admits, shall mean and include his legal heirs, executors, administrators and assignees successors in interest);

AND:

[Name of the Lessee – Solar Power Generator (SPG)], represented by [Name of the authorized signatory of the Lessee], (hereinafter referred to as the "Lessee, which expression shall, wherever the context so requires or admits, shall mean and include its executors, administrators and assignees successors in interest).

The Lessor and Lessee are individually referred to as a 'Party' and collectively referred to as 'Parties'.

- WHEREAS the Lessor is the owner in possession of the barren/ agricultural/ non-agricultural land measuring [area in acres] acres situated at Panchayat [Panchayat name], Village [Village name], RI Circle [RI Circle name], District [District name] in Odisha which is more fully described in the Schedule hereunder and hereinafter referred to as the "Schedule Property" as given in Annexure 1.
- II. WHEREAS the Lessee being a [sole proprietor firm/ partnership firm/ company] incorporated under the [relevant act in India], having its registered office at [registered office address] to plan, develop and operate solar energy-based power plant under MNRE Scheme notified on 8th March 2019 and further amendments.
- III. (a) WHEREAS pursuant to the request of the Lessee, the Lessor has agreed to grant the lease, the Lessee has agreed to take on lease from the Lessor the Schedule Property which is more fully described in Schedule written hereunder and hereinafter referred to as the "Schedule Property" for setting up of the [Project capacity 500 kW to 2,000 MW] kW solar energy based power plant (hereinafter referred to as the "Project").

(b) That pursuant to the request of the Lessee, the Lessor has submitted an application under Section 8-A, of The Odisha Land Reforms Act, 1960 for the conversion of the Schedule Property. The Lessee, on behalf of the Lessor, shall presume that the Schedule Property is deemed to have been converted for non-agricultural purposes, in case it is required to convert the land for non-agricultural purposes. However, the Lessee shall be responsible for obtaining the approval for converting the status of the Schedule Property, as per the Applicable Law.

- IV. WHEREAS the Lessor and Lessee have mutually determined the lease rent charges as detailed in Article 3 of this Lease Agreement.
- V. NOW THIS LEASE AGREEMENT WITNESSES THAT in consideration of the above and of the mutual covenants of the Parties hereto, the Lessor hereby grants and the Lessee hereby accepts the lease of the Schedule Property on the following terms and conditions:

DIST

1. PURPOSE OF LEASE:

The grant of lease by the Lessor to the Lessee in respect of the Schedule Property is for the purpose of developing the Project (including associated infrastructure like office, fence etc.) under the MNRE Scheme notified on 8th March 2019 and further amendments.

2. PERIOD OF THE LEASE

The period of this Lease Agreement shall be for twenty-seven (27) years from the Effective Date ("Lease Period") which may be renewed at the option of the Lessee and Lessor for further period, on such mutually agreeable terms as may be agreed at the time of renewal, by both the Parties, by executing and registering a separate lease agreement.

3. <u>RENT</u>

- a) The rent payable by the Lessee to the Lessor for the Schedule Property shall be INR [in number]/-(Indian Rupees [in words]) only per annum per acre for the Lease Period. The portion of the Schedule Property less than one acre shall be calculated in terms of decimal and the rent payable for the same shall be at INR [in number]/- (Indian Rupees [in words]) only per decimal or part thereof, per annum for the Lease Period.
- b) The annual rent shall be paid in twelve (12) equal installments and each installment shall be paid, in advance, by the 5th day of every month, by crediting the same to the Lessor's bank account, the details of which may be furnished by the Lessor from time to time.
- c) The SPG shall make payment of monthly rent directly in advance, by the 5th day of every month, by crediting the same to the lessor's (Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA) bank account as per the Land Lease Agreement until achieving the COD and for the first month of the PPA Term. In this regard, the SPG shall submit the bank receipts to GRIDCO/DISCOM by 10th of every month citing that the rent payments are made on or before 5th of every month;
- d) The Lessee shall make payment of monthly rent through GRIDCO/DISCOM in line with the Power Purchase Agreement (PPA) signed between the Lessee and GRIDCO/DISCOM during the PPA Term, except for the first month of PPA Term. In such a case, GRIDCO/DISCOM will pay the rent to Lessor on monthly basis from the proceeds payable to the Lessee for the energy supplied by Lessee as per the PPA. In case there is a shortfall of payment for the payment to be made to the Lessor, then the Bidder shall be responsible to make a payment to GRIDCO/DISCOM for the equivalent amount thirty (30) Days prior to the due date of payment.
- e) The SPG shall make payment of monthly rent directly in advance, by the 5th day of every month, by crediting the same to the lessor's (Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA) bank account as per the Land Lease Agreement from the date of expiry of PPA term of 25 years (as per the PPA terms and conditions) until the expiry of Lease Period. In this regard, the SPG shall submit the bank receipts to GRIDCO/DISCOM by 10th of every month citing that the rent payments are made on or before 5th of every month and this obligation will survive post expiry of the PPA term of 25 years;
- f) The rent hereby reserved shall be paid by enhancing the same at the end of every [•] year(s), at
 [•%] or [INR in amount] on the rent hereby agreed.
- g) If the Lessee delays the payment of rent by the due date of every month, for any reason, the same shall be paid by adding the interest at the rate [•]% for the said delayed period.



h) The monthly rent schedule is attached in Annexure 2 (Rent Schedule).

4. GENERAL TERMS

- a) In consideration of the rent herein agreed as payable to the Lessor being paid by the Lessee through GRIDCO/DISCOM regularly and on complying with other terms and conditions and covenants by the Lessee, the Lessee shall peacefully possess and enjoy the Schedule Property during the Lease Period without any interruption by the Lessor.
- b) The Lessor shall allow the Lessee or its representatives to conduct a survey and other related work.
- c) The Lessor has no objections for the Lessee to establish the Project in the Schedule Property, which is the purpose of the grant of this lease and to that effect, the Lessee entering into any agreements, deeds with companies, individuals, developers/ third party, etc. in respect of the Schedule Property.
- d) The Lessor has no objections for the Lessee or its representatives for installation of machinery, equipment, etc. to establish the Project in the Schedule Property and all work relating to thereto, including but not limited to laying poles, wires, etc.

5. EVENT OF SALE, ACCEPTANCE OF LEASE BY THE NEW LESSOR

- a) In the event of the Lessor transferring their rights/ interest in any manner during the existence of the lease to any other Person, the same may be allowed without affecting the rights of the Lessee under the Lease Agreement in any manner and the Lessor shall inform the Lessee about the acquiring of the right/ interest in respect of the Schedule Property and on receipt of such information, the Lessee shall accept such Person's lessorship of the Schedule Property and obtain a written confirmation from such Person to the effect that he will be bound by the terms of the Lease Agreement.
- b) In the event of the Lessor transferring their rights/ interest to any other Person, the same may be informed to the Lessee and the Lessor shall ascertain and obtain all the necessary documents from the Person to the effect that the Person will be bound by the terms and conditions of the Lease Agreement for the balance Lease Period or for using the said documents for renewal of the lease agreement, if required mutually by the parties.
- During the subsistence of the Lease Agreement, the Lessor shall not carry any activity, in the Schedule Property, other than those agreed in this Lease Agreement;
- d) The change in the legal status of the Lessee shall not affect the terms and conditions of this Lease Agreement.
- e) Two (2) original Lease Agreements shall be made for the Lessee and the Lessor.
- f) In the event of any dispute in respect of the Schedule Property, the Lessee shall deposit the rent in the concerned civil court. In the event of retention of the rent with the Lessee, the Lessee shall pay the same together with interest thereon at the rate [.]% for such period.
- g) The Lessee shall not offer or create any charge or encumbrance by offering the same as by way of mortgage, security, etc. in favor of any banks or financial institutions in respect of the loans or advances or any other financial facilities that may be availed by the Lessee.

DISTR/

h) The Lessor shall pay all the tax/ any other statutory or other charges, as applicable, in respect of the Schedule Property during the Lease Period or extended lease period, if applicable.

6. PAYMENT OF STAMP DUTY AND REGISTRATION CHARGES:

The stamp duty and other registration charges, as applicable for this Lease Agreement shall be paid by the Lessee during the Lease Period.

7. FORCE MAJEURE:

It is also agreed and understood between the Parties that in case of any mishap due to fire, earthquake, strike, floods, tempest, war, riot, civil war or civil commotions, mob violence, civil disturbance, act of God or on account of terrorist attack, the Lessor shall not be liable for any loss or damage that may be occasioned to the Lessee/ its merchandise.

8. ADDRESSES FOR CORRESPONDENCE, ETC

Any notice and/ or communications between the Parties shall be deemed to be sufficient if delivered by hand under acknowledgment or sent by registered post acknowledgment due to the following address or the address that may be intimated in writing to the Lessee by the Lessor from time to time or email post an acknowledgment:

LESSOR'S:

Name of the contact person: [•] Designation of the contact person: [•] Address: [•] Email: [•] Contact no.: [•]

LESSEE'S:

Name of the contact person: [•] Designation of the contact person: [•] Address: [•] Email: [•] Contact no.: [•]

9. LESSOR'S DUTIES, COVENANTS, AND OBLIGATIONS

- a) The Lessor hereby covenants with the Lessee that the Lessee regularly paying the rents hereby reserved and performing and observing all the covenants of the Lessee herein contained, shall be entitled, during the subsistence of this lease to enjoy the Schedule Property without let, hindrance or interference from the Lessor or any other Person claiming through or under him; Still, in the event of the Lessee restrained from enjoying the peaceful possession of the Schedule Property or on account of any action by the Government during the Lease Period and in the event of dispossession of the Lessee from the Schedule Property or any portion thereof forcibly, due to any default of the Lessor, the Lessor shall make good the reasonable loss that may be suffered by the Lessee.
- b) The Lessor shall offer necessary support and cooperation to the Lessee in its process to obtain required permissions, approvals, clearances, etc., from any statutory authority or other local bodies for the purpose of obtaining and license, permissions, etc., for installation of the Project. However, obtaining such permissions, approvals, clearances, etc., shall be the sole responsibility of Lessee.

10. LESSEE'S COVENANT AND OBLIGATIONS

The Lessee hereby covenants with the Lessor as under:

- a) The Schedule Property shall be utilized for the purpose, as mentioned under Article 0;
- b) The Lessee shall pay the rents regularly and promptly, as mentioned under Article 3.

11. TERMINATION AND RE-ENTRY

The Lease shall be terminable under all or any of the following circumstances, namely --

- a) by efflux of time;
- b) in the event of breach by either Party of the terms, conditions and covenants hereof;
- c) if the Schedule Property or any part thereof is severely damaged or destroyed due to any unforeseen circumstances or Force Majeure situation, etc., and these damages are not restored to by the Lessor within a reasonable time or if the demised premises is acquired compulsorily by any authority (in this case, the Lessee will have the recourse of arbitration against such authority);
- d) After the expiry of the Lease Period, the Lessee shall handover the Schedule Property to the Lessor as it existed previously at the time of this Lease Agreement (subject to normal wear and tear). The Lessee will ensure that the Project after the 'end of life' (when they become defective/ non-operational/ non-repairable) are disposed of in accordance with the "e-waste (Management and Handling) Rules, 2011" notified by the Government and as revised and amended from time to time, failing which the cost of restoring the Schedule Property to the previously existing condition will be recovered by the Lessor from the proceeds payable to the Lessee in lieu of energy supplied by Lessee as per the Power Purchase Agreement.
- e) If the PPA signed between the Lessee and GRIDCO is terminated due to any reasons whatsoever.

12. VARIATION:

The Lessor and the Lessee hereto acknowledge that this Lease Agreement supersedes all prior communications between them including all oral or written proposals. Any variation, addition and modifications of this Lease Agreement between the Parties shall be valid only if in writing by the Lessor and Lessees authorized representative.

13. ARBITRATION:

a) Any disputes or differences arising between the Parties hereto as to the effect, interpretation or application any of the clauses of this Lease Agreement or as to their rights, duties or liabilities thereunder, or as to any act, matter or thing arising out of, or consequent to, or in connection with this Lease Agreement shall be referred to and resolved by arbitration by referring the same for arbitration to any retired District Judge and shall be resolved finally at his arbitration under arbitration and Conciliation Act 1996 and its Amendments or any other Enactment. The arbitration proceedings shall be held at Odisha, preferably at Bhubaneswar/ Cuttack, and shall be in English Language.

DISTRI

b) This Lease Agreement shall be governed by the laws of India. The Courts at Odisha alone shall have the jurisdiction to entertain and or try any dispute arising out of or in connection with or in relation to the terms of this Lease Agreement.

IN WITNESS WHEREOF the Parties hereto have executed these presents in the presence of the witnesses attesting hereunder on the day, month and year mentioned hereinabove.

Signed and delivered by the Lessor

Signed and delivered by the Lessor

Signature: Name of the Authorized Signatory: Date: Signature: Name of the Authorized Signatory: Date:

Witnesses:

In the presence of: 1.

Signature: Name: Designation:

2.

Signature: Name: Designation: Witnesses:

In the presence of: 1.

Signature: Name: Designation:

2.

Signature: Name: Designation:



Annexure 1 (SCHEDULE PROPERTY)

All that piece and parcel of Schedule Property is mentioned below:

- Name: [name of Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA]
- Mobile no.: [Mobile no. of Individual Farmers or Group of Farmers or Cooperatives or Panchayats or FPO or WUA]
- AADHAAR no.: [AADHAAR no. (applicable for Individual Farmers or Group of Farmers)]
- Registration details: [Registration no., date of registration, registered under [insert the act under which it is registered] (appliable for Cooperatives or Panchayats or FPO or WUA)]
- Mouja: [mouja name]
- Thana, thana no.: [thana name], [thana no.]
- Tehsil, tehsil no.: [tehsil name], [tehsil no.]
- Panchayat: [panchayat name]
- Village: [village name]
- RI circle: [RI circle name]
- District: [district name]
- Pin code: [6-digit pin code]
- Google coordinate: [for example 20.135172, 85.084869 format]

Khata no.	Plot no.	Kissam	Extent/ rakba of land (in acres)	Current usage (Agriculture/ non-agriculture/ barren)

Note: Please note that this Annexure is a sample format and the Bidder shall provide the above information for each Project applied for an individual substation. In case the Bidder has land spread across 2 different administrative units, for example, in 2 different tensil or village, then the information shall be provided under separate Annexures.



Annexure 2 (RENT SCHEDULE)

The annual and monthly rent schedule is attached below:

Year count	Annual rent (INR)	Monthly rent (INR)
Year 1		
Year 2		

440		
Year 26		
Year 27		

